

Data Analytics for Finance & Macro Research Centre



Factor Models with Downside Risk

D. Massacci, L. Sarno, L. Trapani

Working paper No. 2021/7 | October 2021 | ISSN 2516-5933



Factor Models with Downside Risk^{*}

Daniele Massacci[†] Lucio Sarno[‡] Lorenzo Trapani[§]

This Version: October 2021

^{*}We are grateful for comments and suggestions to Adam Farago, Alexandros Kostakis, Michael Weber, Dacheng Xiu and to participants at presentations held at: the 31st EC^2 Conference, Paris, 11-12 December 2020; the 14th International Conference on Computational and Financial Econometrics (CFE 2020); the 9th Italian Congress of Econometrics and Empirical Economics; the 13th Annual SoFiE Conference, San Diego; the University of Essex; the University of Liverpool; and the University of Verona. All errors are our own.

[†]Daniele Massacci is with King's College London. Email: daniele.massacci@kcl.ac.uk.

[‡]Lucio Sarno is with the University of Cambridge and the Centre for Economic Policy Research (CEPR). Email: l.sarno@jbs.cam.ac.uk.

[§]Lorenzo Trapani is with the University of Nottingham. Email: lorenzo.trapani@nottingham.ac.uk.

Factor Models with Downside Risk

We propose a conditional model of asset returns in the presence of common factors and downside risk. Specifically, we generalize existing latent factor models in three ways: we show how to estimate the threshold which identifies the 'disappointment' event triggering the bad state of the world; we permit different factor structures for asset returns in good and bad states; we show how to recover the observable factors' risk premia from the estimated latent ones in different states. The usefulness of the model is illustrated through two applications to cross-sections of asset returns in equity markets and other major asset classes.

1 Introduction

A growing body of research argues that the cross-section of asset returns should and does reflect a premium for bearing downside risk. This risk premium compensates investors for holding assets that covary strongly with the market when the market declines or is abnormally low: such assets are undesirable precisely because they offer particularly low returns in bad times. For example, Ang, Chen, and Xing (2006) show that there is a downsize risk premium in the crosssection of US stock returns, a result confirmed by Farago and Tédongap (2018) using a five-factor conditional model. Lettau, Maggiori, and Weber (2014) show that a conditional capital asset pricing model that allows for downside risk can explain not only US stocks, but also the cross section of returns in currency and other major asset classes. These results are consistent with the original conceptual framework of Markovitz (1959), who advocated using semivariance as a measure of risk, rather than variance, because semivariance measures downside losses rather than upside gains.

In a parallel development in the context of asset pricing models, the literature has also sought to tackle the vexing issue of potentially biased estimation of unconditional (i.e., linear, and hence free from downside risk) asset pricing models arising from omitting pricing factors. Since it is not at all clear which pricing factors should be used (the true factors), virtually any model which uses observable factors is at risk of omitting a relevant pricing factor, thereby suffering from omitted variable bias. This misspecification problem is addressed rigorously in a recent, seminal contribution by Giglio and Xiu (2021), who propose a three-pass procedure to conduct inference on the risk premia of any observable factor. The procedure solves both the omitted variable bias problem, and the much debated "factor zoo" issue (see Cochrane (2011); and Harvey and Liu (2020)), by considering a latent factor model which allows to span the underlying true factor space.

The two research areas summarized above indicate that an *unconditional*, linear asset pricing model based on observable factors, which constitutes the workhorse model of empirical asset pricing, can suffer from at least two possible misspecification issues: the omission of relevant pricing factors, and the presence of a downside regime. Both these issues can be resolved by using a *conditional*, latent factor model which allows for the presence of downside risk. However, despite the intuitive appeal of such a setup, inference for asset pricing models with downside risk and latent factors is a challenging task. In addition to the nonlinearity of the model, allowing for different regimes means that the number of common latent factors may be different between good and bad states of the world; the loadings on the factors may also change across states; the "disappointment event" that triggers the downside state is, in general, unknown; and, finally, the estimation of risk premia in this context, taking downside risk into account, needs to be fully developed. Our paper offers a methodological contribution, addressing all these issues, and presenting a nonlinear model of asset returns in the presence of common factors and downside risk; further, we consider a *latent* factor model approach, which allows to span the entire factor space as advocated in Giglio and Xiu (2021). This setup is general enough to encompass previous models with downside risk, while allowing for different common factors across states and estimation of the threshold determining the downside state.¹

We make at least three contributions to the literature on factor models of asset returns. First, while we follow the literature in determining the disappointment event in terms of the relative position of the market return with respect to a threshold value, unlike previous studies we estimate the value of the threshold rather than setting it *a priori* equal to a prespecified value; we also offer a methodology to test whether the threshold value is equal to a prespecified value such as the ones used in the literature. This is achieved by generalizing the inferential theory for panel threshold models proposed by Massacci (2017) to the case where the number of common factors and their loadings are allowed to differ across states; and we also derive the relevant inferential theory for the estimated threshold. Second, we develop a procedure to estimate the number of common factors in a model with downside risk, by explicitly allowing for the case where such number may differ across regimes. This is done in a similar fashion to Trapani (2018), although it is not a trivial extension since it requires a preliminary round of estimation where the full

¹The motivation for a downside risk premium arises from various theoretical perspectives. In some cases, the rationale is behavioral and linked to theories of loss aversion, as in the setups of Ang, Chen, and Xing (2006) and Farago and Tédongap (2018), whereas in other cases it is based on rational asset pricing models, as in Lettau, Maggiori, and Weber (2014). The specific nature of the mechanism capable of generating the downside risk premium is not a focus of this paper. We simply take seriously the finding that such premium can exist in equilibrium and ask how it should be identified in an empirical asset pricing model.

factor model is estimated (thereby including the threshold, and the spaces spanned by factors and loadings), and the strong consistency of all estimates is required. Finally, we complete the inferential theory by developing a further step, to allow for the mapping of the estimated latent factors in each stage onto economically relevant observable ones. This step can be viewed as an extension of Giglio and Xiu (2021) to a nonlinear setting. Overall, the distilled essence of these results is a general conditional model of asset returns with downside risk that can be estimated and tested without imposing arbitrary restrictions on the stochastic process of returns. In our view, this setup constitutes at least a workhorse model for conducting empirical asset pricing in a downside risk framework.

We employ this methodology to study empirically the factor structure and estimate risk premia in the presence of downside risk using two datasets previously employed in other studies - a dataset composed of equity portfolios (the same as used by Farago and Tédongap (2018)) and a dataset spanning multiple asset classes (based on a variety of sources similar to Lettau, Maggiori, and Weber (2014)). In both cases, our approach produces superior goodness of fit relative to both unconditional (linear) factor models and restricted conditional models that impose the number of factors or the threshold. The empirical estimation of the proposed latent factor model with downside risk proves illuminating when compared to an unconditional model of asset returns. Indeed, the unconditional model performs poorly in both states of the world, with systematic pricing errors that are positive in the downside state and negative in the satisfactory state - i.e., the unconditional model predicts much higher returns than the realized ones in the downside state, and lower returns than the realized ones in the satisfactory state. Puzzingly, when analyzing the full sample performance of the unconditional model, its performance can appear satisfactory, but this is only because averaging across negative and positive pricing errors in different states gives the false impression that the model can capture the average excess returns accurately. In other words, assessing the performance of the unconditional factor model over the full sample can give the illusion that it performs well when in fact it performs poorly in both regimes. In contrast, the latent factor model with downside risk appears to have no systematic tendency to generate positive or negative pricing errors in either regime, and therefore also unconditionally, fitting the cross-sections of asset returns used here very well. The inadequacy of an unconditional model is well illustrated by Figure 1, which plots unconditional portfolio equity returns against average portfolio equity returns conditional on either good or bad states (defined with a threshold of -3and -6 percent on the left-hand-side and right-hand-side plots, respectively). The scatter plots make clear how a linear approximation underestimates returns in good times and overestimates returns in bad times. It is also clear that a linear relationship cannot possibly fit these returns data in the downside state, making apparent the need for a nonlinear model that changes across states.



Figure 1: Conditional and unconditional realised returns for equity portfolios

Notes. The scatter plot shows realized average returns in upside and downside against unconditional average returns for the 130 equity portfolios described in Section 6.1.1 for fixed and estimated threshold values, which are discussed in Section 6.2.1.

Our empirical results highlight the importance of estimating the threshold (rather than imposing it *a priori*) and of allowing for different common factors across states. For example, with respect to the estimation of the threshold, we find a remarkable similarity in its point estimate across the two datasets employed, with the estimate being around -6 percent. This estimate is twice as large in absolute value as the most common value imposed in estimation by researchers, which is -3 percent (see Farago and Tédongap (2018); and also, albeit implicitly, Lettau, Maggiori, and Weber (2014)), thereby implying that the downside risk state is characterized by a

smaller fraction of the return distribution with more extreme negative returns. We also find that, while up to six common factors are needed in the satisfactory (or good) state of the world to capture the factor structure of returns, only one common factor is needed in the downside risk state. This is consistent with the widely held view, and much anecdotal evidence, that returns display a far lower-dimensional factor structure in bad times than in good times, i.e. that diversification benefits diminish in bad times when they are needed most (see e.g. Cappiello, Engle, and Sheppard (2006)). This result arises endogenously in the model, estimated using our procedure that explicitly allows the factors to change across states. Finally, the model uncovers the existence of highly asymmetric dynamics in risk premia, with several factors displaying quite different risk premia estimates from those obtained from a linear asset pricing model. In short, the empirical results provide a strong case for the full-blown, unconstrained estimation of latent factor models with downside risk in the context of asset pricing.

Beyond the aforementioned contributions, our paper relates to (at least) two additional strands of literature. The first one studies asset pricing with asymmetric pricing of "good" and "bad" covolatility: in a continuous time framework, Bollerslev, Li, and Todorov (2016) show that continuous and jump CAPM betas are priced in a different way; Bollerslev, Patton, and Quaedvlieg (2021) decompose the market beta into four semibetas, which depend on the covariation between the return on the market and that on individual assets; Bollerslev (2020) provides a comprehensive overview of contributions on pricing "good" and "bad" realized volatilities. Our paper is also linked to the literature on consumption-based asset pricing with disappointment aversion: Delikouras (2017) develops a model in which disappointment events take place when lifetime utility falls below its certainty equivalent; Delikouras and Kostakis (2019) consider the more general set up in which the disappointment threshold is a multiple of the certainty equivalent.

The remainder of the paper is organized as follows. Section 2 introduces the asset pricing model. Section 3 deals with identification of risk premia. Section 4 presents the methodology to determine the number of pricing factors. Section 5 discusses estimation and inference of pricing factors, risk exposures and risk premia. Section 6 performs the empirical analysis. Section 7 concludes. In the paper, we only present the main results for the sake of a concise discussion; assumptions, simulations, technical lemmas and proofs are all relegated to the Internet Appendix.

2 Model

In this section, we first present, for convenience, the setup of the unconditional asset pricing model that is most commonly used in the literature (Section 2.1). We then discuss the conditional asset pricing model with downside risk, whose estimation is the main contribution of this paper (Section 2.2).

At the outset, it is useful to establish some notation. We denote the ordinary limit as " \rightarrow ". $\mathbb{I}(\cdot)$ denotes the indicator function. We use ι_N to denote an N-dimensional vector of ones; I_k is an identity matrix of dimension k; finally, we set $C_{N,T} = \min \{N^{1/2}, T^{1/2}\}$.

2.1 Unconditional asset pricing model

Let $R_{i,t}$ be the return (in excess of the risk-free rate) on the test asset *i* at time *t*. The linear, or unconditional, asset pricing model can be written as

$$R_{i,t} = \gamma_0 + \alpha_i + \beta'_i \gamma_1 + \beta'_i \mathbf{u}_t + \epsilon_{i,t}, \tag{1}$$

with $1 \le i \le N$, $1 \le t \le T$, where N and T denote the cross-section and time series dimensions of the available sample, respectively, and

$$\mathbf{u}_t = \mathbf{f}_t - \boldsymbol{\mu}_f. \tag{2}$$

In (2), $\mathbf{f}_t = (f_{1,t}, \dots, f_{P,t})'$ is a $P \times 1$ vector of latent factors with $E(\mathbf{f}_t) = \boldsymbol{\mu}_f$, so that \mathbf{u}_t in (1) represents the $P \times 1$ zero mean vector of factor innovations; further, in (1), γ_0 is the zero-beta rate; $\boldsymbol{\gamma}_1 = (\gamma_{1,1}, \dots, \gamma_{1,P})'$ is the $P \times 1$ vector of risk premia; $\boldsymbol{\beta}_i = (\beta_{i,1}, \dots, \beta_{i,P})'$ is the $P \times 1$ vector of risk exposures; α_i is the pricing error; $\epsilon_{i,t}$ is the idiosyncratic component such that $E(\epsilon_{i,t}) = 0$ and $E(\mathbf{u}_t \epsilon_{i,t}) = 0$. In matrix notation, the model becomes

$$\mathbf{R}_t = \gamma_0 \boldsymbol{\iota}_N + \boldsymbol{\alpha} + \mathbf{B} \gamma_1 + \mathbf{B} \mathbf{u}_t + \boldsymbol{\epsilon}_t, \tag{3}$$

where $\mathbf{R}_t = (R_{1t}, \ldots, R_{Nt})', \, \boldsymbol{\alpha} = (\alpha_1, \ldots, \alpha_N)', \text{ and } \mathbf{B} = (\boldsymbol{\beta}_1, \ldots, \boldsymbol{\beta}_N)'.$

Following Giglio and Xiu (2021), we let some of the true factors be omitted or measured with error (or both). In particular, the K observable factors \mathbf{g}_t can be either tradable or nontradable (or both), and have the structure

$$\mathbf{g}_t = \mathbf{a} + \mathbf{\Lambda} \mathbf{u}_t + \mathbf{e}_t. \tag{4}$$

As shown in Giglio and Xiu (2021), given γ_1 , the $K \times 1$ vector of risk premia of \mathbf{g}_t is $\gamma_{\mathbf{g}} = \mathbf{\Lambda} \gamma_1$.

2.2 Conditional asset pricing model with downside risk

The conditional factor model of interest is:

$$R_{i,t} = \mathbb{I}(\mathcal{D}_{t})\left(\gamma_{\mathcal{D},0} + \alpha_{\mathcal{D},i} + \beta'_{\mathcal{D},i}\gamma_{\mathcal{D},1} + \beta'_{\mathcal{D},i}\mathbf{u}_{\mathcal{D},t}\right) + \mathbb{I}(\mathcal{S}_{t})\left(\gamma_{\mathcal{S},0} + \alpha_{\mathcal{S},i} + \beta'_{\mathcal{S},i}\gamma_{\mathcal{S},1} + \beta'_{\mathcal{S},i}\mathbf{u}_{\mathcal{S},t}\right) + \epsilon_{i,t},$$
(5)

where

$$\mathbf{u}_{j,t} = \mathbf{f}_{j,t} - \boldsymbol{\mu}_{j,f}, \quad j = \mathcal{D}, \mathcal{S}.$$
(6)

The model in (5) generalizes (1) to allow for downside risk, as captured by the disappointment event \mathcal{D}_t , which is defined formally below. The satisfactory event \mathcal{S}_t occurs whenever \mathcal{D}_t does not take place, i.e. $\mathbb{I}(\mathcal{D}_t) + \mathbb{I}(\mathcal{S}_t) = 1$ and $\mathbb{I}(\mathcal{D}_t)\mathbb{I}(\mathcal{S}_t) = 0$. Section 2.3 below discusses \mathcal{D}_t and \mathcal{S}_t in detail. In (6), $\mathbf{f}_{j,t} = (f_{j,1,t}, \ldots, f_{j,P_j,t})'$ is the $P_j \times 1$ vector of latent factors satisfying $E[\mathbf{f}_{j,t} | \mathbb{I}(j_t) = 1] = \mu_{j,f}; \mathbf{u}_{j,t}$ is the $P_j \times 1$ vector of factor innovations such that $E[\mathbf{u}_{jt} | \mathbb{I}(j_t) = 1] =$ 0. As for (5): $\gamma_{j,0}$ is the zero-beta rate and $\boldsymbol{\gamma}_{j,1} = (\gamma_{j,1,1}, \ldots, \gamma_{j,1,P_j})'$ is the $P_j \times 1$ vector of risk premia; $\boldsymbol{\beta}_{j,i} = (\beta_{j,i,1}, \ldots, \beta_{j,i,P_j})'$ is the $P_j \times 1$ vector of risk exposures; $\alpha_{j,i}$ is the pricing error; the idiosyncratic components $\epsilon_{i,t}$ satisfy $E[\mathbf{u}_{j,t}\epsilon_{i,t} | \mathbb{I}(j_t) = 1] = 0$. The model in matrix form is

$$\mathbf{R}_{t} = \mathbb{I}(\mathcal{D}_{t}) \left(\gamma_{\mathcal{D},0} \boldsymbol{\iota}_{N} + \boldsymbol{\alpha}_{\mathcal{D}} + \mathbf{B}_{\mathcal{D}} \boldsymbol{\gamma}_{\mathcal{D},1} + \mathbf{B}_{\mathcal{D}} \mathbf{u}_{\mathcal{D},t} \right) + \mathbb{I}\left(\mathcal{S}_{t} \right) \left(\gamma_{\mathcal{S},0} \boldsymbol{\iota}_{N} + \boldsymbol{\alpha}_{\mathcal{S}} + \mathbf{B}_{\mathcal{S}} \boldsymbol{\gamma}_{\mathcal{S},1} + \mathbf{B}_{\mathcal{S}} \mathbf{u}_{\mathcal{S},t} \right) + \boldsymbol{\epsilon}_{t},$$
(7)

where $\boldsymbol{\alpha}_j = (\alpha_{j,1}, \dots, \alpha_{j,N})'$, and $\mathbf{B} = (\boldsymbol{\beta}_{j,1}, \dots, \boldsymbol{\beta}_{j,N})'$. We generalize the mapping between the true K latent factors, and observable factors as

$$\mathbf{g}_{t} = \mathbb{I}\left(\mathcal{D}_{t}\right)\left(\mathbf{a}_{\mathcal{D}} + \boldsymbol{\Lambda}_{\mathcal{D}}\mathbf{u}_{\mathcal{D},t}\right) + \mathbb{I}\left(\mathcal{S}_{t}\right)\left(\mathbf{a}_{\mathcal{S}} + \boldsymbol{\Lambda}_{\mathcal{S}}\mathbf{u}_{\mathcal{S},t}\right) + \mathbf{e}_{t}.$$
(8)

The $K \times 1$ vector of risk premia of \mathbf{g}_t becomes

$$\boldsymbol{\gamma}_{\mathbf{g}} = \mathbb{I}\left(\mathcal{D}_{t}\right)\boldsymbol{\gamma}_{\mathcal{D},\mathbf{g}} + \mathbb{I}\left(\mathcal{S}_{t}\right)\boldsymbol{\gamma}_{\mathcal{S},\mathbf{g}}, \quad \boldsymbol{\gamma}_{j,\mathbf{g}} = \boldsymbol{\Lambda}_{j}\boldsymbol{\gamma}_{j,1}, \quad j = \mathcal{D}, \mathcal{S}.$$
(9)

When the disappointment event occurs, we thus have $\gamma_{\mathbf{g}} = \gamma_{\mathcal{D},\mathbf{g}} = \Lambda_{\mathcal{D}} \gamma_{\mathcal{D},1}$.

2.3 Disappointment event

Following, inter alia, Farago and Tédongap (2018), we define the disappointment event as $\mathcal{D}_t = \{r_{W,t} \leq \theta\}$, where $r_{W,t}$ is the log-return on the market and θ is the corresponding threshold value that determines the occurrence of \mathcal{D}_t . The satisfactory event then is $\mathcal{S}_t = \{r_{W,t} > \theta\}$. Both \mathcal{D}_t and \mathcal{S}_t depend on θ : to stress this dependence, we write $d_{j,t}(\theta) = \mathbb{I}(j_t)$, for $j = \mathcal{D}, \mathcal{S}^2$.

3 Identification

3.1 Identification of risk premia

In the conditional factor model in (7), there are two kinds of identification issues: (i) the rotational indeterminacy typical of statistical factor models; and (ii) the unobserved heterogeneity problem induced by the pricing errors.

Consider (i): risk exposures, pricing factors and risk premia in the linear model in (3) are

²The conditional asset pricing model in (7) allows for unrestricted zero-beta rates $\gamma_{\mathcal{D},0}$ and $\gamma_{\mathcal{S},0}$, and for nonzero pricing errors $\alpha_{\mathcal{D}}$ and $\alpha_{\mathcal{S}}$. None of these invalidates our methodology. In particular, non-zero pricing errors, which for instance may arise from failing no arbitrage conditions, do not affect estimation and inference on the risk premia.

identified up to a rotation. Indeed, for any $P \times P$ positive definite matrix **L** we have

$$egin{array}{rcl} \mathbf{R}_t &=& \gamma_0 oldsymbol{\iota}_N + oldsymbol{lpha} + \mathbf{B} oldsymbol{\gamma}_1 + \mathbf{B} \mathbf{u}_t + oldsymbol{\epsilon}_t \ &=& \gamma_0 oldsymbol{\iota}_N + oldsymbol{lpha} + \mathbf{B} \mathbf{L} \mathbf{L}^{-1} oldsymbol{\gamma}_1 + \mathbf{B} \mathbf{L} \mathbf{L}^{-1} \mathbf{u}_t + oldsymbol{\epsilon}_t \end{array}$$

Hence, **B**, γ_1 and \mathbf{u}_t are only identified up to **L**, as they are all unobservable. However, as shown in Giglio and Xiu (2021), the risk premia for the observable factors \mathbf{g}_t in (4) are identified, since it holds that $\gamma_{\mathbf{g}} = \mathbf{\Lambda} \gamma_1 = \mathbf{\Lambda} \mathbf{L} \mathbf{L}^{-1} \gamma_1$.

The conditional pricing model in (5) faces essentially the same identification features. For any $P_j \times P_j$ positive definite matrix \mathbf{L}_j , with $j = \mathcal{D}, \mathcal{S}$, we have

$$\mathbf{R}_{t} = d_{\mathcal{D},t}(\theta) \left(\gamma_{\mathcal{D},0} \boldsymbol{\iota}_{N} + \boldsymbol{\alpha}_{\mathcal{D}} + \mathbf{B}_{\mathcal{D}} \mathbf{L}_{\mathcal{D}} \mathbf{L}_{\mathcal{D}}^{-1} \boldsymbol{\gamma}_{\mathcal{D},1} + \mathbf{B}_{\mathcal{D}} \mathbf{L}_{\mathcal{D}} \mathbf{L}_{\mathcal{D}}^{-1} \mathbf{u}_{\mathcal{D},t} \right) \\ + d_{\mathcal{S},t}(\theta) \left(\gamma_{\mathcal{S},0} \boldsymbol{\iota}_{N} + \boldsymbol{\alpha}_{\mathcal{S}} + \mathbf{B}_{\mathcal{S}} \mathbf{L}_{\mathcal{S}} \mathbf{L}_{\mathcal{S}}^{-1} \boldsymbol{\gamma}_{\mathcal{S},1} + \mathbf{B}_{\mathcal{S}} \mathbf{L}_{\mathcal{S}} \mathbf{L}_{\mathcal{S}}^{-1} \mathbf{u}_{\mathcal{S},t} \right) + \boldsymbol{\epsilon}_{t}$$

The risk premia $\gamma_{\mathbf{g}}$ defined in (9), however, are identified, since

$$\begin{aligned} \boldsymbol{\gamma}_{\mathbf{g}} &= d_{\mathcal{D},t}(\theta) \boldsymbol{\Lambda}_{\mathcal{D}} \boldsymbol{\gamma}_{\mathcal{D},1} + d_{\mathcal{S},t}(\theta) \boldsymbol{\Lambda}_{\mathcal{S}} \boldsymbol{\gamma}_{\mathcal{S},1} \\ &= d_{\mathcal{S},t}(\theta) \boldsymbol{\Lambda}_{\mathcal{D}} \mathbf{L}_{\mathcal{D}} \mathbf{L}_{\mathcal{D}}^{-1} \boldsymbol{\gamma}_{\mathcal{D},1} + d_{\mathcal{S},t}(\theta) \boldsymbol{\Lambda}_{\mathcal{S}} \mathbf{L}_{\mathcal{S}} \mathbf{L}_{\mathcal{S}}^{-1} \boldsymbol{\gamma}_{\mathcal{S},1} \end{aligned}$$

As far as the second identification issue is concerned, we solve the unobserved heterogeneity problem induced by the pricing errors by expressing the model for \mathbf{R}_t in terms of deviations from the conditional means. Formally, let $T_j(\theta) = \sum_{t=1}^T d_{j,t}(\theta)$ be the number of times that the event j occurs, for $j = \mathcal{D}, \mathcal{S}$. Define the conditional average returns $\overline{\mathbf{R}}_j(\theta) = T_j(\theta)^{-1} \sum_{t=1}^T d_{j,t}(\theta) \mathbf{R}_t$: $\overline{\mathbf{R}}_j(\theta)$ is the mean of \mathbf{R}_t when j occurs. In order to estimate pricing factors and risk exposures, we consider the model

$$\widetilde{\mathbf{R}}_{t}(\theta) = \mathbf{R}_{t} - d_{\mathcal{D},t}(\theta)\overline{\mathbf{R}}_{\mathcal{D}}(\theta) - d_{\mathcal{S},t}(\theta)\overline{\mathbf{R}}_{\mathcal{S}}(\theta) = d_{\mathcal{D},t}(\theta)\mathbf{B}_{\mathcal{D}}\widetilde{\mathbf{u}}_{\mathcal{D},t}(\theta) + d_{\mathcal{S},t}(\theta)\mathbf{B}_{\mathcal{S}}\widetilde{\mathbf{u}}_{\mathcal{S}t}(\theta) + \tilde{\boldsymbol{\epsilon}}_{t}$$
(10)

where $\widetilde{\mathbf{u}}_{j,t}(\theta) = \mathbf{u}_t - \overline{\mathbf{u}}_j(\theta)$ is the deviation of \mathbf{u}_t from the conditional mean $\overline{\mathbf{u}}_j(\theta) = T_j(\theta)^{-1} \sum_{t=1}^T d_{j,t}(\theta) \mathbf{u}_t$; and $\widetilde{\boldsymbol{\epsilon}}_t = d_{\mathcal{D},t}(\theta) [\boldsymbol{\epsilon}_t - \overline{\boldsymbol{\epsilon}}_{\mathcal{D}}(\theta)] + d_{\mathcal{S},t}(\theta) [\boldsymbol{\epsilon}_t - \overline{\boldsymbol{\epsilon}}_{\mathcal{S}}(\theta)$, where $\overline{\boldsymbol{\epsilon}}_j(\theta) = T_j(\theta)^{-1} \sum_{t=1}^T d_{j,t}(\theta) \boldsymbol{\epsilon}_{j,t}$ is the conditional mean of ϵ_t when j occurs.

3.2 The disappointment threshold

As implicitly noted in Section 3.1, the identification of factors, exposures and risk premia requires knowledge of the disappointment event \mathcal{D}_t , and thus of the threshold θ . Typically, the literature pre-specifies a value for θ , and the subsequent estimation of the asset pricing model is then conditioned on that value being true.³ In general, however, θ is not a structural parameter on which theory can provide guidance, which makes it difficult to assume a particular value of θ . In this paper, we study the Least Squares estimator of θ (Section 4.1).

4 Inference

In this section, we study inference on (5): we begin by presenting the estimation and hypothesis testing on θ (Section 4.1); we then discuss the estimation of the number of common factors in each regime (Section 4.2); and, finally, we present the Principal Component estimation of $\beta_{j,i}$ and $\gamma_{j,1}$ (Section 4.3).

We begin by introducing some further notation. The superscript "0" denotes the true value of a parameter; when this is not used, we refer to a generic value. We also use the short-hand notation $T_j = T_j(\theta^0)$ and $\pi_j = \pi_j(\theta^0)$; further, the true number of factors in each regime is denoted as P_D^0 and P_S^0 respectively, also using, when needed, $P^0 = P_D^0 + P_S^0$. Finally, here and throughout the whole paper, P^* denotes the probability conditional on $\{R_{i,t}, 1 \le i \le N, 1 \le t \le T\}$; we use " $\overset{D^*}{\rightarrow}$ " and " $\overset{P^*}{\rightarrow}$ " to indicate convergence in distribution and in probability according to P^* respectively.

4.1 Estimating θ^0

We begin by studying the estimation of θ^0 . In Section C in the Internet Appendix, we show that the estimator is strongly consistent.

³For example, as mentioned in the introduction, Farago and Tédongap (2018) *a priori* set $\theta = -0.03$ and run robustness checks for $\theta \in \{0, -0.015, -0.04\}$.

Some further notation is required. Let the $N \times 1$ vector of demeaned returns be $\widetilde{\mathbf{R}}_t = (\widetilde{R}_{1,t}, \ldots, \widetilde{R}_{N,t})'$, for $1 \leq t \leq T$, defined as in (10). For generic numbers of factors $P_{\mathcal{D}}$, $P_{\mathcal{S}}$ and $P = P_{\mathcal{D}} + P_{\mathcal{S}}$, define the $N \times P$ matrix of loadings $\mathbf{B}^P = (\mathbf{B}_{\mathcal{D}}^{P_{\mathcal{D}}}, \mathbf{B}_{\mathcal{S}}^{P_{\mathcal{S}}})$, and the associated $P_{\mathcal{D}} \times T$ and $P_{\mathcal{S}} \times T$ matrices of demeaned factor innovations $\mathbf{U}_{\mathcal{D}}^{P_{\mathcal{D}}} = (\mathbf{u}_{\mathcal{D},1}^{P_{\mathcal{D}}}, \ldots, \mathbf{u}_{\mathcal{D},T}^{P_{\mathcal{D}}})$ and $\mathbf{U}_{\mathcal{S}}^{P_{\mathcal{S}}} = (\mathbf{u}_{\mathcal{S},1}^{P_{\mathcal{S}}}, \ldots, \mathbf{u}_{\mathcal{S},T}^{P_{\mathcal{S}}})$, with $\mathbf{U}^P = (\mathbf{U}_{\mathcal{D}}^{P_{\mathcal{D}}'}, \mathbf{U}_{\mathcal{S}}^{P_{\mathcal{S}}'})'$.

The objective function in terms of \mathbf{B}^{P} , \mathbf{U}^{P} and θ , is given by the sum of squared residuals (divided by NT):

$$S\left(\mathbf{B}^{P},\mathbf{U}^{P},\theta\right) = \frac{1}{NT}\sum_{t=1}^{T}\left\|\widetilde{\mathbf{R}}_{t} - d_{\mathcal{D},t}\left(\theta\right)\mathbf{B}_{\mathcal{D}}^{P_{\mathcal{D}}}\mathbf{u}_{\mathcal{D},t}^{P_{\mathcal{D}}} - d_{\mathcal{S},t}\left(\theta\right)\mathbf{B}_{\mathcal{S}}^{P_{\mathcal{S}}}\mathbf{u}_{\mathcal{S},t}^{P_{\mathcal{S}}}\right\|^{2}.$$
(11)

For given P, the estimators $\widehat{\mathbf{B}}^P = \left(\widehat{\mathbf{B}}_{\mathcal{D}}^{P_{\mathcal{D}}}, \widehat{\mathbf{B}}_{\mathcal{S}}^{P_{\mathcal{S}}}\right)$ - with $\widehat{\mathbf{B}}_j^{P_j} = \left(\widehat{\beta}_{j,1}, ..., \widehat{\beta}_{j,N}\right)'$ for $j = \mathcal{D}, \mathcal{S}$, and $\widehat{\mathbf{U}}^P$ and $\widehat{\theta}^P$ are defined as

$$\left\{\widehat{\mathbf{B}}^{P}, \widehat{\mathbf{U}}^{P}, \widehat{\theta}^{P}\right\} = \arg\min_{\mathbf{B}^{P}, \mathbf{U}^{P}, \theta} S\left(\mathbf{B}^{P}, \mathbf{U}^{P}, \theta\right).$$

In order to solve this minimization, we begin by concentrating the objective function $S\left(\mathbf{B}^{P}, \mathbf{U}^{P}, \theta\right)$ with respect to \mathbf{B}^{P} and θ , subject to the constraints $N^{-1}\mathbf{B}_{j}^{P_{j}'}\mathbf{B}_{j}^{P_{j}} = I_{P_{j}}$ for $j = \mathcal{D}, \mathcal{S}$. It follows that the estimator of \mathbf{U}^{P} is

$$\widehat{\mathbf{u}}_{\mathcal{D},t}^{P_{\mathcal{D}}}\left(\mathbf{B}_{\mathcal{D}}^{P_{\mathcal{D}}},\theta\right) = N^{-1} \left(d_{\mathcal{D},t}\left(\theta\right)\mathbf{B}_{\mathcal{D}}^{P_{\mathcal{D}}}\left(\theta\right)\right)' \widetilde{\mathbf{R}}_{t},\\ \widehat{\mathbf{u}}_{\mathcal{S},t}^{P_{\mathcal{S}}}\left(\mathbf{B}_{\mathcal{S}}^{P_{\mathcal{S}}},\theta\right) = N^{-1} \left(d_{\mathcal{S},t}\left(\theta\right)\mathbf{B}_{\mathcal{S}}^{P_{\mathcal{S}}}\left(\theta\right)\right)' \widetilde{\mathbf{R}}_{t}.$$

Hence, we can define

$$S\left(\mathbf{B}^{P},\theta\right) = \frac{1}{NT}\sum_{t=1}^{T} \left\|\widetilde{\mathbf{R}}_{t} - d_{\mathcal{D},t}\left(\theta\right)\mathbf{B}_{\mathcal{D}}^{P_{\mathcal{D}}}\widehat{\mathbf{u}}_{\mathcal{D},t}^{P_{\mathcal{D}}}\left(\mathbf{B}_{\mathcal{D}}^{P_{\mathcal{D}}},\theta\right) - d_{\mathcal{S},t}\left(\theta\right)\mathbf{B}_{\mathcal{S}}^{P_{\mathcal{S}}}\widehat{\mathbf{u}}_{\mathcal{S},t}^{P_{\mathcal{S}}}\left(\mathbf{B}_{\mathcal{S}}^{P_{\mathcal{S}}},\theta\right)\right\|^{2}$$

Let

$$\widehat{\boldsymbol{\Sigma}}_{j,\widetilde{\mathbf{R}}}\left(\boldsymbol{\theta}\right) = \frac{1}{NT} \sum_{t=1}^{T} d_{j,t}\left(\boldsymbol{\theta}\right) \widetilde{\mathbf{R}}_{t}\left(\boldsymbol{\theta}\right) \widetilde{\mathbf{R}}_{t}\left(\boldsymbol{\theta}\right)^{\prime},\tag{12}$$

for $j = \mathcal{D}, \mathcal{S}$. It is well known (e.g. Bai (2003)) that the Principal Component estimator $\widehat{\mathbf{B}}_{j}^{P_{j}}(\theta)$ for $\mathbf{B}_{j}^{P_{j}}$ given θ is \sqrt{N} times the $N \times P_{j}$ matrix of eigenvectors of $\widehat{\boldsymbol{\Sigma}}_{j,\widetilde{\mathbf{R}}}(\theta)$ corresponding to its largest P_j eigenvalues, under the orthonormalization restriction $N^{-1}\left(\widehat{\mathbf{B}}_j^{P_j}(\theta)' \widehat{\mathbf{B}}_j^{P_j}(\theta)\right) = \mathbf{I}_{P_j}$, for $j = \mathcal{D}, \mathcal{S}$.

We now define

$$S\left(\theta\right) = \frac{1}{NT} \sum_{t=1}^{T} \left\| \widetilde{\mathbf{R}}_{t} - d_{\mathcal{D},t}\left(\theta\right) \widehat{\mathbf{B}}_{\mathcal{D}}^{P_{\mathcal{D}}}\left(\theta\right) \widehat{\mathbf{u}}_{\mathcal{D},t}^{P_{\mathcal{D}}}\left(\widehat{\mathbf{B}}_{\mathcal{D}}^{P_{\mathcal{D}}}\left(\theta\right);\theta\right) - d_{\mathcal{S},t}\left(\theta\right) \widehat{\mathbf{B}}_{\mathcal{S}}^{P_{\mathcal{S}}}\left(\theta\right) \widehat{\mathbf{u}}_{\mathcal{S},t}^{P_{\mathcal{S}}}\left(\widehat{\mathbf{B}}_{\mathcal{S}}^{P_{\mathcal{S}}}\left(\theta\right);\theta\right) \right\|^{2}.$$

Let $\widehat{\mathbf{B}}^{P}(\theta) = \left[\widehat{\mathbf{B}}_{\mathcal{D}}^{P_{\mathcal{D}}}(\theta), \widehat{\mathbf{B}}_{\mathcal{S}}^{P_{\mathcal{S}}}(\theta)\right]$ and $\widehat{\mathbf{U}}^{P}(\theta) = \left[\widehat{\mathbf{U}}_{\mathcal{D}}^{P_{\mathcal{D}}'}\left(\widehat{\mathbf{B}}_{\mathcal{D}}^{P_{\mathcal{D}}}(\theta);\theta\right), \widehat{\mathbf{U}}_{\mathcal{S}}^{P_{\mathcal{S}}'}\left(\widehat{\mathbf{B}}_{\mathcal{S}}^{P_{\mathcal{S}}}(\theta);\theta\right)\right]'$. For $\overline{P} = \overline{P}_{\mathcal{D}} + \overline{P}_{\mathcal{S}}$ with $\overline{P}_{\mathcal{D}} \ge P_{\mathcal{D}}^{0}$ and $\overline{P}_{\mathcal{S}} \ge P_{\mathcal{S}}^{0}$, we finally have the estimator

$$\widehat{\theta} = \arg\min_{\theta} S\left[\widehat{\mathbf{B}}^{\overline{P}}\left(\theta\right), \widehat{\mathbf{U}}^{\overline{P}}\left(\theta\right), \theta\right];$$
(13)

 $\hat{\theta}$ can be obtained e.g. through a grid search. In Section B of the Internet Appendix, we complement the theory with a small-scale simulation exercise, reporting bias and MSE for the estimated θ^0 . We find that the estimator works well in general, especially when the factor structure of the two regimes differs (more); the estimates tend to be biased towards zero as θ^0 becomes more and more negative (thus having fewer and fewer observations in the downside regime), but this vanishes as the sample sizes increase.

In the Internet Appendix, we show that $\hat{\theta}$ is a strongly consistent estimator of θ^0 , deriving its rate. In principle, the limiting distribution of $\hat{\theta}$ can be derived along similar steps as Chan (1993), and can be employed to carry out inference on θ^0 . However, this is bound to depend on several nuisance parameters, and therefore the asymptotic distribution of $\hat{\theta}$ would be only of theoretical interest. Hence, in this paper we propose a different approach to test for null hypotheses of the type

$$H_0: \theta^0 = c, \tag{14}$$

where c is a value of interest; we note that being able to test for (14) also offers the possibility of computing confidence intervals, by inverting the test. This procedure does not require the limiting distribution of $\hat{\theta}$, and it relies only on the rate of convergence, using an approach similar to the one in Horváth and Trapani (2019). In particular, our approach is very similar to the one proposed in Section 4.2, and therefore, in order to avoid repetitions, the details are relegated to Section A.2 of the Internet Appendix, to which we refer the interested reader.

4.2 Determining the number of common factors

We extend the procedure suggested by Trapani (2018) to estimate the number of factors in each regime, $P_{\mathcal{D}}^0$ and $P_{\mathcal{S}}^0$. As in the previous section, here we only report the relevant algorithm; details and the theory are in Section A.3 of the Internet Appendix.

Let $\hat{d}_{\mathcal{D},t} = \mathbb{I}\left(\hat{\mathcal{D}}_{t}\right)$ and $\hat{d}_{\mathcal{S},t} = 1 - \hat{d}_{\mathcal{D},t}$, and define $\hat{\pi}_{\mathcal{S}} = T^{-1} \sum_{t=1}^{T} \hat{d}_{\mathcal{D},t}$ and $\hat{\pi}_{\mathcal{D}} = 1 - \hat{\pi}_{\mathcal{S}}$. We use the sample covariance matrices

$$\widehat{\Sigma}_{\mathcal{D}} = \frac{1}{T\widehat{\pi}_{\mathcal{D}}} \sum_{t=1}^{T} \widetilde{\mathbf{R}}_{t} \widetilde{\mathbf{R}}_{t}' \widehat{d}_{\mathcal{D},t}, \qquad (15)$$

$$\widehat{\Sigma}_{\mathcal{S}} = \frac{1}{T\widehat{\pi}_{\mathcal{S}}} \sum_{t=1}^{T} \widetilde{\mathbf{R}}_{t} \widetilde{\mathbf{R}}_{t}' \widehat{d}_{\mathcal{S},t}, \qquad (16)$$

and denote the *i*-th largest eigenvalue of $\widehat{\Sigma}_{\mathcal{D}}$ and $\widehat{\Sigma}_{\mathcal{S}}$ as $\widehat{g}_{j}^{(i)}$, with $j = \mathcal{D}$ or \mathcal{S} .

Using Lemma A.1 and Theorem A.2 of Section A.3 in the Internet Appendix, we can determine the number of common factors in each regime using the following algorithm, based on two separate steps.

In the first step, we study a test for the individual eigenvalues, i.e.

$$\begin{cases} H_0: \quad g_j^{(i)} = c_j^{(i)} N \\ H_A: \quad g_j^{(i)} \le c_j^{(i)} \end{cases}, \tag{17}$$

for some $0 < c_j^{(i)} < \infty$ and $j = \mathcal{D}, \mathcal{S}$. In essence, the null hypothesis is that the *i*-th largest eigenvalue of the covariance matrix of the data - in each regime - diverges, thus suggesting that there are at least *i* common factors. Conversely, upon rejecting the null hypothesis, the conclusion can be drawn that there are fewer than *i* common factors in regime *j*. Thus, the second step of the analysis is to determine $P_{\mathcal{D}}^0$ and $P_{\mathcal{S}}^0$ by carrying out a sequence of tests for $i = 1, ..., P_{\text{max}}$, with P_{max} a user-defined upper bound. We begin with (17). Define

$$\psi\left(\widehat{g}_{j}^{(i)}\right) = \exp\left(\frac{N^{-\varrho_{j}}\widehat{g}_{j}^{(i)}}{\overline{g}_{j}\left(p\right)}\right),\tag{18}$$

where

$$\overline{g}_j(p) = \frac{1}{N-p} \sum_{h=p+1}^N \widehat{g}_j^{(h)}.$$

The purpose of $\overline{g}_j(p)$ is to rescale $\hat{g}_j^{(i)}$; in Trapani (2018), it is suggested to set p = 0 (that is, to use the trace to rescale $\hat{g}_j^{(i)}$); other choices are also possible, such as p = i, which is recommended in Barigozzi and Trapani (2021). In (18), we have defined

$$\varrho_j = \begin{cases} \varepsilon & \text{when } \frac{\ln T_j}{\ln N} \ge \frac{1}{2} \\ 1 - \frac{1}{2} \frac{\ln T_j}{\ln N} + \varepsilon & \text{when } \frac{\ln T_j}{\ln N} < \frac{1}{2} \end{cases},$$
(19)

where $\varepsilon > 0$ is a (small) user-defined number.

We now describe the test for (17).

Step 1 Generate an *i.i.d.*, N(0, 1) sequence $\{\xi_{j,m}^{(i)}, 1 \le m \le M\}$, where the $\xi_{j,m}^{(i)}$ s are independent across *i* and *j*.

Step 2 Define the Bernoulli sequence $\zeta_{j,m}^{(i)}(s) = \mathbb{I}\left(\psi\left(\widehat{g}_{j}^{(i)}\right) \times \xi_{i,m}^{(j)} \leq s\right).$

Step 3 Compute

$$\Upsilon_{j}^{(i)} = \int_{-\infty}^{+\infty} \left(v_{j}^{(i)}\left(s\right) \right)^{2} \frac{1}{\sqrt{2\pi}} \exp\left(-\frac{1}{2}s^{2}\right) ds,$$

where

$$v_{j}^{(i)}(s) = \frac{2}{M^{1/2}} \sum_{m=1}^{M} \left(\zeta_{j,m}^{(i)}(s) - \frac{1}{2} \right)$$

We show that, under the null, $\Upsilon_j^{(i)} \xrightarrow{D^*} \chi_1^2$, thus providing a rule to decide between H_0 and H_A in (17). On account of this, it is possible to propose a sequential procedure to determine $P_{\mathcal{D}}^0$ and $P_{\mathcal{S}}^0$.

This is based on the following two-step algorithm, for $j = \mathcal{D}, \mathcal{S}$.

- **Step 1** Run the test for $H_0: g_j^{(i)} = \infty$ based on $\Upsilon_j^{(1)}$. If the null is rejected, set $\hat{P}_j = 0$ and stop, otherwise go to the next step.
- Step 2 Starting from i = 1, run the test for $H_0 : g_j^{(i+1)} = \infty$ based on $\Upsilon_j^{(i+1)}$. If the null is rejected, set $\hat{P}_j = i$ and stop; otherwise repeat the step until the null is rejected, or until the pre-specified maximum number P_{max} is reached.

Let $\alpha = \alpha(N,T)$ denote the level of the individual tests and $c_{\alpha} = c_{\alpha}(N,T)$ be the corresponding critical value. We show that, as long as the level of the individual tests goes to zero, \hat{P}_j is a consistent estimator.

Theorem 1. We assume that Assumptions 1-6 in Section A.1 of the Internet Appenendix are satisfied. As $\min(N, T_j) \to \infty$ under (A.11), if $c_{\alpha}(N, T) \to \infty$ with $c_{\alpha}(N, T) = o(M)$, then it holds that

$$\widehat{P}_j = P_j^0 + o_{P^*}\left(1\right),$$

for almost all realizations of $\{R_{i,t}, 1 \leq i \leq N, 1 \leq t \leq T\}$, and $j = \mathcal{D}, \mathcal{S}$.

In the implementation of the algorithm, it is necessary to have an upper bound P_{max} ; this is chosen as

$$P_{\max} = O\left(C_{N,T}^{1-c}\right),\tag{20}$$

for some user-defined c > 0. The selection rule in (20) is as an extension of the otherwise customarily employed Schwert's rule; see Schwert (1989), and also the comment in Bai and Ng (2002). In our empirical analysis, and in the simulations, we use and recommend $P_{\text{max}} = \left[\min\left\{N^{1/3}, T^{1/3}\right\}\right]$. We note that, in Section B of the Internet Appendix, we report some evidence on the performance of the estimator of the number of common factors; we have (also) tried some variants of (20) as robustness checks, but results are virtually unaffected.

4.3 Estimating common factors and loadings

Once $\hat{\theta}$ and \hat{P}_j (for $j = \mathcal{D}, \mathcal{S}$) have been obtained, estimation of factors and loadings in each regime follows by standard arguments, e.g. applying Principal Components (see e.g. Bai (2003)),

as also implicitly mentioned in Section 4.1. In particular, let

$$\widehat{\boldsymbol{\Sigma}}_{j,\widetilde{\mathbf{R}}}\left(\widehat{\theta}\right) = \frac{1}{NT} \sum_{t=1}^{T} d_{j,t}\left(\widehat{\theta}\right) \widetilde{\mathbf{R}}_{t}\left(\widehat{\theta}\right) \widetilde{\mathbf{R}}_{t}\left(\widehat{\theta}\right)',$$

for $j = \mathcal{D}, \mathcal{S}$, be an estimator of $\widehat{\Sigma}_{j,\widetilde{\mathbf{R}}}(\theta)$ defined in (12). The estimator $\widehat{\mathbf{B}}_{j}^{\widehat{P}_{j}}(\widehat{\theta})$ for $\mathbf{B}_{j}^{P_{j}}$ is given by \sqrt{N} times the $N \times \widehat{P}_{j}$ matrix of eigenvectors of $\widehat{\Sigma}_{j,\widetilde{\mathbf{R}}}(\widehat{\theta})$ corresponding to its largest \widehat{P}_{j} eigenvalues, under the orthonormalization restriction $N^{-1}\left(\widehat{\mathbf{B}}_{j}^{\widehat{P}_{j}}(\widehat{\theta})'\widehat{\mathbf{B}}_{j}^{\widehat{P}_{j}}(\widehat{\theta})\right) = \mathbf{I}_{\widehat{P}_{j}}$, for $j = \mathcal{D}, \mathcal{S}$. We then estimate $\{\mathbf{u}_{j,t}\}_{t=1}^{T}$ by OLS regression, viz.

$$\widehat{\mathbf{u}}_{j,t}^{\widehat{P}_{j}}\left(\widehat{\mathbf{B}}_{j}^{\widehat{P}_{j}}\left(\widehat{\theta}\right),\widehat{\theta}\right) = N^{-1}\left(d_{\mathcal{D},t}\left(\widehat{\theta}\right)\widehat{\mathbf{B}}_{j}^{\widehat{P}_{j}}\left(\widehat{\theta}\right)\right)'\widetilde{\mathbf{R}}_{t}$$

for $j = \mathcal{D}, \mathcal{S}$.

5 Three-pass estimation of risk premia

We now extend the Fama-MacBeth two-pass procedure (Fama and MacBeth (1973)) and estimate downside risk premia for the observable factors \mathbf{g}_t as defined in (10) in the spirit of Giglio and Xiu (2021). In particular, we propose the following three-pass procedure:

- **First-pass:** Based on the results in Section 4, estimate θ^0 , P_j^0 , $\{\mathbf{u}_{j,t}\}_{t=1}^T$, and $\{\beta_{j,i}\}_{i=1}^N$ for $j = \mathcal{D}, \mathcal{S}$: recall that the notation for the corresponding estimators is $\hat{\theta}$, \hat{P}_j , $\{\widehat{\mathbf{u}}_{j,t}^{\hat{P}_j}\}_{t=1}^T$, and $\{\widehat{\beta}_{j,i}^{\hat{P}_j}\}_{i=1}^N$ for $j = \mathcal{D}, \mathcal{S}$, respectively.
- **Second-pass:** Estimate the risk premia of $\mathbf{f}_{j,t}$ by running regime-specific cross-sectional regressions (equation (21) below).
- **Third-pass:** Estimate Λ_j within each regime by running a regression of \mathbf{g}_t on $\hat{\mathbf{u}}_{j,t}^{\widehat{P}_j}$ based on (8) (equation (22) below) and rotate the risk premia of $\mathbf{f}_{j,t}$ (equation (23) below).

We have discussed the first pass in Section 4.2, where consistency of $\hat{\theta}$ and \hat{P}_j are derived. We now turn to the second and third one. In this section, we report only the main results on the risk premia; further results are in Section A.4 of the Internet Appendix.

5.1 Regime-specific cross-sectional regression

Given the $N \times \hat{P}_j$ matrices of beta estimates $\hat{\mathbf{B}}_j^{\hat{P}_j} = \left(\hat{\beta}_{j,1}^{\hat{P}_j}, \dots, \hat{\beta}_{j,N}^{\hat{P}_j}\right)'$, define the $N \times \left(1 + \hat{P}_j\right)$ matrices $\widehat{\mathbf{X}}_j = \left(\iota_N, \widehat{\mathbf{B}}_j^{\hat{P}_j}\right)$, for $j = \mathcal{D}, \mathcal{S}$. Let $\Gamma_j = \left(\gamma_{j,0}, \gamma'_{j,1}\right)'$, for $j = \mathcal{D}, \mathcal{S}$, and $\mathbf{R}_t = (R_{1,t}, \dots, R_{N,t})'$, for $t = 1, \dots, T$.

The regime specific cross-sectional regression of \mathbf{R}_t on $\widehat{\mathbf{X}}_j$ yields

$$\widehat{\boldsymbol{\Gamma}}_{j,t} = \left(\widehat{\mathbf{X}}_{j}'\widehat{\mathbf{X}}_{j}\right)^{-1} \left(\widehat{\mathbf{X}}_{j}'\mathbf{R}_{t}\right) \widehat{d}_{j,t},\tag{21}$$

for $j = \mathcal{D}, \mathcal{S}$ and $1 \leq t \leq T$.

Further, let $\hat{T}_j = T\hat{\pi}_j$ for $j = \mathcal{D}, \mathcal{S}$. Then, Γ_j is estimated as

$$\widehat{\boldsymbol{\Gamma}}_{j} = \begin{pmatrix} \widehat{\gamma}_{j,0} \\ \widehat{\gamma}_{j,1} \end{pmatrix} = \frac{1}{\widehat{T}_{j}} \sum_{t=1}^{T} \widehat{\boldsymbol{\Gamma}}_{j,t} = \left(\widehat{\mathbf{X}}_{j}' \widehat{\mathbf{X}}_{j} \right)^{-1} \left[\widehat{\mathbf{X}}_{j}' \overline{\mathbf{R}}_{j} \left(\widehat{\boldsymbol{\theta}} \right) \right],$$

for $j = \mathcal{D}, \mathcal{S}$, where $\overline{\mathbf{R}}_j(\widehat{\theta}) = \widehat{T}_j^{-1} \sum_{t=1}^T \widehat{d}_{j,t} \mathbf{R}_t$.

5.2 Estimation of regime-specific risk premia

Let
$$\widetilde{\mathbf{g}}_{j,t} = \mathbf{g}_t \widehat{d}_{j,t}\left(\widehat{\theta}\right) - \overline{\mathbf{g}}_j$$
, with $\overline{\mathbf{g}}_j = \widehat{T}_j^{-1} \sum_{t=1}^T \mathbf{g}_t \widehat{d}_{j,t}\left(\widehat{\theta}\right)$, for $j = \mathcal{D}, \mathcal{S}$.

The estimator for Λ_j within each regime is

$$\widehat{\mathbf{\Lambda}}_{j} = \left[\sum_{t=1}^{T} \widehat{d}_{j,t}\left(\widehat{\theta}\right) \widetilde{\mathbf{g}}_{j,t} \widehat{\mathbf{u}}_{j,t}'\right] \left[\sum_{t=1}^{T} \widehat{d}_{j,t}\left(\widehat{\theta}\right) \widehat{\mathbf{u}}_{j,t} \widehat{\mathbf{u}}_{j,t}'\right]^{-1}, \qquad (22)$$

for $j = \mathcal{D}, \mathcal{S}$. Similarly, the estimator for $\gamma_{j,\mathbf{g}}$ is

$$\widehat{\gamma}_{j,\mathbf{g}} = \widehat{\Lambda}_j \widehat{\gamma}_{j,1},\tag{23}$$

for $j = \mathcal{D}, \mathcal{S}$.

5.3 Asymptotics: consistency and limiting distribution of $\hat{\gamma}_{j,g}$

We report here the asymptotics for $\hat{\gamma}_{j,\mathbf{g}}$; results for $\hat{\Gamma}_j$ and $\hat{\Lambda}_j$ are in Section A.4.3 of the Internet Appendix.

We begin by showing the consistency of our estimators.

Theorem 2. We assume that Assumptions 1-8 in Section A.1 of the Internet Appendix are satisfied. Then, as $\min(N, T_j) \to \infty$, it holds that

$$\widehat{\gamma}_{j,\mathbf{g}} - \gamma_{j,\mathbf{g}}^{0} = o_P\left(1\right) + o_{P^*}\left(1\right),$$

for $j = \mathcal{D}, \mathcal{S}$, for almost all realizations of $\{R_{i,t}, 1 \leq i \leq N, 1 \leq t \leq T\}$.

Theorem 2 states that $\hat{\gamma}_{j,\mathbf{g}}$ is consistent; no restrictions are required on the relative rate of divergence of N and T as they pass to infinity.

Next, we present the limiting distribution of the estimation error.

Theorem 3. We assume that Assumptions 1-9 in Section A.1 are satisfied, and that $\min(N, T_j) \rightarrow \infty$, with

$$\pi_j \in (0,1), \qquad (24)$$

$$\frac{T^{1/2}}{N} \to 0. \tag{25}$$

Then, it holds that

$$\left(\frac{1}{T}\Sigma_{\gamma,j} + \frac{1}{N}\Sigma_{\alpha\gamma,j}\right)^{-1/2} \left(\widehat{\gamma}_{j,\mathbf{g}} - \gamma_{j,\mathbf{g}}^{0}\right) \xrightarrow{D^{*}} N\left(0, I_{K}\right),$$
(26)

for $j = \mathcal{D}, \mathcal{S}$, for almost all realizations of $\{R_{i,t}, 1 \leq i \leq N, 1 \leq t \leq T\}$, where $\Sigma_{\gamma,j}$ and $\Sigma_{\alpha\gamma,j}$ are defined in Section A.4 of the Internet Appendix.

A comment on (25) may be in order. By Theorem 2, all estimators are consistent, irrespective of the relative rate of divergence between N and T as they pass to infinity. Theorem 3, conversely, requires a restriction, namely that $T = o(N^2)$. However, it can be envisaged that most datasets (unless T is very "short", which is typically not the case in empirical work) should satisfy this requirement.

Finally, we note that, when estimating Γ_j , we also obtain an estimator of the zero-beta rate $\gamma_{j,0}$ for $j = \mathcal{D}, \mathcal{S}$. Based on equation (5), we note that the quantity

$$E\left(\mathbb{I}\left(\mathcal{D}_{t}\right)\gamma_{\mathcal{D},0}+\mathbb{I}\left(\mathcal{S}_{t}\right)\gamma_{\mathcal{S},0}\right)=\pi_{\mathcal{D}}\gamma_{\mathcal{D}}+\pi_{\mathcal{S}}\gamma_{\mathcal{S}},$$

can be interpreted as an "unconditional" zero-beta rate. In Section A.4.4, we propose a test for

$$H_0: \pi_{\mathcal{D}}\gamma_{\mathcal{D},0} + \pi_{\mathcal{S}}\gamma_{\mathcal{S},0} = 0.$$
⁽²⁷⁾

Next, we apply the conditional factor model and the methods described above to two crosssections of asset returns.

6 Empirical analysis

6.1 Data

We apply our methodology to two sets of test assets: a cross-section of equity portfolios (Section 6.1.1), and a set of returns for multiple asset classes (Section 6.1.2).

6.1.1 Equity panel

We use returns (in excess of the risk-free rate) from the following set of N = 130 US equity portfolios: 25 portfolios sorted by size and book-to-market; 25 portfolios sorted by size and momentum; 10 size-sorted portfolios; 10 book-to-market portfolios; 10 momentum portfolios; 25 portfolios sorted by size and operating profitability; 25 portfolios sorted by size and investment. The risk-free rate is the one-month US Treasury bill rate. The market return used to determine the disappointment event (as explained in Section 2.3) is the value-weighted average log-return computed with respect to all stocks available from CRSP.

The data are taken from Kenneth French's website.⁴ We consider the sample period from July 1963 to December 2018, a total of T = 666 monthly observations. Farago and Tédongap (2018) analyze each of these sets of portfolios individually.⁵ We consider the entire cross-section of returns, which is aligned with Lewellen, Nagel, and Shanken (2010), who advocate the use of sizable cross-sections of asset returns.

6.1.2 Data for multiple asset classes

We also consider a cross section that includes equity, government bond, corporate bond and currency returns, with N = 57 portfolios in total. This allows us to analyze the effect of downside risk on multiple asset classes; Cochrane (2011) suggests investigating the factor structure across multiple asset classes to study the corresponding underlying discount factors.

The data for these test assets are obtained from He, Kelly, and Manela (2017).⁶ In particular, we employ 25 equity portfolios sorted by size and book-to-market ratio; 10 maturity-sorted government bond portfolios from the CRSP "Fama Bond Portfolios" file with maturities in six month intervals up to five years; 10 corporate bond portfolios sorted on yield spreads from Nozawa (2017); 6 currency portfolios sorted on interest rate differential (carry) from Lettau, Maggiori, and Weber (2014); 6 currency portfolios sorted on momentum from Menkhoff, Sarno, Schmeling, and Schrimpf (2012b). As for the equity panel, the market return to determine the disappointment event as detailed in Section 2.3 is the value-weighted average log-return computed with respect to all stocks available from CRSP. The data sample runs from October 1983 to January 2010, with T = 316 time series observations.

6.2 Estimation and inference

We begin by reporting results on estimation and inference on the threshold level θ (Section 6.2.1) and on the number of factors in each regime (Section 6.2.2).

 $^{^4\}mathrm{See}\ \mathtt{https://mba.tuck.dartmouth.edu/pages/faculty/ken.french/data_library.html.}$

⁵See Section 3.2 and Table 1 in Farago and Tédongap (2018).

⁶The data are kindly made available on Asaf Manela's website at http://apps.olin.wustl.edu/faculty/manela/data.html.

6.2.1 Estimation and inference on θ

Results on estimation are in Panel A of Table 1.⁷ In the case of the equity sample, the estimated disappointment threshold $\hat{\theta}$ is equal to -0.061: this generates $\hat{T}_{\mathcal{D}} = 45$ downside observations, with a relative frequency of approximately 6.8%. Notice that $\hat{\theta}$ is lower than the fixed threshold $\bar{\theta} = -0.03$ used, e.g., by Farago and Tédongap (2018), which would generate 105 downside periods with a relative frequency approximately equal to 15.8% in our data. An analogous result is obtained in the case of the sample of multi-asset portfolios, where the estimated disappointment threshold $\hat{\theta} = -0.062$ and gives $\hat{T}_{\mathcal{D}} = 21$ downside observations, with frequency approximately equal to 15.5%. Therefore, for both sets of test assets, the estimated downside state is less frequent but more severe than implied by the fixed threshold typically assumed in the literature on downside risk.

To enhance our understanding of the downside regime, we evaluate the correlation between the downside indicators, as resulting from using the estimated threshold $\hat{\theta}$ and the value $\bar{\theta} = -0.03$ in Farago and Tédongap (2018), and various measures of economic and financial conditions that may be related to the downside regime. We consider the following set of variables: the NBER US recession indicator (*REC*); the log-difference of industrial production (ΔIP); the CBOE volatility index based on S&P 500 index options (VIX); the CBOE volatility index based on S&P 100 index options (VXO); the economic policy uncertainty index of Baker, Bloom, and Davis (2016) (EPU); the Equity Market Volatility tracker of Baker, Bloom, and Davis (2016) (EMV); the Policy-Related Equity Market Volatility tracker of Baker, Bloom, Davis, and Kost (2019) (*PREMV*); the Geopolitical Risk index of Caldara and Iacoviello (2018) (GPR); the TED spread (TED); the disaster probability for the US of Barro and Liao (2021) (SPX); the spread for the T-bill between 10 and 2 year maturities (T10Y2Y). Broadly speaking, the correlations point towards a connection between the downside indicator and the measures of economic and financial conditions we consider. This finding is consistent with a vast literature on asset pricing, which shows that volatility risk is a priced risk factor (see, *inter alia*, Ang, Hodrick, Xing, and Zhang (2006), Ang, Hodrick, Xing, and Zhang (2009), and Menkhoff, Sarno, Schmeling, and Schrimpf (2012a)). Note,

⁷Estimation of θ is carried out as explained in Section 4.1.

however, that the bilateral correlations are far from perfect and they often revolve around 0.5; this indicates that the determinants of the threshold go beyond volatility. Indeed, in the case of the equity panel, the correlation with the NBER recession indicator is weakly significant and positive. One interesting exception is the correlation with the term spread at different maturities, which is statistically insignificantly different from zero in both panels.⁸

We then run inference on the threshold parameter, and the results are shown in Panel B of Table 1. We first conduct a test, whose details are spelt out in Section A.2, for the null hypothesis $H_0: \theta = \overline{\theta}$: this allows us to formally assess whether the choice of the fixed threshold $\overline{\theta} = -0.03$ is supported by the data. As can be seen, the null is strongly rejected with a p-value of virtually zero and hence the estimated threshold $\hat{\theta}$ is significantly different from $\overline{\theta}$: this provides evidence in favor of our approach that allows to estimate the threshold for the downside regime. In addition to testing for the significance of the difference between $\hat{\theta}$ and $\overline{\theta}$, we use the same testing approach to test for the null that the estimated thresholds are the same across the two datasets of equity and multi-asset portfolios (last row of Table 1): these results show that the estimated threshold is the same in the two samples cannot be rejected at conventional levels of significance.

6.2.2 Estimating the number of common factors

Results are in Table 2.⁹ We estimate the number of common factors for three cases: the linear unconditional model described in Section 2.1 as applied to the whole available sample (this case is denoted henceforth without using any subscripts); the satisfactory regime j = S and the downside counterpart j = D, both estimated using the conditional model introduced in Section 2.2. In light of the results in Section 6.2.2, we only report results using the estimated threshold value $\hat{\theta}$. Alongside the estimated number of common factors, we evaluate the proportion of the

⁸An interesting question is what drives the value of θ , and the model could allow for θ to be endogenously determined by some state variables. We leave this further generalization for future research.

⁹We have used the algorithm described in Section 4.2. We have carried out the individual tests with M = N in Step 1; similarly to our findings in the simulations (reported in the Internet Appendix), we assessed the robustness of the results by using also $M = \lfloor N/2 \rfloor$ and M = 2N, but recorded virtually no changes. We have also set $\varepsilon = 0.01$ in (19). In the determination of the number of factors, we have carried out the individual tests using a nominal level $\alpha = \frac{0.05}{T}$.

total variance explained by each factor (and the cumulative one). Letting $g_j^{(i)}$ denote the *i*-th largest eigenvalue for each of the two regimes $j = \mathcal{D}, \mathcal{S}$ (and, similarly, using $g^{(i)}$ for the whole sample), the proportion of the total variance associated with the *i*-th common factor is measured as

$$\nu_j^{(i)} = \frac{g_j^{(i)}}{\sum_{i=1}^N g_j^{(i)}},\tag{28}$$

with $\nu^{(i)}$ defined similarly. The cumulative percentage of the total variance explained by \hat{P}_j estimated factors is given by

$$\nu_j = \frac{\sum_{i=1}^{P_j} g_j^{(i)}}{\sum_{i=1}^N g_j^{(i)}},\tag{29}$$

with ν defined analogously for the whole sample.

The most striking finding in Table 2 is the discrepancy between the number of estimated common factors in the two regimes. We note two main points. First, observing $\hat{P}_{\mathcal{D}}$, only one common factor is needed in the downside state for each of the two cross-sections of returns examined. Crucially, $\hat{P}_{\mathcal{S}}$ is always greater than one (being equal to six and four for the equity and multi-asset cross-sections respectively) and, interestingly, the proportion of the variance explained by the common factors in the downside regime is invariably at least the same as in the upside regime. The fact that the data tend to be driven by few(er) common factors during bad times suggests that dependence among assets increases in a downturn. In turn, this result is intuitively clear and consistent with the anecdotal evidence that factor diversification tends to disappear when needed most (see e.g. Page and Panariello (2018) and the references therein).¹⁰ Secondly, applying the test to the unconditional, linear model would lead to the overestimation of the number of factors, finding $\hat{P} = 6$ or 4 respectively in the two datasets.

These findings clearly indicate that it is important to allow for a different number of factors in each regime. From a statistical point of view, it is natural that removing a restriction leads to better inference - in this case, the number of common factors is neither understated (which would correspond to an omitted variable problem) nor overstated (which would be tantamount

¹⁰This result is also consistent with the literature that shows that, in crisis times, the macroeconomy is driven by a smaller number of common factors (we refer, *inter alia*, to Stock and Watson (2009), Cheng, Liao, and Schorfheide (2016) and Li, Todorov, Tauchen, and Lin (2017)).

to including irrelevant variables). From an empirical viewpoint, the results clearly imply that restricting the number of common factors to be the same across regimes is not an adequate choice.

6.3 Model fit

In this section, we compare the performance of our latent factor model with downside risk against a linear specification, in the same vein as Lettau, Maggiori, and Weber (2014). We report three measures of goodness of fit: the R^2 , the root of the mean squared prediction error (RMSPE), and the ratio between the RMSPE and the root mean squared return (RMSR). The results in Table 3 show that the conditional model performs well, with a very high R^2 both in the S regime and in the \mathcal{D} regime, as well as for the full sample; similar results are found using the other measures of goodness of fit based on the RMSPE.

Our findings also suggest that the performance of the linear, unconditional model over the full sample is quite satisfactory with an R^2 of 0.79 and 0.98 (when the number of common factors P is set to its estimate \hat{P} for the multi-asset and equity samples respectively, and relatively low RMSPE. However, we know from earlier evidence that the number of latent factors that matter for pricing the cross-section of asset returns in the downside regime is much smaller (namely, 1) than the number of factors required in the \mathcal{S} regime. Given that the data experience such strong regime dependence, it seems surprising that the unconditional, regime-independent pricing model performs as well as it does. To better understand the gain from using a conditional asset pricing model that allows for downside risk, it is therefore useful to check the performance of the unconditional model for observations in the \mathcal{S} and \mathcal{D} regimes separately. The results in Table 3 show that the performance of the unconditional model is very poor in *both* the \mathcal{D} and S regimes, with very low explanatory power and large RMSPEs. In fact, the R^2 is generally negative for observations in the \mathcal{D} regime. This suggests that the seemingly good performance of the unconditional model over the full sample is illusory, in the sense that it comes from somehow combining large pricing errors from two regimes which generate small pricing errors on aggregate, over the full sample.

The latter point can be seen clearly in Figure 2, where (for both sets of test assets) we report scatter plots of realized versus model-implied returns, for both the unconditional model, and the conditional model under three different cases: the \mathcal{D} regime, the \mathcal{S} regime, and the full sample. This figure makes it apparent how the unconditional model performs very poorly in the downside regime. Note the large positive errors: the unconditional model predicts much higher returns than the realized ones, which tend to be very negative in this regime. Hence, the unconditional model is not able to capture such extreme, bad events. The conditional model performs much better, and there appears to be no systematic tendency to generate positive or negative errors in the downside regime \mathcal{D} . Turning to the \mathcal{S} regime, the conditional model performs very well, with the majority of test asset returns lying close to, or on, the 45 degree line. In contrast, the returns implied by the unconditional model display a monotonic pattern (hence are highly positively correlated) with realized returns in the \mathcal{S} regime, but do not lie on the 45 degree line for any of the test asset returns. This means that, even in the \mathcal{S} regime, the unconditional model has a tendency to generate systematic (non-random) pricing errors; however, these pricing errors have the opposite sign than in the \mathcal{D} regime, since the unconditional model predicts lower returns than realized in the \mathcal{S} regime. Over the full sample, the averaging across the errors of these two regimes - large positive errors for a small number of observations in the \mathcal{D} regimes and more moderate negative pricing errors for a large number of observations in the \mathcal{S} regime - gives the appearance that the unconditional model performs well, but this is of course fallacious: the unconditional model performs poorly in *both* regimes.

The analysis above clarifies the gain from using a conditional latent factor model of asset pricing that allows for different regimes, and hence different sets of factors in good and bad times. The pricing model is very different in these two regimes: this captures the stark difference in the time-varying factor structure of asset returns, which is very low-dimensional in bad times, and requires more factors in good times. A model that does not allow for this feature, which is prominent across both datasets examined here, can spuriously generate a strong correlation between model-implied and realized returns over the full sample, but in fact it performs poorly in both good and bad times. At an applied level, the conditional model is consistent with the notion that asset returns become very highly correlated in bad times and their factor structure reduces in dimensionality to one factor, and this feature is key for the model to generate predicted returns that capture downside risk.



Figure 2: Predicted and expected returns

Notes. The scatter plot shows realized average returns against predicted average returns in downside, upside and over the entire sample for the 130 equity portfolios and the 57 multi-asset portfolios described in Sections 6.1.1 and 6.1.2, respectively.

6.4 Risk premia

Table 4 displays the results from estimation of the risk premia obtained from the three-pass procedure detailed in Section 5. We consider the following set of observable pricing factors: the return on the market in excess of the risk-free rate (RmRf); the four additional factors that complete the Fama and French (2015) five-factor model, that is size (SMB), value (HML), operating profitability (RMW), and investment (CMA); the momentum factor (MOM), taken from Carhart (1997), whose importance is further stressed in Asness, Moskowitz, and Pedersen (2013); the liquidity factor (LIQ) from Pástor and Stambaugh (2003); and ΔVXO , which is the change in the VXO index, capturing expected volatility in the S&P100 embedded in option prices.¹¹ For all factors we expect a positive risk premium, except for ΔVXO , as volatility is expected to have a negative price of risk. For both equity and multi-asset portfolios, we perform the analysis with respect to conditional and unconditional specifications; the latter case is essentially the three-pass procedure of Giglio and Xiu (2021), for which results are reported in the first column of Table 4 denoted "Full Sample". The second and third column of Table 4 report estimated risk premia from the conditional model with downside risk for each of the two regimes, while the final column reports the weighted average of the conditional, regime-specific estimates with weights equal to the number of observations in each regime.

Starting from equity portfolios, in the case of the unconditional model all risk premia are correctly signed and of reasonable magnitude. They are also generally statistically significantly different from zero, with the exception of RMW and CMA, which are statistically insignificant, and also acknowledging that the risk premium for SMB is only statistically significant at the 10 percent level. These results are fairly comparable to the ones reported by Giglio and Xiu (2021) on a different equity cross-section. The unconditional model, however, hides highly asymmetric dynamics between the regimes: in the occurrence of the downside event, RmRf, SMB and LIQgenerate excess returns that are negative and statistically significant (only at the 10 percent level for the case of LIQ), while ΔVXO gives positive excess returns; in good times, a higher number

¹¹The pricing factor RmRf is defined as RmRf = Rm - Rf, where Rm is the market return, and Rf is the risk-free rate as measured by the one-month Treasury bill rate. The series for Rm, Rf, SMB, HML, RMW, CMA and MOM are available from Kenneth French website: see footnote 4. LIQ is available from Luboš Pástor's website. The VXO data are from the CBOE website.

of factors, namely RmRf, SMB, MOM and LIQ are positively priced, while ΔVXO is negatively priced. The average implied risk premium, reported in the last column of the table, is precisely estimated (statistically significant at the one percent level) only for RmRf and ΔVXO , being for both these factors larger in magnitude than the risk premia estimated using the unconditional model specification.

Moving to multi-asset portfolios, we find statistically significant estimates of risk premia associated with four factors (the market excess return, MOM, LIQ and ΔVXO), although equity momentum enters with a negative sign in this cross section. However, the conditional models provide again a more informative picture: the risk premia of RmRf and SMB are negative and significant in downside periods; RmRf, MOM, LIQ and ΔVXO are statistically significant in good times, although MOM enters with a negative sign. The average risk premia in the last column confirm the most precisely estimated risk premia are the ones associated with RmRf and ΔVXO as in the equity cross-section, although MOM also displays statistical significance with a negative sign. This suggests that RmRf and ΔVXO are clearly associated with risk premia that are sizable, statistically significant and strongly regime-dependent but also that equity momentum is not related in a straighforward way to returns from portfolios outside the equity market universe.

7 Conclusions

In this paper, we propose a methodology to model and price the cross section of asset returns in the presence of common factors and downside risk. Both features have been shown to offer superior explanatory power in several empirical papers (e.g. Lettau, Maggiori, and Weber (2014)). We extend the framework used in the extant literature in at least three ways: we allow for downside risk via a threshold specification which allows for the estimation of the (usually set *a priori*) threshold level; we consider different factor structures in the two regimes, allowing also for different numbers of factors; we adapt the methodology developed by Giglio and Xiu (2021) to recover the observable factors risk premia from the estimated latent ones in both regimes.

The proposed model is illustrated through two empirical applications to a large cross-section of equity returns and a smaller, but more diverse, cross-section of returns from different asset classes. The results show that: estimating, rather than setting *a priori*, the threshold level improves goodness of fit, also determining substantial differences in the estimation of risk premia with respect to an exogenously fixed level; estimating the number of common factors separately between regimes yields very different estimates between regimes, with the downside risk state having a small number of common factors (one in our data), whilst returns in the good state of the world are driven by more common factors; estimating the risk premia of popular observable factors is more accurate and yields estimates consistent with strong asymmetries in the two regimes.

Finally, the empirical analysis shows clearly the limitations of unconditional asset pricing models that assume a constant relationship between risk and return, while illustrating that the proposed conditional model is capable to adequately characterize the key properties of different cross-sections of asset returns.

References

Ang, A., J. Chen, and Y. Xing (2006). Downside risk. Review of Financial Studies 19, 1191–1239.

Ang, A., R. J. Hodrick, Y. Xing, and X. Zhang (2006). The cross-section of volatility and expected returns. *Journal of Finance 61*, 259–299.

Ang, A., R. J. Hodrick, Y. Xing, and X. Zhang (2009). High idiosyncratic volatility and low returns: International and further us evidence. *Journal of Financial Economics* 91, 1–23.

Asness, C. S., T. J. Moskowitz, and L. H. Pedersen (2013). Value and momentum everywhere. *Journal of Finance 68*, 929–985.

Bai, J. (2003). Inferential theory for factor models of large dimensions. *Econometrica* 71, 135–171.

Bai, J. and S. Ng (2002). Determining the number of factors in approximate factor models. *Economet*rica 70, 191–221.

Bai, Z. and J. Yao (2012). On sample eigenvalues in a generalized spiked population model. *Journal of Multivariate Analysis 106*, 167–177.

Bai, Z. and J.-f. Yao (2008). Central limit theorems for eigenvalues in a spiked population model. In Annales de l'IHP Probabilités et statistiques, Volume 44, pp. 447–474.

Baker, S. R., N. Bloom, and S. J. Davis (2016). Measuring economic policy uncertainty. *Quarterly Journal of Economics 131*, 1593–1636.

Baker, S. R., N. Bloom, S. J. Davis, and K. J. Kost (2019). Policy news and stock market volatility. NBER Working Paper 25720.

Barigozzi, M. and L. Trapani (2021). Testing for common trends in nonstationary large datasets. *Journal* of Business & Economic Statistics. Forthcoming.

Barro, R. J. and G. Y. Liao (2021). Rare disaster probability and options pricing. *Journal of Financial Economics* 139, 750–769.

Bates, B. J., M. Plagborg-Møller, J. H. Stock, and M. W. Watson (2013). Consistent factor estimation in dynamic factor models with structural instability. *Journal of Econometrics* 177, 289–304.

Bollerslev, T. (2020). Realized semi(co)variation: Signs that all volatilities are not created equal. Working Paper, Duke University.

Bollerslev, T., S. Z. Li, and V. Todorov (2016). Roughing up beta: Continuous versus discontinuous betas and the cross section of expected stock returns. *Journal of Financial Economics* 120, 464–490.

Bollerslev, T., A. J. Patton, and R. Quaedvlieg (2021). Realized semibetas: Disentangling "good" and "bad" downside risks. *Journal of Financial Economics*. Forthcoming.

Caldara, D. and M. Iacoviello (2018). Measuring geopolitical risk. FRB International Finance Discussion Paper 1222.

Cappiello, L., R. F. Engle, and K. Sheppard (2006). Asymmetric dynamics in the correlations of global equity and bond returns. *Journal of Financial Econometrics* 4, 537–572.

Carhart, M. M. (1997). On persistence in mutual fund performance. Journal of Finance 52, 57-82.

Chan, K.-S. (1993). Consistency and limiting distribution of the least squares estimator of a threshold autoregressive model. *Annals of Statistics 21*, 520–533.

Cheng, X., Z. Liao, and F. Schorfheide (2016). Shrinkage estimation of high-dimensional factor models with structural instabilities. *Review of Economic Studies* 83, 1511–1543.

Cochrane, J. H. (2011). Presidential address: Discount rates. Journal of Finance 66, 1047–1108.

Delikouras, S. (2017). Where's the kink? Disappointment events in consumption growth and equilibrium asset prices. *Review of Financial Studies 30*, 2851–2889.

Delikouras, S. and A. Kostakis (2019). A single-factor consumption-based asset pricing model. *Journal* of Financial and Quantitative Analysis 54, 789–827.

Fama, E. F. and K. D. French (2015). A five-factor asset pricing model. *Journal of Financial Economics* 116, 1–22.

Fama, E. F. and J. D. MacBeth (1973). Risk, return, and equilibrium: Empirical tests. *Journal of Political Economy 81*, 607–636.

Farago, A. and R. Tédongap (2018). Downside risks and the cross-section of asset returns. *Journal of Financial Economics 129*, 69–86.

Giglio, S. and D. Xiu (2021). Asset pricing with omitted factors. *Journal of Political Economy 129*, 1947–1990.

Hansen, B. E. (2000). Sample splitting and threshold estimation. *Econometrica* 68, 575–603.

Harvey, C. R. and Y. Liu (2020). A census of the factor zoo. Technical report. Working Paper, Duke University and Purdue University.

He, Z., B. Kelly, and A. Manela (2017). Intermediary asset pricing: new evidence from many asset classes. Journal of Financial Economics 126, 1–35.

Horváth, L. and L. Trapani (2019). Testing for randomness in a random coefficient autoregression model. Journal of Econometrics 209, 338–352.

Lettau, M., M. Maggiori, and M. Weber (2014). Conditional risk premia in currency markets and other asset classes. *Journal of Financial Economics* 114, 197–225.

Lettau, M. and M. Pelger (2020). Estimating latent asset-pricing factors. *Journal of Econometrics 218*, 1–31.

Lewellen, J., S. Nagel, and J. Shanken (2010). A skeptical appraisal of asset pricing tests. *Journal of Financial Economics 96*, 175–194.

Li, J., V. Todorov, G. Tauchen, and H. Lin (2017). Rank tests at jump events. *Journal of Business & Economic Statistics*. Forthcoming.

Markovitz, H. M. (1959). Portfolio selection: Efficient diversification of investments. John Wiley.

Massacci, D. (2017). Least squares estimation of large dimensional threshold factor models. *Journal of Econometrics 197*, 101–129.

McLeish, D. L. (1974). Dependent central limit theorems and invariance principles. Annals of Probability 2, 620–628.

Menkhoff, L., L. Sarno, M. Schmeling, and A. Schrimpf (2012a). Carry trades and global foreign exchange volatility. *Journal of Finance 67*, 681–718.

Menkhoff, L., L. Sarno, M. Schmeling, and A. Schrimpf (2012b). Currency momentum strategies. *Journal* of Financial Economics 106, 660–684.

Merikoski, J. K. and R. Kumar (2004). Inequalities for spreads of matrix sums and products. *Applied Mathematics E-Notes* 4, 150–159.

Moricz, F. (1983). A general moment inequality for the maximum of the rectangular partial sums of multiple series. *Acta Mathematica Hungarica* 41, 337–346.

Nozawa, Y. (2017). What drives the cross-section of credit spreads?: A variance decomposition approach. Journal of Finance 72, 2045–2072.

Page, S. and R. A. Panariello (2018). When diversification fails. *Financial Analysts Journal* 74, 19–32.

Pástor, L. and R. F. Stambaugh (2003). Liquidity risk and expected stock returns. *Journal of Political Economy 111*, 642–685.

Peligrad, M. (1987). On the central limit theorem for ρ -mixing sequences of random variables. Annals of Probability 15, 1387–1394.

Peligrad, M., S. Utev, and W. Wu (2007). A maximal l_p -inequality for stationary sequences and its applications. *Proceedings of the American Mathematical Society* 135, 541–550.

Salzer, H. E., R. Zucker, and R. Capuano (1952). Table of the zeros and weight factors of the first twenty hermite polynomials. *Journal of Research of the National Bureau of Standards* 48, 111–116.

Schwert, G. W. (1989). Why does stock market volatility change over time? Journal of Finance 44, 1115–1153.
Shao, Q.-M. (1995). Maximal inequalities for partial sums of ρ -mixing sequences. Annals of Probability 23, 948–965.

Stock, J. H. and M. W. Watson (2009). Forecasting in dynamic factor models subject to structural instability. In J. Castle and N. Shephard (Eds.), *The Methodology and Practice of Econometrics: A Festschrift in Honour of David F. Hendry.* Oxford University Press.

Trapani, L. (2018). A randomized sequential procedure to determine the number of factors. *Journal of the American Statistical Association 113*, 1341–1349.

Uematsu, Y. and T. Yamagata (2019). Estimation of weak factor models. ISER DP Paper 1053.

Wang, W. and J. Fan (2017). Asymptotics of empirical eigenstructure for high dimensional spiked covariance. *Annals of Statistics* 45, 1342.

Panel A: Estimation							
	Eq	luity	Mult	i-asset			
	fixed	estimated	fixed	estimated			
heta	-0.030	-0.061	-0.030	-0.062			
$T_{\mathcal{D}}$	105	45	49	21			
$\pi_{\mathcal{D}}$	0.158	0.068	0.155	0.066			
			•				
$Corr\left(\mathcal{D},REC ight)$	0.166^{*}	0.244^{*}	0.197	0.338			
$Corr\left(\mathcal{D},\Delta IP ight)$	-0.054	-0.084	-0.032	-0.169			
$Corr\left(\mathcal{D}, VIX ight)$	0.508^{***}	0.538^{***}	0.475***	0.541^{**}			
$Corr\left(\mathcal{D}, VXO\right)$	0.503^{***}	0.545^{***}	0.486***	0.572^{***}			
$Corr\left(\mathcal{D}, EPU ight)$	0.166^{*}	0.293^{**}	0.200	0.354			
$Corr\left(\mathcal{D}, EMV ight)$	0.450^{***}	0.525^{***}	0.414***	0.494^{**}			
$Corr\left(\mathcal{D}, PREMV ight)$	0.426^{***}	0.516^{***}	0.399***	0.493^{**}			
$Corr\left(\mathcal{D},GPR ight)$	-0.003	0.045	0.030	0.096			
$Corr\left(\mathcal{D}, TED ight)$	0.163^{*}	0.210	0.185	0.247			
$Corr\left(\mathcal{D},SPX ight)$	0.431^{***}	0.493^{***}	0.388***	0.483^{**}			
$Corr\left(\mathcal{D},T10Y2Y\right)$	-0.059	-0.010	0.060	0.150			
Panel B: Inference							
	Eq	luity	Mult	zi-asset			
$H_0: \theta^0 = -0.03$	0.0	00***	0.000***				
$\underline{H_0:\theta_E^0=\theta_{MA}^0}$	1.	000					

Table 1: Estimation and inference on θ

Notes. The table contains all the relevant inference on θ in the datasets considered. In Panel A, for each dataset we report the estimated value, $\hat{\theta}$, and, as a term of comparison, the exogenously fixed value $\bar{\theta} = -0.030$, which is proposed in Farago and Tédongap (2018). For both values of θ , we also report the proportion of time periods in the downside regime \mathcal{D} , denoted as $\pi_{\mathcal{D}}$, and the corresponding number of time periods, denoted as $T_{\mathcal{D}}$. In Panel A we also report the correlation between the downside indicator functions implicitly defined by both $\hat{\theta}$ and $\bar{\theta}$, and the following variables: *REC* is the NBER U.S. recession indicator; ΔIP is the log difference of industrial production; *VIX* is the CBOE volatility index based on S&P 500 index options; *VXO* is the CBOE volatility index based on S&P 100 index options; *EPU* is the economic policy uncertainty index of Baker, Bloom, and Davis (2016); *EMV* is the Equity Market Volatility tracker of Baker, Bloom, and Davis (2016); *PREMV* is the Policy-Related Equity Market Volatility tracker of Baker, Bloom, Davis, and Kost (2019); *GPR* is the Geopolitical Risk index of Caldara and Iacoviello (2018); *TED* is the TED spread; *SPX* is the disaster probability for the U.S. of Barro and Liao (2021); *T10Y2Y* is the spread for the T-bill between 10 and 2 year maturities. In Panel B we report p-values for tests based on Section A.2. The first test is for the null hypothesis that θ^0 is equal to -0.03. The test at the bottom of the table is for the null that θ^0 is the same across the two datasets: we refer to the values of θ^0 in the equity and multi-asset sample as to θ^0_E and θ^0_{MA} respectively. ***, ** and * indicate significance at 1%, 5% and 10% level, respectively.

Panel A: Unconditional Model									
Equity					Multi-asset				
Co	Common Individual		Co	Common		vidual			
\widehat{P}	ν	$g^{(i)}$	$ u^{(i)}$	\widehat{P}	ν	$g^{(i)}$	$ u^{(i)}$		
6	0.945	1	0.833	4	0.881	1	0.712		
		2	0.050			2	0.085		
		3	0.028			3	0.053		
		4	0.019			4	0.031		
		5	0.009						
		6	0.007						

 Table 2: Number of estimated common factors

Panel B: Conditional model, Regime ${\mathcal S}$							
	Equ	uity			Multi	-asset	
Cor	nmon	Indi	vidual	Common		Individual	
$\widehat{P}_{\mathcal{S}}$	ν_{S}	$g_{\mathcal{S}}^{(i)}$	$ u_{\mathcal{S}}^{(i)}$	\widehat{P}_{S}	ν_{S}	$g_{\mathcal{S}}^{(i)}$	$\nu_{\mathcal{S}}^{(i)}$
6	0.928	1	0.781	4	0.843	1	0.621
		2	0.068			2	0.115
		3	0.037			3	0.071
		4	0.021			4	0.036
		5	0.012				
		6	0.009				
	Panel	C: Co	onditiona	al mo	del, Reg	gime \mathcal{I})
	Equ	uity			Multi	-asset	
Cor	nmon	Indi	vidual	Cor	nmon	Indi	vidual
$\widehat{P}_{\mathcal{D}}$	$\nu_{\mathcal{D}}$	$g_{\mathcal{D}}^{(i)}$	$ u_{\mathcal{D}}^{(i)}$	$\widehat{P}_{\mathcal{D}}$	$\nu_{\mathcal{D}}$	$g_{\mathcal{D}}^{(i)}$	$ u_{\mathcal{D}}^{(i)}$
1	0.924	1	0.924	1	0.873	1	0.873

Notes. The table contains the number of estimated common factors in each regime. For each dataset considered, we report inference on the number of common factors in the two regimes - the "normal" regime, denoted as S, and the "downside" regime, denoted as D. In Panel A, we estimate the number of common factors for the whole sample, which is tantamount to estimating the number of common factors in a linear specification without thresholds (the "unconditional model"). In Panels B and C, we report results for the conditional model for regime S and D respectively. Each sub-table in the panel (for different regimes and different ways of setting the threshold) contains two types of information. The aggregate information (in the columns headed as "Common") contains the estimated number of common factors, \hat{P} , and the percentage of the variance explained by all the common factors (this is denoted as ν_j , for j = D, S, using the notation in equation (28)); similarly, ν refers to the whole sample estimation. The individual common factor (this is denoted as "Individual") contains the percentage of the variance explained by the *i*-th individual common factor (this is denoted as $\nu_j^{(i)}$, for j = D, S, using the notation in equation (29)); similarly, $\nu^{(i)}$ refers to the whole sample estimation.

Panel A: Conditional model with downside risk										
BMSPE										
			Equity				1	Multi-asset		
	$D = \hat{D}$					$D = \hat{D}$	-			
	$\Gamma_j = \Gamma_j$					$\Gamma_j = \Gamma_j$				
Regime D	2.373					1.826				
Regime S	0.105					0.354				
Average	0.258					0.452				
					BMSPF	C/RMSR				
			Equity			/	1	Multi-asset		
	DÔ		Equity			D Ô		abbee		
	$P_j = P_j$					$P_j = P_j$				
Regime D	0.239					0.275				
Regime S	0.072					0.278				
Average	0.083					0.278				
					P	32				
			Equity		-	c .	1	Multi-asset		
	D Ô		Equity			D Ô		W10101-05500		
	$P_j = P_j$					$P_j = P_j$				
Regime D	0.942					0.923				
Regime S	0.994					0.915				
Average	0.990					0.915				
~										
				D1.T	о т • • •	1.4.1.4.1	1 1 . 1			
				Panel E	s: Linear, u	nconditiona	i model			
					RM	SPE				
Realized Returns			Equity]	Multi-asset		
	$P = \hat{P}$	P = 1	P = 5	P = 10	P = 15	$P = \hat{P}$	P = 1	P = 5	P = 10	P = 15
Begime \mathcal{D}	10.570	10.570	10 570	10 570	10 570	7 316	7 304	7 306	7 302	7 302
Pogimo S	0.774	0.810	0 774	0 771	0 770	0.622	0.702	0.571	0.559	0.551
E U C	0.114	0.810	0.114	0.771	0.110	0.033	0.703	0.571	0.001	0.331
Full Sample	0.105	0.255	0.106	0.085	0.078	0.370	0.460	0.240	0.201	0.181
					RMSPE	E/RMSR				
Realized Returns			Equity]	Multi-asset		
	$P = \hat{P}$	P = 1	P = 5	P = 10	P = 15	$P = \hat{P}$	P = 1	P = 5	P = 10	P = 15
Regime \mathcal{D}	1.067	1.067	1.067	1.067	1.067	1.101	1.099	1.100	1.099	1.099
Begime S	0.533	0.558	0.533	0.531	0.531	0.497	0.552	0.448	0.438	0.432
Full Commis	0.333	0.358	0.555	0.331	0.331	0.497	0.552	0.448	0.438	0.432
Full Sample	0.149	0.359	0.149	0.119	0.110	0.459	0.570	0.297	0.249	0.225
					I	₹ 4				
Realized Returns			Equity]	Multi-asset		
	$P = \hat{P}$	P = 1	P = 5	P = 10	P = 15	$P = \hat{P}$	P = 1	P = 5	P = 10	P = 15
Begime \mathcal{D}	-0.043	-0.003	0.005	0.005	0.005	-0.063	-0.008	-0.081	-0.196	-0.338
Pogimo S	0.107	-0.003	0.000	0.005	0.000	0.003	0.008	0.206	0.244	0.161
Regime o	0.197	0.202	0.234	0.230	0.230	0.200	0.208	0.300	0.244	0.101
Full Sample	0.978	0.871	0.978	0.986	0.988	0.790	0.675	0.912	0.938	0.950

 Table 3: Goodness of fit measures

Notes. This table shows the following goodness of fit measures for the conditional model with downside risk (Panel A), and for the linear, unconditional model (Panel B): the root-mean-squared pricing error (RMSPE); the RMSPE to root-mean-squared returns ratio (RMSPE/RMSR); the R-squared (R^2) . In Panel A, the RMSPE in regime $j = \mathcal{D}, \mathcal{S}$ is computed as $\sqrt{\alpha'_j \alpha_j} / N$, with $\alpha_j = \bar{\mathbf{R}}_j - \hat{\mathbf{X}}_j \hat{\mathbf{\Gamma}}_j$, where $\bar{\mathbf{R}}_j = \hat{T}_j^{-1} \sum_{t=1}^T \hat{d}_{j,t} \mathbf{R}_t$ is the realized return; the RMSPE/RMSR is computed as $\sqrt{\alpha'_j \alpha_j} / \sqrt{\bar{\mathbf{R}}'_j \bar{\mathbf{R}}_j}$; R^2 is computed as $1 - (\alpha'_j \alpha_j / \bar{\mathbf{R}}'_j \bar{\mathbf{R}}_j) \left[(N-1) / (N-\hat{P}_j) \right]$; Average is computed as the weighted average of the quantity of interest in the two regimes, with weights equal to the sample frequencies of the regimes. In Panel B, realized returns over the full sample are computed as $\bar{\mathbf{R}} = \hat{T}^{-1} \sum_{t=1}^{T} \mathbf{R}_t$; the RMSPE in regime $j = \mathcal{D}, \mathcal{S}$ is equal to $\sqrt{\alpha'_{j\mathcal{U}}\alpha_{j\mathcal{U}}} / N$, with $\alpha_{j\mathcal{U}} = \bar{\mathbf{R}}_j - \hat{\mathbf{X}}\hat{\mathbf{\Gamma}}$, and the RMSPE over the full sample is $\sqrt{\alpha' \alpha} / N$, with $\alpha = \bar{\mathbf{R}} - \hat{\mathbf{X}}\hat{\mathbf{\Gamma}}$; the RMSPE/RMSR in regime $j = \mathcal{D}, \mathcal{S}$ is computed as $\sqrt{\alpha'_{j\mathcal{U}}\alpha_{j\mathcal{U}}} / \sqrt{\bar{\mathbf{R}'\bar{\mathbf{R}}}}$ and as $\sqrt{\alpha' \alpha} / \sqrt{\bar{\mathbf{R}'\bar{\mathbf{R}}}}$ over the full sample; the R^2 in regime $j = \mathcal{D}, \mathcal{S}$ is computed as $1 - (\alpha'_{j\mathcal{U}}\alpha_{j\mathcal{U}}/\bar{\mathbf{R}'\bar{\mathbf{R}}}) \left[(N-1) / (N-\hat{P}) \right]$ and as $j = \mathcal{D}, \mathcal{S}$ is computed as $1 - (\alpha' \alpha / \bar{\mathbf{R}'\bar{\mathbf{R}}}) \left[(N-1) / (N-\hat{P}) \right]$ over the full sample.

	Full Sample		Regime D		Regime S		Avg	
	Equity	Multi-asset	Equity	Multi-asset	Equity	Multi-asset	Equity	Multi-asset
RmRf	0.520^{**}	0.924^{***}	-6.104^{***}	-5.78^{***}	1.199***	1.612^{***}	0.705***	1.120^{***}
	(0.257)	(0.296)	(1.202)	(1.785)	(0.230)	(0.232)	(0.230)	(0.247)
SMB	0.250^{*}	-0.043	-5.619^{***}	-4.788^{***}	0.409***	0.153	0.001	-0.175
	(0.145)	(0.216)	(0.973)	(0.699)	(0.152)	(0.221)	(0.156)	(0.212)
HML	0.298^{**}	0.363	-1.704	-2.862	0.208^{*}	0.205	0.079	0.001
	(0.135)	(0.270)	(1.462)	(2.172)	(0.127)	(0.261)	(0.154)	(0.283)
RMW	0.126	0.049	-0.352	-0.710	0.035	-0.078	0.009	-0.120
	(0.099)	(0.119)	(0.975)	(1.020)	(0.087)	(0.144)	(0.105)	(0.151)
CMA	0.143	0.040	-0.822	-1.197	0.052	-0.068	-0.008	-0.143
	(0.094)	(0.163)	(0.914)	(1.167)	(0.087)	(0.174)	(0.102)	(0.180)
MOM	0.662^{***}	-0.459^{**}	-1.798	-1.682	0.611***	-0.845^{**}	0.448^{*}	-0.901^{**}
	(0.161)	(0.186)	(2.015)	(2.084)	(0.199)	(0.422)	(0.230)	(0.418)
LIQ	0.300^{**}	0.460^{***}	-7.311^{*}	-7.550	0.300***	0.509^{**}	-0.214	-0.026
	(0.134)	(0.172)	(4.339)	(5.281)	(0.108)	(0.213)	(0.310)	(0.403)
$\Delta V X O$	-0.389^{**}	-0.386^{**}	1.945^{**}	2.002	-0.856^{***}	-0.731^{***}	-0.644^{***}	-0.531^{**}
	(0.179)	(0.165)	(0.897)	(1.355)	(0.191)	(0.227)	(0.189)	(0.233)

 Table 4: Estimation results

Notes. The table contains estimation results for the equity and multi-asset portfolios described in Section 6.1. The sample of equity portfolios runs from July 1963 to December 2018. The sample of multi-asset portfolios runs between October 1983 and January 2010. The sample for ΔVXO is available from February 1986 onward. ***, ** and * indicate significance at 1%, 5% and 10% level, respectively.

Internet Appendix (not for publication)

Factor models with downside risk

A Complements to Sections 4 and 5

In this section of the Internet Appendix, we complement the theory spelt out in Sections 4 and 5, adding further details and explanations. In particular, in Section A.1, we list and discuss all the relevant assumptions used in the paper. In Section A.2, we discuss in greater depth the test presented in Section 4.1. In Section A.3, we present all the relevant theory for the estimation of the number of common factors in each regime, discussed in Section 4.2. Finally, in Section A.4, we complete the asymptotic theory developed in Section 5.3.

Here and in the rest of the appendix, we use the notation c_0 , c_1 ,... to denote positive and finite constants, which do not depend on the sample sizes and whose value can change from line to line, and we use the expression "a.s." as short-hand for "almost surely".

A.1 Assumptions and further notation

Let

$$\underline{\mathbf{u}}_{t}^{0} = \mathbf{u}_{\mathcal{D},t}^{0} \cup \mathbf{u}_{\mathcal{S},t}^{0}, \tag{A.1}$$

where $\underline{\mathbf{u}}_{t}^{0}$ has dimension P_{u}^{0} such that max $(P_{\mathcal{D}}^{0}, P_{\mathcal{S}}^{0}) \leq P_{u}^{0} \leq P_{\mathcal{D}}^{0} + P_{\mathcal{S}}^{0}$, and let $\mathbf{B}_{u,j}^{0} = \left(\beta_{u,j,1}^{0}, ..., \beta_{u,j,N}^{0}\right)'$ be, for $j = \mathcal{D}, \mathcal{S}$, the $N \times P_{u}^{0}$ matrix suitably filled with zeros so that $\mathbf{B}_{u,j}^{0} \underline{\mathbf{u}}_{t}^{0} = \mathbf{B}_{j}^{0} \mathbf{u}_{j,t}^{0}$ for $j = \mathcal{D}, \mathcal{S}$. Finally, let

$$\delta_i^0 = \beta_{u,\mathcal{S},i}^0 - \beta_{u,\mathcal{D},i}^0. \tag{A.2}$$

Assumption 1. There exists an $\eta \in \left(\frac{1}{2}, 1\right]$ such that $\delta_i^0 \neq 0$ for $i = 1, ..., \lfloor N^\eta \rfloor$ and $\sum_{i=\lfloor N^\eta \rfloor+1}^N \|\delta_i^0\| < \infty$.

Assumption 1 is related to the results in Bates, Plagborg-Møller, Stock, and Watson (2013), who show that, if no more than a fraction $O(N^{1/2})$ of the cross-sectional units in a large dimensional factor model have a structural break in the loadings, then the principal components estimator applied to a (misspecified) linear model has the same convergence rate as in the no break case - namely, $O_p(C_{N,T}^{-1})$ (see Bai and Ng (2002)). Thus, Assumption 1 requires that at least a fraction $O(N^{\eta})$ of the N assets have regime specific factor loadings, for $\frac{1}{2} < \eta \leq 1$. Assumption 2. It holds that, for all N: (i) (a) the largest eigenvalue of $T_j^{-1} \sum_{t=1}^T E(\epsilon_t \epsilon'_t d_{j,t}(\theta^0))$ is finite for $j = \mathcal{D}, \mathcal{S}$, and (b) the smallest eigenvalue of $T_j^{-1} \sum_{t=1}^T E(\epsilon_t \epsilon'_t d_{j,t}(\theta^0))$ is positive for $j = \mathcal{D}, \mathcal{S}$; (ii)

$$\max_{1 \le i \le N} \sum_{k=1}^{N} \frac{1}{T_j} \sum_{t=1}^{T} \sum_{s=1}^{T} \left| E\left(\epsilon_{i,t} \epsilon_{k,s} d_{j,t}\left(\theta^{0}\right) d_{j,s}\left(\theta^{0}\right) \right) \right| < \infty,$$

for $j = \mathcal{D}, \mathcal{S}$.

Assumption 3. It holds that (i) $E \|\mathbf{f}_{j,t}\|^{4+2r} < \infty$ for some r > 0, with $j = \mathcal{D}, \mathcal{S}$ and $1 \le t \le T$; (ii) it holds that, as $T \to \infty$

$$\frac{1}{T}\sum_{t=1}^{T} E\left(\mathbf{f}_{j,t}^{0} d_{j,t}\left(\theta\right) \mathbf{f}_{j,t}^{\prime} d_{j,t}\left(\theta^{0}\right)\right) \to \Sigma_{\mathbf{f},j}\left(\theta,\theta^{0}\right).$$

where $\Sigma_{\mathbf{f},j}(\theta, \theta^0)$ is positive definite for $j = \mathcal{D}, \mathcal{S}$ and all θ ; (iii) $\left\|\beta_{j,i}^0\right\| < \infty$ for $j = \mathcal{D}, \mathcal{S}$ and $1 \leq i \leq N$; (iv) $N^{-1}\mathbf{B}'_{j}\mathbf{B}_{j} \to \Sigma_{\mathbf{B},j}$ as $N \to \infty$, where $\Sigma_{\mathbf{B},j}$ are positive definite matrices for $j = \mathcal{D}, \mathcal{S}$; (v) the eigenvalues of $\Sigma_{\mathbf{B},j}\Sigma_{\mathbf{f},j}(\theta^0, \theta^0)$ are distinct for $j = \mathcal{D}, \mathcal{S}$.

Assumption 2 stipulates that the idiosyncratic part of the model (i.e., the errors $\epsilon_{i,t}$) can be cross-sectionally correlated; however, by part (i) of the assumption such cross-sectional correlation is weak. Conversely, by Assumption 3, the "signal" part of the model (i.e., the one related to the common factor structure) introduces strong cross-sectional correlation between the $\widetilde{\mathbf{R}}_{i,t}$ s. In particular, as we show in Lemma A.1 below, Assumption 3 entails that the eigenvalues of the covariance matrix of the $\widetilde{\mathbf{R}}_{i,t}$ s have a spiked structure, whereby the largest eigenvalues diverge proportionally to N whereas the other ones are bounded.

Assumption 4. It holds that (i) $E(\epsilon_{i,t}) = 0$ and $E|\epsilon_{i,t}|^8 < \infty$; (ii) $E(d_{j,t}(\theta) d_{j,v}(\theta) \epsilon_{i,t}\epsilon_{i,v}) = \tau_{j,i,t,v}(\theta)$ with $|\tau_{j,i,t,v}(\theta)| < |\tau_{j,t,v}|$ for $1 \le i \le N$, and $T_j^{-1} \sum_{t=1}^T \sum_{v=1}^T |\tau_{j,t,v}| < \infty$ for $j = \mathcal{D}, \mathcal{S}$; (iii) $E\left(T_j^{-1} \sum_{t=1}^T d_{j,t}(\theta) \epsilon_{i,t}\epsilon_{l,t}\right) = \sigma_{j,i,l,t}(\theta)$ with $|\sigma_{j,i,l,t}(\theta)| \le |\sigma_{j,i,l}|$ for $1 \le t \le T$ and $N^{-1} \sum_{i=1}^N \sum_{l=1}^N |\sigma_{j,i,l}| < \infty$ for $j = \mathcal{D}, \mathcal{S}$; (iv)

$$E\left|T_{j}^{-1/2}\sum_{t=1}^{T}\left(d_{j,t}\left(\theta\right)\epsilon_{i,t}\epsilon_{l,t}-E\left(d_{j,t}\left(\theta\right)\epsilon_{i,t}\epsilon_{l,t}\right)\right)\right|^{4}<\infty,$$

for $j = \mathcal{D}, \mathcal{S}$ and $1 \le i, l \le N$; (v) it holds that $E\left(\mathbf{f}_{j,s}^{0}\epsilon_{i,t}|z_{t}\right) = 0$ for $j = \mathcal{D}, \mathcal{S}$ and $1 \le t, s \le T$, with $E\left[N^{-1}\sum_{i=1}^{N} \left\|T_{j}^{-1/2}\sum_{t=1}^{T}d_{j,t}\left(\theta\right)\mathbf{f}_{j,t}^{0}\epsilon_{i,t}\right\|^{2}\right] < \infty,$

for all θ and $j = \mathcal{D}, \mathcal{S}$.

Assumption 4 mimics similar assumptions in the literature (see e.g. Assumptions C and D in Bai (2003)), and essentially it states that the idiosyncratic errors are weakly dependent. Parts (iv) and (v) of the assumption are, essentially, weak (serial) dependence conditions.

Assumption 5. It holds that (i) $\{\mathbf{f}_{j,t}^{0}, z_{t}, \epsilon_{t}\}$ is a strictly stationary, ergodic, ρ -mixing sequence with mixing numbers $\rho_{j,m}$ satisfying $\sum_{m=1}^{\infty} \rho_{j,m}^{1/2} < \infty$ for $j = \mathcal{D}, \mathcal{S}$ (ii) $E\left(\left\|\mathbf{f}_{j,t}^{0}\epsilon_{i,t}\right\|^{4}|z_{t}\right) < \infty$ a.s. and $E\left(\left\|\mathbf{f}_{j,t}^{0}\right\|^{4}|z_{t}\right) < \infty$ a.s. for $j = \mathcal{D}, \mathcal{S}$ and $1 \leq i \leq N$; (iii) z_{t} has bounded and strictly positive density; (iv) the density of z_{t} is continuous at θ^{0} ; (v) $E\left(\mathbf{f}_{j,t}^{0}\mathbf{f}_{j,t}^{0'}|z_{t}=\theta\right)$ is a.s. positive definite and continuous at $\theta = \theta^{0}$ for $j = \mathcal{D}, \mathcal{S}$ with $\sum_{i=\lfloor N^{\eta} \rfloor + 1}^{N} \delta_{i}^{0'} E\left(\mathbf{f}_{j,t}^{0}\mathbf{f}_{j,t}^{0'}|z_{t}=\theta^{0}\right) \delta_{i}^{0} < \infty$ a.s.; (vi) $\{\mathbf{f}_{j,t}^{0}\}$ and $\{\epsilon_{i,t}\}$ are two mutually independent groups conditionally on z_{t} for $j = \mathcal{D}, \mathcal{S}$ and $1 \leq i \leq N$.

Assumption 5 draws upon Assumption 1 in Hansen (2000) (see also Massacci (2017)), to which we refer to for further discussion. In our context, it is used in order to obtain the convergence rate of the estimators for factor innovations and loadings. We point out that some of the other assumptions above - chiefly the ones that contain "time series results" (e.g., Assumptions 4(iv)-(v)) - would follow from Assumption 5, although they could also be derived from other dependence assumptions.

A final note on the assumptions above. Assumption 3(iv) is typical in the literature on inference on factor models (we refer, inter alia, to the contributions by Bai and Ng (2002) and Bai (2003)) and, in essence, it requires that the common factors be "strong" or "pervasive". A well-known consequence of the assumption is that - as we show in Lemma A.1 - the eigenvalues of the second moment matrix of the data diverge at a rate exactly equal to N, which we exploit in the construction of our tests for the determination of the number of common factors. Conversely, Assumption 3(iv) rules out the potentially interesting case of having "weak" common factors. As pointed out in Uematsu and Yamagata (2019), a "weak" common factor can be defined as a common factor whose corresponding eigenvalue in the second moment of the matrix diverges, but at a rate slower than N. Such a case could be of interest in general, and especially in the context of a threshold factor model, where e.g. a pervasive factor affects only a fraction of the units in one regime, and is absent in the other regime.

Assumption 6. As $\min(N, T) \to \infty$, it holds that

$$\frac{(\ln N)^{1+\epsilon} (\ln T)^{1/2+\epsilon}}{T^{1/2} \pi_j} = 0,$$

for $j = \mathcal{D}, \mathcal{S}$.

By Assumption 6, $\hat{g}_j^{(i)} - g_j^{(i)}$ is of a smaller order of magnitude than N. As mentioned above, this ensures the spiked structure of the spectrum of the sample second moment matrices.ⁱ

Assumption 7. ⁱⁱ It holds that, for $j = \mathcal{D}, \mathcal{S}$: (i) $E\left(\sum_{i=1}^{N} w_i \alpha_{j,i}\right)^2 \leq c_0 N$ for all $\{w_i\}_{i=1}^{N}$ such that $\sum_{i=1}^{N} w_i^2 < \infty$; (ii) $\{\alpha_{j,i}\}_{i=1}^{N}$ is independent of all other quantities; (iii) it holds that, as $N \to \infty$, $\left[Var\left(\sum_{i=1}^{N} v_i \alpha_{j,i}\right)\right]^{-1/2} \sum_{i=1}^{N} v_i \alpha_i \xrightarrow{D} N(0, I_{\nu})$ for every ν -dimensional sequence v_i , $1 \leq i \leq N$, such that $\|N^{-1}\sum_{i=1}^{N} v_i v_i'\| < \infty$.

Assumption 8. It holds that: (i) in (4), (a) $\|\Lambda\| = O(1)$ and (b) $E(\mathbf{e}_t) = 0$ with $E \|\mathbf{e}_t\|^4 < \infty$; (ii) $T^{-1} \sum_{t=1}^T \mathbf{e}_t \mathbf{u}_{j,t}^{0'} = O_P(1)$ for $j = \mathcal{D}, \mathcal{S}$.

Assumption 9. For $j = \mathcal{D}, \mathcal{S}$, it holds that (i) $E\left(\mathbf{e}_t | \mathbf{u}_{j,t}^0\right) = 0$ a.s. for all t and (ii) as $T \to \infty$

$$T^{-1/2} \left(\begin{array}{c} \sum_{t=1}^{T} vec\left(\mathbf{e}_{t} \mathbf{u}_{j,t}^{0\prime}\right) d_{j,t}\left(\theta^{0}\right) \\ \sum_{t=1}^{T} \mathbf{u}_{j,t}^{0} d_{j,t}\left(\theta^{0}\right) \end{array} \right) \xrightarrow{D} N \left[\left(\begin{array}{c} 0 \\ 0 \end{array} \right), \left(\begin{array}{c} V_{\mathbf{eu},j} & \Pi_{\mathbf{ue},j} \\ \Pi_{\mathbf{ue},j} & V_{\mathbf{u},j} \end{array} \right) \right],$$

ⁱIn essence, the assumption is a restriction on the rates at which π_j can drift to zero: in particular, the assumption is satisfied as long as $\pi_j^{-1} = O(T^{1/2-\epsilon})$ for any $\epsilon > 0$. Note that the assumption also (very mildly) restricts the relative rate of divergence between N and T as they pass to infinity - in essence, however, it allows for N to grow at an arbitrarily high polynomial order with T, i.e. it allows for $N = O(T^{\kappa})$ for any value of κ .

ⁱⁱAssumption 7 contains some regularity conditions on the pricing errors α_i in (1). By part (*i*), we require that the second moment of the α_i s grows linearly with N; a possible case in which this would happen is when α_i is *i.i.d.* with zero mean and finite variance. However, the assumption is more general than this, and it allows e.g. for the α_i s to have nonzero mean. Indeed, if $E(\alpha_i) \neq 0$ for $1 \leq i \leq N_{\alpha}$ with $N_{\alpha} = O(N^{1/2})$, and $E(\sum_{i=N_{\alpha}+1}^{N} w_i \alpha_i)^2 \leq c_0 N$, the assumption would hold anyway. Note also that part (*iii*) of the assumption could be derived using more primitive assumptions on the α_i s (see e.g. the very general CLT in McLeish (1974)).

where

$$V_{\mathbf{eu},j} = \lim_{T \to \infty} E\left(T^{-1} \sum_{t,s=1}^{T} vec\left(\mathbf{e}_{t} \mathbf{u}_{j,t}^{0\prime}\right) \left(vec\left(\mathbf{e}_{s} \mathbf{u}_{j,s}^{0\prime}\right)\right)' d_{j,t}\left(\theta^{0}\right) d_{j,s}\left(\theta^{0}\right)\right),$$

$$V_{\mathbf{u},j} = \lim_{T \to \infty} E\left(T^{-1} \sum_{t,s=1}^{T} \mathbf{u}_{j,t}^{0} \mathbf{u}_{j,s}'^{0} d_{j,t}\left(\theta^{0}\right) d_{j,s}\left(\theta^{0}\right)\right),$$

$$\Pi_{\mathbf{ue},j} = \lim_{T \to \infty} E\left(T^{-1} \sum_{t,s=1}^{T} \mathbf{u}_{j,t}^{0} \left(vec\left(\mathbf{e}_{s} \mathbf{u}_{j,s}^{0\prime}\right)\right)' d_{j,t}\left(\theta^{0}\right) d_{j,s}\left(\theta^{0}\right)\right).$$

Assumption 10. It holds that $\alpha_{j,i}$ is i.i.d. across i with $E\left(\alpha_{j,1}^2\right) = \sigma_{\alpha,j}^2 < \infty$ for $j = \mathcal{D}, \mathcal{S}$.

Assumption 10 is a simplifying assumption, which complements Assumption 7 by requiring the sphericity of the $\alpha_{j,i}$.

Assumption 11. It holds that: (i) $\alpha_{\mathcal{D}}$ and $\alpha_{\mathcal{D}}$ are independent; (ii) as $T \to \infty$

$$T^{-1/2} \sum_{t=1}^{T} \begin{pmatrix} \operatorname{vec} \left(e_t u_{\mathcal{D},t} d_{\mathcal{D},t} \right) \\ u_{\mathcal{D},t} d_{\mathcal{D},t} \\ \operatorname{vec} \left(e_t u_{\mathcal{S},t} d_{\mathcal{S},t} \right) \\ u_{\mathcal{D},t} d_{\mathcal{S},t} \end{pmatrix} \xrightarrow{d} N\left(0, V \right)$$

with

$$V = \begin{cases} V_{ue,\mathcal{D}} & \\ \Pi_{ue,\mathcal{D}} & V_{u,\mathcal{D}} & \\ \Pi_{ue,\mathcal{DS}} & \Pi_{ue,S;u,\mathcal{D}} & V_{ue,\mathcal{S}} & \\ \Pi_{ue,\mathcal{D};u,\mathcal{S}} & \Pi_{u,\mathcal{D};u,\mathcal{S}} & \Pi_{ue,\mathcal{S}} & V_{u,\mathcal{S}} \end{cases},$$

where $V_{ue,j}$, $V_{u,j}$, and $\Pi_{ue,j}$ are defined in Assumption 9 (for j = D, S) and

$$\Pi_{ue,\mathcal{S};u,\mathcal{D}}^{\prime} = \frac{1}{T} E\left(\left(\sum_{t=1}^{T} u_{\mathcal{D},t} d_{\mathcal{D},t}\right) \left(\sum_{t=1}^{T} vec\left(e_{t} u_{\mathcal{S},t} d_{\mathcal{S},t}\right)\right)^{\prime}\right),$$

$$\Pi_{ue,\mathcal{D};u,\mathcal{S}}^{\prime} = \frac{1}{T} E\left(\left(\sum_{t=1}^{T} u_{\mathcal{S},t} d_{\mathcal{S},t}\right) \left(\sum_{t=1}^{T} vec\left(e_{t} u_{\mathcal{D},t} d_{\mathcal{D},t}\right)\right)^{\prime}\right),$$

$$\Pi_{ue,\mathcal{D}\mathcal{S}}^{\prime} = \frac{1}{T} E\left(\left(\sum_{t=1}^{T} vec\left(e_{t} u_{\mathcal{D},t} d_{\mathcal{D},t}\right)\right) \left(\sum_{t=1}^{T} vec\left(e_{t} u_{\mathcal{S},t} d_{\mathcal{S},t}\right)\right)^{\prime}\right),$$

$$\Pi_{u,\mathcal{D};u,\mathcal{S}}^{\prime} = \frac{1}{T} E\left(\left(\sum_{t=1}^{T} u_{\mathcal{D},t} d_{\mathcal{D},t}\right) \left(\sum_{t=1}^{T} u_{\mathcal{S},t} d_{\mathcal{S},t}\right)^{\prime}\right).$$

A.2 Complements to Section 4.1: the test for $\theta^0 = c$

We discuss our test for H_0 : $\theta^0 = c$. Some arguments are very similar to the ones used in Section A.3, and we therefore present them only once, to avoid repetitions, in the next section.

We know that, by Lemma C.7

$$\widehat{\theta} - \theta^0 = o_{a.s.} \left(T^{-1} \overline{v}_{N,T} \left(\epsilon \right) \right),$$

where $\overline{v}_{N,T}(\epsilon)$ is defined in (C.4). Define now the deterministic sequence $s_{N,T}$ such that

$$\lim_{\min(N,T)\to\infty} s_{N,T} = \infty,$$
$$\lim_{\min(N,T)\to\infty} s_{N,T} T^{-1} \overline{v}_{N,T} (\epsilon) = 0,$$

for all $\epsilon > 0$. By construction, we have $s_{N,T}\left(\hat{\theta} - c\right) = s_{N,T}\left(\theta^0 - c\right) + o_{a.s.}(1)$. This, and Lemma C.7, entail that, under H_0

$$P\left(\omega:\lim_{\min(N,T)\to\infty}s_{N,T}\left(\widehat{\theta}-c\right)=0\right)=1,$$

and therefore we can assume that, under the null

$$\lim_{\min(N,T)\to\infty}s_{N,T}\left(\widehat{\theta}-c\right)=0.$$

Conversely, whenever $\theta^0 \neq c$, it follows that

$$P\left(\omega:\lim_{\min(N,T)\to\infty}s_{N,T}\left(\widehat{\theta}-c\right)=\infty\right)=1,$$

and therefore we can assume that, under H_A

$$\lim_{\min(N,T)\to\infty}s_{N,T}\left(\widehat{\theta}-c\right)=\infty$$

This dichotomous behaviour can be reversed by choosing a transformation $h(\cdot)$ such that $\lim_{x\to\infty} h(x) = 0$ and $\lim_{x\to 0} h(x) = \infty$, thus defining

$$f_{NT} = h\left(s_{N,T}\left(\widehat{\theta} - c\right)\right).$$

Exactly as above, we can assume, on account of continuity, that

$$\lim_{\min(N,T)\to\infty} h\left(s_{N,T}\left(\widehat{\theta}-c\right)\right) = \infty,$$
$$\lim_{\min(N,T)\to\infty} h\left(s_{N,T}\left(\widehat{\theta}-c\right)\right) = 0,$$

under H_0 and H_A respectively.

We can now construct the following test (also similar to the one in Section 4.2 below):

Step 1 Generate an *i.i.d.* sequence $\{\xi_m^{\theta}, 1 \le m \le M_{\theta}\}$, with common distribution N(0, 1); **Step 2** Generate the Bernoulli sequence $\zeta_m^{\theta}(s) = I\left(f_{N,T}^{1/2}\xi_m^{\theta} \le s\right)$;

Step 3 Define

$$S^{\theta} = \int_{-\infty}^{+\infty} \left| \sigma^{\theta} \left(s \right) \right|^2 \frac{1}{\sqrt{2\pi}} \exp\left(-\frac{1}{2} s^2 \right) ds,$$

where

$$\sigma^{\theta}(s) = \frac{2}{M_{\theta}^{1/2}} \sum_{m=1}^{M_{\theta}} \left(\zeta_m^{\theta}(s) - \frac{1}{2} \right).$$

Theorem A.1. We assume that the assumptions of Lemma C.7 are satisfied. Then, as $\min(N, M_{\theta}, T) \rightarrow \infty$ with

$$\frac{M_{\theta}}{h^{1/2}\left(s_{N,T}T^{-1}\overline{v}_{N,T}\left(\epsilon\right)\right)} \to 0,\tag{A.3}$$

it holds that $S^{\theta} \xrightarrow{D^*} \chi_1^2$, under H_0 , and that $R^{-1}S^{\theta} \xrightarrow{P^*} c_0 > 0$ under H_A , for almost all realizations of $\{R_{i,t}, 1 \le i \le N, 1 \le t \le T\}$.

The proof of Theorem A.1 is very similar to the proofs in Horváth and Trapani (2019), and it is therefore omitted to save space. Letting χ_1^2 denote a chi-squared distribution with one degree of freedom, and defining the critical value c_{α} as $P(\chi_1^2 \ge c_{\alpha}) = \alpha$ for a nominal level α , Theorem A.1 entails

$$\lim_{\min(N,T,M_{\theta})\to\infty} P^* \left(S^{\theta} \ge c_{\alpha} \right) = 1, \tag{A.4}$$

under H_A , and

$$\lim_{\min(N,T,M_{\theta})\to\infty} P^* \left(S^{\theta} \ge c_{\alpha} \right) = \alpha, \tag{A.5}$$

under H_0 . Equation (A.4) has the "traditional" interpretation: when the null is false, the test asymptotically - rejects with probability 1. Conversely, equation (A.5) does not have a straightforward meaning. The test is constructed using a randomization based on $\{\xi_m^\theta\}$ which does not vanish asymptotically, and therefore the asymptotics of S^θ is driven by the added randomness. Thus, different researchers using the same data will obtain different values of S^θ , and, consequently, different *p*-values. In order to ameliorate this, we - along the lines of Horváth and Trapani (2019) - propose to generate the test statistic S_b^θ for $1 \le b \le B$ times, using, at each iteration, an *i.i.d.* sequence $\{\xi_{m,b}^\theta\}$, $1 \le m \le M_\theta$, with common distribution N(0, 1), independent across *b*. Define

$$Q_{\alpha} = \frac{1}{B} \sum_{b=1}^{B} \mathbb{I} \left(S_{b}^{\theta} \le c_{\alpha} \right).$$
(A.6)

The Law of the Iterated Logarithm entails thatⁱⁱⁱ

$$\lim \inf_{B \to \infty} \lim_{\min(N,T,M_{\theta}) \to \infty} \sqrt{\frac{B}{2 \ln \ln B}} \frac{Q_{\alpha} - (1 - \alpha)}{\sqrt{\alpha (1 - \alpha)}} = -1.$$

Hence, we propose the following "strong rule" to decide in favour of H_0 :

$$Q_{\alpha} \ge (1-\alpha) - \sqrt{\alpha (1-\alpha)} \sqrt{\frac{2 \ln \ln B}{B}}.$$
(A.7)

Finally, we point out that, in our empirical exercises, we have used $f_{NT} = \left(\sqrt{T} \left| \hat{\theta} - c \right| \right)^{-1}$ and $M_{\theta} = T^{1/4}$.

The test above can be generalized in order to compare estimates of θ from different datasets. Indeed, let the subscripts 1 and 2 denote each dataset, and define the sample sizes as (N_1, T_1) and (N_2, T_2) , and the relevant estimates as $\hat{\theta}_1$ and $\hat{\theta}_2$. Then, upon defining a sequence $s_{N,T}^* = s_{N,T}^*(T_1, T_2)$ such that

$$\lim_{\min(N_1,T_1,N_2,T_2)\to\infty} s_{N,T}^* = \infty,$$
$$\lim_{\min(N_1,T_1,N_2,T_2)\to\infty} \frac{s_{N,T}^*}{\min\{T_1,T_2\}} \max\{\overline{v}_{N_1,T_1}(\epsilon), \overline{v}_{N_2,T_2}(\epsilon)\} = 0,$$

the test described above can be applied readily using

$$f_{N,T}^* = \frac{1}{\sqrt{\min\{T_1, T_2\}}} \left| \hat{\theta}_1 - \hat{\theta}_2 \right|,$$

e.g. using, in Step 1 of the algorithm, $M_{\theta} = \min \left\{ T_1^{1/4}, T_2^{1/4} \right\}$.

ⁱⁱⁱDetails on this argument are in Horváth and Trapani (2019).

A.3 Complements to Section 4.2: theory

Equations (15) and (16) in Section 4.2 contain two estimators of the population covariance matrices, defined as

$$\Sigma_{\mathcal{D}} = \frac{1}{T\pi_{\mathcal{S}}} \sum_{t=1}^{T} E\left(\widetilde{\mathbf{R}}_{t} \widetilde{\mathbf{R}}_{t}' d_{\mathcal{D},t} \left(\theta^{0}\right)\right), \qquad (A.8)$$

$$\Sigma_{\mathcal{S}} = \frac{1}{T\pi_{\mathcal{S}}} \sum_{t=1}^{T} E\left(\widetilde{\mathbf{R}}_{t} \widetilde{\mathbf{R}}_{t}' d_{\mathcal{S},t} \left(\theta^{0}\right)\right).$$
(A.9)

Recall that we denote as $g_j^{(i)}$ the *i*-th largest eigenvalue of $\Sigma_{\mathcal{D}}$ and $\Sigma_{\mathcal{S}}$ (according as $j = \mathcal{D}$ or \mathcal{S}). Lemma A.1. We assume that Assumptions 1-3 are satisfied. Then it holds that, for $j = \mathcal{D}, \mathcal{S}$

$$\underline{c}_{j}^{(i)}N \le g_{j}^{(i)} \le \overline{c}_{j}^{(i)}N, \text{ for } 0 \le i \le P_{j}^{0},$$

for some $0 < \underline{c}_j^{(i)} \leq \overline{c}_j^{(i)} < \infty$, and

$$g_j^{(i)} \le c_j^{(i)}, \text{ for } i \ge P_j^0 + 1,$$

for some $c_j^{(i)} < \infty$.

Theorem A.2. We assume that Assumptions 1-5 are satisfied. Then it holds that, for all $1 \le i \le N$ and j = D, S

$$\widehat{g}_{j}^{(i)} = g_{j}^{(i)} + o_{a.s.} \left(\frac{NT^{1/2}}{T\pi_{j}^{0}} \left(\ln N \right)^{1+\epsilon} \left(\ln T \right)^{1/2+\epsilon} \right),$$

for every $\epsilon > 0$.

Lemma A.1 and Theorem A.2 can be read together. According to Lemma A.1, the spectrum of each second moment matrix has a spiked structure; more specifically, the largest P_j^0 eigenvalues diverge to positive infinity as $N \to \infty$, whereas the others are bounded. The fact that the divergence rate is exactly N is a direct consequence of Assumption 3(iv) and also of having assumed $\eta = 1$. However, as long as the largest P_j^0 eigenvalues diverge, even at a slower rate, the arguments below are still valid although the construction of the tests may be more convoluted. Theorem A.2 offers the sample counterpart to Lemma A.1, giving a bound for the estimation error $\hat{g}_{j}^{(i)} - g_{j}^{(i)}$.^{iv}

Hereafter, we provide some comments, and the relevant asymptotics, on the individual tests for (17), viz.

$$\begin{cases} H_0: & g_j^{(i)} = c_j^{(i)} N \\ H_A: & g_j^{(i)} \le c_j^{(i)} \end{cases}$$

Since Theorem A.2 only contains rates, we propose a randomized version of the estimated eigenvalues $\hat{g}_{j}^{(i)}$, based on Trapani (2018). Based on Lemma A.1, each $g_{j}^{(i)}$ should diverge or not according to whether there are at least *i* common factors or fewer. Thus, we would expect the same separation result to also hold for $\hat{g}_{j}^{(i)}$. Indeed, whenever $g_{j}^{(i)}$ diverges as fast as N (which, in (17), represents the null hypothesis), $\hat{g}_{j}^{(i)}$ also does, at the same rate: this is an immediate consequence of Theorem A.2. Conversely, there is no guarantee that $\hat{g}_{j}^{(i)}$ converges when $g_{j}^{(i)}$ does so: in this case (which corresponds to the alternative hypothesis in (17)), from Theorem A.2, it is still possible that the estimation error $\hat{g}_{j}^{(i)} - g_{j}^{(i)}$ will diverge at a rate $NT^{-1/2}$, modulo some logarithmic terms.

Therefore, in order to ensure that $\hat{g}_j^{(i)}$ has the same separation result as its population counterpart $g_i^{(j)}$, we propose to use the statistic $\psi\left(\hat{g}_j^{(i)}\right)$ defined in (18). In (19), we have defined

$$\varrho_j = \begin{cases} \varepsilon & \text{when } \frac{\ln T_j}{\ln N} \ge \frac{1}{2} \\ 1 - \frac{1}{2} \frac{\ln T_j}{\ln N} + \varepsilon & \text{when } \frac{\ln T_j}{\ln N} < \frac{1}{2} \end{cases}$$

^{iv}As is typical in (large) Random Matrix Theory, there is no guarantee that $\hat{g}_{j}^{(i)} - g_{j}^{(i)}$ will drift to zero as min $(N, T) \to \infty$. See for example the related contributions by Bai and Yao (2008), Bai and Yao (2012) and Wang and Fan (2017), and also Lemma 2 in Trapani (2018), where a similar result is derived, under different assumptions and for a linear model. Indeed, $\hat{g}_{j}^{(i)} - g_{j}^{(i)}$ may even diverge to infinity. However, as long as $\hat{g}_{j}^{(i)} - g_{j}^{(i)}$ is of smaller magnitude than $g_{j}^{(i)}$ for $j \leq P_{j}^{0}$, the spiked structure of the spectrum of the second moment matrices is preserved, in that $\hat{g}_{i}^{(i)}$ is of smaller magnitude when $i > P_{i}^{0}$ compared to the case $i \leq P_{i}^{0}$.

in that $\widehat{g}_{j}^{(i)}$ is of smaller magnitude when $j > P_{j}^{0}$ compared to the case $j \le P_{j}^{0}$. ^vNote that we are not imposing that $0 < \pi_{\mathcal{D}}, \pi_{\mathcal{S}} < 1$; we are therefore entertaining the possibility that e.g. $\pi_{\mathcal{D}} = \pi_{\mathcal{D}}(T) \to 0$, which may be well suited to a situation in which one of the two regimes occurs very rarely.

The rationale for this choice is that we need to choose ρ_j such that

$$\lim_{\min(N,T)\to\infty} \frac{N^{1-\varrho_j} T^{1/2}}{T\pi_j} \left(\ln N\right)^{1+\epsilon} \left(\ln T\right)^{1/2+\epsilon} = 0;$$
(A.10)

(19) is therefore proposed under the condition $0 < \pi_{\mathcal{D}}, \pi_2 < 1$, although it is valid for any values of $\pi_{\mathcal{D}}$ and $\pi_{\mathcal{S}}$. If this is violated, in principle (A.10) can be employed to construct a refined proposal for ρ_j . Based on (19) and on the strong rates in Theorem A.2, it is easy to see that

$$\lim_{\min(N,T)\to\infty} \psi\left(\widehat{g}_{j}^{(i)}\right) = \infty \text{ under } H_{0},$$
$$\lim_{\min(N,T)\to\infty} \psi\left(\widehat{g}_{j}^{(i)}\right) = 1 \text{ under } H_{A}.$$

Let \varkappa be defined such that

$$0 < \varkappa < \frac{\underline{c}_j^{(i)}}{\frac{1}{N-p}\sum_{h=p+1}^N \overline{c}_j^{(h)}},$$

where $\underline{c}_{j}^{(i)}$ and the $\overline{c}_{j}^{(h)}$ s are defined in Lemma A.1. It holds that

Theorem A.3. We assume that Assumptions 1-6 are satisfied with $\eta = 1$ in Assumption 1. Let M = M(N), so that $\lim_{N\to\infty} M(N) = \infty$. Then, as $\min(N, T_j) \to \infty$ with

$$M \exp\left(-\varkappa N^{1-\varrho_j}\right) \to 0,$$
 (A.11)

it holds that under H_0

$$\Upsilon_j^{(i)} \xrightarrow{D^*} \chi_1^2, \tag{A.12}$$

for almost all realizations of $\{R_{i,t}, 1 \leq i \leq N, 1 \leq t \leq T\}$, $j = \mathcal{D}, \mathcal{S}$ and all $1 \leq i \leq N$. Under H_A , it holds that there exists a positive, finite constant c_0 such that

$$\frac{1}{4M}\Upsilon_j^{(i)} \xrightarrow{P^*} c_0, \tag{A.13}$$

for almost all realizations of $\{R_{i,t}, 1 \leq i \leq N, 1 \leq t \leq T\}$, $j = \mathcal{D}, \mathcal{S}$ and all $1 \leq i \leq N$.

The theorem, and the algorithm above, suggest that some tuning is needed prior to implementing the test. To begin with, note that the power of tests based on $\Upsilon_j^{(i)}$ increases with M by (A.13); conversely, in the proof of (A.12) it is shown that the test statistic has a non-centrality parameter which is controlled by (A.11), and which vanishes faster the smaller M. Thus, the choice of M reflects the well-known size-power trade-off. In Section B and in the empirics, we use M = N, which satisfies (A.11) and is our recommended choice.

In Step 2, we recommend to draw the values of s from a standard normal. By integrating s out in Step 4 of the algorithm, the statistic $\Upsilon_j^{(i)}$ becomes scale-invariant, in the sense that $\Upsilon_j^{(i)}$ does not depend on the support of $s.^{\text{vi}}$ This form of scale invariance is clearly desirable. From a practical point of view, we propose to use a Gauss-Hermite quadrature to approximate the integral that defines $\Upsilon_j^{(i)}$, viz.

$$\widehat{\Upsilon}_{j}^{(i)} = \frac{1}{\sqrt{\pi}} \sum_{i=1}^{n_{S}} w_{i} \left(v_{j}^{(i)} \left(\sqrt{2}x_{i} \right) \right)^{2}, \qquad (A.14)$$

where the x_i , $1 \le i \le n_S$, are the zeros of the Hermite polynomial $H_{n_S}(x)$ and

$$w_i = \frac{\sqrt{\pi} 2^{n_S - 1} (n_S - 1)!}{n_S \left[H_{n_S - 1} \left(\sqrt{2} x_i \right) \right]^2}.$$
(A.15)

Thus, when constructing $v_j^{(i)}(s)$, we construct n_S of these statistics, each with $s = \sqrt{2}x_i$.^{vii}

^{vi}We point out that, in the original paper by Trapani (2018), this is not the case since the methodology is implemented by extracting several values of s from a uniform distribution with support, say, S: doing this makes the test statistics dependent on S, thus making them not scale invariant.

^{vii}The values of the roots x_i , and of the corresponding weights w_i , are tabulated e.g. in Salzer, Zucker, and Capuano (1952).

A.4 Complements to Section 5.2

A.4.1 Assumptions

For $j = \mathcal{D}, \mathcal{S}$, consider the matrices $M_{\mathbf{u}}^{(j)} = \lim_{T \to \infty} T^{-1} \sum_{t=1}^{T} \mathbf{u}_{j,t}^{0} \mathbf{u}_{j,t}^{0'} d_{j,t} (\theta^{0})$. We then define

$$\begin{split} \Sigma_{\gamma,j} &= c_{j}^{-2} \mathbf{\Lambda}_{j}^{0} V_{\mathbf{u},j} \mathbf{\Lambda}_{j}^{0\prime} + \left(\left(\gamma_{j,1}^{0\prime} \left(M_{\mathbf{u}}^{(j)} \right)^{-1} \right) \otimes I_{K} \right) V_{\mathbf{eu},j} \left(\left(\left(M_{\mathbf{u}}^{(j)} \right)^{-1} \gamma_{j,1}^{0} \right) \otimes I_{K} \right) \\ &+ c_{j}^{-1} \mathbf{\Lambda}_{j}^{0} \Pi_{\mathbf{ue},j} \left(\left(\left(M_{\mathbf{u}}^{(j)} \right)^{-1} \gamma_{j,1}^{0} \right) \otimes I_{K} \right) + c_{j}^{-1} \left(\left(\gamma_{j,1}^{0\prime} \left(M_{\mathbf{u}}^{(j)} \right)^{-1} \right) \otimes I_{K} \right) \Pi_{\mathbf{ue},j}^{\prime} \mathbf{\Lambda}_{j}^{0\prime}, \\ \Sigma_{\alpha\gamma,j} &= \sigma_{\alpha,j}^{2} \mathbf{\Lambda}_{j}^{0} \left[\frac{1}{N} \mathbf{B}_{j}^{0\prime} \mathbf{B}_{j}^{0} - \left(\frac{1}{N} \mathbf{B}_{j}^{0\prime} i_{N} \right) \left(\frac{1}{N} \mathbf{B}_{j}^{0\prime} i_{N} \right)^{\prime} \right]^{-1} \mathbf{\Lambda}_{j}^{0\prime}, \end{split}$$

where i_N is an $N \times 1$ vector of ones.

A.4.2 Estimation of long-run covariance matrices

We propose the following estimators

$$\widehat{\Sigma}_{\gamma,j} = c_j^{-2} \widehat{\Lambda}_j \widehat{V}_{\mathbf{u},j} \widehat{\Lambda}'_j + \left(\left(\widehat{\gamma}'_{j,1} \left(\widehat{M}_{\mathbf{u}}^{(j)} \right)^{-1} \right) \otimes I_K \right) \widehat{V}_{\mathbf{eu},j} \left(\left(\left(\left(\widehat{M}_{\mathbf{u}}^{(j)} \right)^{-1} \widehat{\gamma}_{j,1} \right) \otimes I_K \right) + c_j^{-1} \widehat{\Lambda}_j \widehat{\Pi}_{\mathbf{ue},j} \left(\left(\left(\left(\widehat{M}_{\mathbf{u}}^{(j)} \right)^{-1} \widehat{\gamma}_{j,1} \right) \otimes I_K \right) + c_j^{-1} \left(\left(\left(\widehat{\gamma}'_{j,1} \left(\widehat{M}_{\mathbf{u}}^{(j)} \right)^{-1} \right) \otimes I_K \right) \widehat{\Pi}'_{\mathbf{ue},j} \widehat{\Lambda}'_j, \\ \widehat{\Sigma}_{\alpha\gamma,j} = \widehat{\sigma}_{\alpha,j}^2 \widehat{\Lambda}_j \left[\frac{1}{N} \widehat{\mathbf{B}}_j^{\widehat{P}_j'} \widehat{\mathbf{B}}_j^{\widehat{P}_j} - \left(\frac{1}{N} \widehat{\mathbf{B}}_j^{\widehat{P}_j'} i_N \right) \left(\frac{1}{N} \widehat{\mathbf{B}}_j^{\widehat{P}_j'} i_N \right)' \right]^{-1} \widehat{\Lambda}'_j,$$
(A.16)
(A.16)

In (A.16), recall that $\widehat{\mathbf{X}}_j = \left(\iota_N, \widehat{\mathbf{B}}_j^{\widehat{P}_j}\right)$. Also, we have used

$$\widehat{M}_{\mathbf{u}}^{(j)} = \frac{1}{T} \sum_{t=1}^{T} \widehat{\mathbf{u}}_{j,t} \widehat{\mathbf{u}}_{j,t}' \widehat{d}_{j,t}, \qquad (A.18)$$

$$\widehat{V}_{\mathbf{u},j} = \widehat{m}_{\mathbf{u},0}^{(j)} + \sum_{k=1}^{h_m} \left(1 - \frac{k}{h_m + 1} \right) \left(\widehat{m}_{\mathbf{u},k}^{(j)} + \widehat{m}_{\mathbf{u},k}^{(j)\prime} \right), \tag{A.19}$$

$$\widehat{V}_{\mathbf{eu},j} = \widehat{m}_{\mathbf{eu},0}^{(j)} + \sum_{k=1}^{h_m} \left(1 - \frac{k}{h_m + 1} \right) \left(\widehat{m}_{\mathbf{eu},k}^{(j)} + \widehat{m}_{\mathbf{eu},k}^{(j)\prime} \right), \tag{A.20}$$

$$\widehat{\Pi}'_{\mathbf{u}\mathbf{e},j} = \frac{1}{T} \sum_{t=1}^{T} \left(vec\left(\widehat{\mathbf{e}}_{t}\widehat{\mathbf{u}}'_{j,t}\right) \right) \widehat{\mathbf{u}}'_{j,t} \widehat{d}_{j,t} + \sum_{k=1}^{h_{m}} \left(1 - \frac{k}{h_{m}+1} \right) \left(\widehat{m}^{(j)}_{\mathbf{\Pi},k} + \widetilde{m}^{(j)}_{\mathbf{\Pi},k} \right), \quad (A.21)$$

with

$$\begin{aligned} \widehat{m}_{\mathbf{u},k}^{(j)} &= \frac{1}{T} \sum_{t=k+1}^{T} \left(\widehat{\mathbf{u}}_{j,t} \widehat{d}_{j,t} \right) \left(\widehat{\mathbf{u}}_{j,t-k} \widehat{d}_{j,t-k} \right)', \\ \widehat{m}_{\mathbf{eu},k}^{(j)} &= \frac{1}{T} \sum_{t=k+1}^{T} \left(\operatorname{vec} \left(\widehat{\mathbf{e}}_{t} \widehat{\mathbf{u}}'_{j,t} \right) \widehat{d}_{j,t} \right) \left(\operatorname{vec} \left(\widehat{\mathbf{e}}_{t-k} \widehat{\mathbf{u}}'_{j,t-k} \right) \widehat{d}_{j,t-k} \right)', \\ \widehat{m}_{\mathbf{\Pi},k}^{(j)} &= \frac{1}{T} \sum_{t=k+1}^{T} \left(\operatorname{vec} \left(\widehat{\mathbf{e}}_{t-k} \widehat{\mathbf{u}}'_{j,t-k} \right) \widehat{d}_{j,t-k} \right) \left(\widehat{\mathbf{u}}_{j,t} \widehat{d}_{j,t} \right)', \\ \widetilde{m}_{\mathbf{\Pi},k}^{(j)} &= \frac{1}{T} \sum_{t=k+1}^{T} \left(\operatorname{vec} \left(\widehat{\mathbf{e}}_{t} \widehat{\mathbf{u}}'_{j,t-k} \right) \widehat{d}_{j,t-k} \right) \left(\widehat{\mathbf{u}}_{j,t-k} \widehat{d}_{j,t-k} \right)', \end{aligned}$$

for k = 0, 1, ... In (A.19)-(A.20), we use $h_m = O(T^{1/3})$, and $\hat{\mathbf{e}}_t = \mathbf{g}_t - (\hat{\mathbf{a}} + \hat{\mathbf{\Lambda}}\hat{\mathbf{u}}_t)$. Finally, in order to compute $\hat{\sigma}^2_{\alpha,j}$, let

$$\widehat{e}_{j,i,t} = R_{i,t}\widehat{d}_{j,t} - \left[\widehat{\gamma}_{j,0} + \widehat{\beta}'_{j,i}\left(\widehat{\gamma}_{j,1} + \widehat{\mathbf{u}}_{j,t}\right)\right]\widehat{d}_{j,t},$$

and define $\hat{e}_{j,i} = \hat{T}_j^{-1} \sum_{t=1}^T \hat{e}_{j,i,t}$ and

$$\widehat{\sigma}_{\alpha,j}^2 = \frac{1}{N} \sum_{i=1}^N \widehat{e}_{j,i}^2.$$

All the estimators defined above are consistent - in order to save space, we omit the proofs, which is based on standard, if tedious, calculations based on using the results reported in the above. The only result that does not follow immediately is the consistency of $\hat{\sigma}_{\alpha,j}^2$, which we state as a lemma.

Lemma A.2. We assume that Assumptions 1-10 hold. Then, as $\min(N, T_j) \to \infty$ for $j = \mathcal{D}, \mathcal{S}$ under (24) and (25), it holds that $\hat{\sigma}_{\alpha,j}^2 = \sigma_{\alpha,j}^2 + o_{P^*}(1)$, for almost all realizations of $\{R_{i,t}, 1 \leq i \leq N, 1 \leq t \leq T\}$.

A.4.3 Estimation of Γ_j and Λ_j : asymptotics

Define the $P_j^0 \times T$ matrices of regime-specific factor innovations $\mathbf{U}_j(\theta) = [\mathbf{u}_{j,1}(\theta), \dots, \mathbf{u}_{j,T}(\theta)],$ for $j = \mathcal{D}, \mathcal{S}$, such that $\mathbf{U}_{\mathcal{D}}(\theta) + \mathbf{U}_{\mathcal{S}}(\theta) = (\mathbf{u}_1, \dots, \mathbf{u}_T) = \mathbf{U}.$

As is well known, Γ_j (and Λ_j) cannot be estimated consistently, but only up to a rotation.

Define the $P_j^0 \times \hat{P}_j$ rotation matrix

$$\widehat{\mathbf{H}}_{jj}^{\widehat{P}_{j}}\left(\theta\right) = \frac{\mathbf{U}_{j}\left(\theta^{0}\right)\mathbf{U}_{j}\left(\theta\right)'}{T}\frac{\mathbf{B}_{j}^{0'}\widehat{\mathbf{B}}_{j}\left(\theta\right)}{N}\widehat{\mathbf{V}}_{j}\left(\theta\right)^{-1},\tag{A.22}$$

for $j = \mathcal{D}, \mathcal{S}$, where $\widehat{\mathbf{V}}_{j}(\theta)$ is the $\widehat{P}_{j} \times \widehat{P}_{j}$ diagonal matrix of the first \widehat{P}_{j} largest eigenvalues of $\widehat{\boldsymbol{\Sigma}}_{j,\widetilde{\mathbf{R}}}(\theta)$ defined in (12). Define also

$$\widehat{\mathbf{H}}_{jj}^{\mathbf{\Gamma}\widehat{P}_{j}}\left(\theta\right) = \begin{bmatrix} 1 & \mathbf{0}_{\widehat{P}_{j}}'\\ \mathbf{0}_{\widehat{P}_{j}} & \widehat{\mathbf{H}}_{jj}^{\widehat{P}_{j}}\left(\theta\right) \end{bmatrix},\tag{A.23}$$

for $j = \mathcal{D}, \mathcal{S}$.

For short, we define $\widehat{\mathbf{H}}_{jj}(\theta)$ as $\widehat{\mathbf{H}}_{jj}^{\widehat{P}_{j}}(\theta)$ when calculated with P_{j}^{0} common factors, and $\widehat{\mathbf{H}}_{jj}^{\Gamma}(\theta)$ similarly. Finally, we define $\widehat{\mathbf{H}}_{jj}(\theta^{0}) = \widehat{\mathbf{H}}_{jj}$ and $\widehat{\mathbf{H}}_{jj}^{\Gamma}(\theta^{0}) = \widehat{\mathbf{H}}_{jj}^{\Gamma}$.

Theorem A.4. We assume that Assumptions 1-8 are satisfied. Then, as $\min(N, T_j) \to \infty$, it holds that

$$\widehat{\boldsymbol{\Gamma}}_{j} - \left(\widehat{\mathbf{H}}_{jj}^{\boldsymbol{\Gamma}}\left(\boldsymbol{\theta}^{0}\right)\right)^{-1} \boldsymbol{\Gamma}_{j}^{0} = o_{P}\left(1\right) + o_{P^{*}}\left(1\right), \qquad (A.24)$$

$$\widehat{\mathbf{\Lambda}}_{j} - \mathbf{\Lambda}_{j}^{0} \widehat{\mathbf{H}}_{jj} \left(\theta^{0} \right) = o_{P} \left(1 \right) + o_{P^{*}} \left(1 \right), \qquad (A.25)$$

for $j = \mathcal{D}, \mathcal{S}$, for almost all realizations of $\{R_{i,t}, 1 \leq i \leq N, 1 \leq t \leq T\}$.

For $j = \mathcal{D}, \mathcal{S}$, consider the matrices: $M_{\mathbf{H},\mathbf{X}}^{(j)} = \lim_{N \to \infty} N^{-1} \widehat{\mathbf{H}}_{jj}^{\Gamma'}(\theta^0) \mathbf{X}_j^{0'} \mathbf{X}_j^0 \widehat{\mathbf{H}}_{jj}^{\Gamma}(\theta^0); M_{\mathbf{H},\mathbf{X},\mathbf{B}}^{(j)} = \lim_{N \to \infty} N^{-1} \widehat{\mathbf{H}}_{jj}^{\Gamma'}(\theta^0) \mathbf{X}_j^{0'} \mathbf{B}_j^0; \text{ and } M_{\mathbf{H},\mathbf{u}}^{(j)} = \left(M_{\mathbf{u}}^{(j)}\right)^{-1} \widehat{\mathbf{H}}_{jj}(\theta^0).$ We then define

$$\Sigma_{\Gamma,j} = c_j^{-2} \left(M_{\mathbf{H},\mathbf{X}}^{(j)} \right)^{-1} M_{\mathbf{H},\mathbf{X},\mathbf{B}}^{(j)} V_{\mathbf{u},j} M_{\mathbf{H},\mathbf{X},\mathbf{B}}^{(j)\prime} \left(M_{\mathbf{H},\mathbf{X}}^{(j)} \right)^{-1},$$

$$\Sigma_{\alpha\Gamma,j} = \sigma_{\alpha,j}^2 \left(M_{\mathbf{H},\mathbf{X}}^{(j)} \right)^{-1},$$

$$\Sigma_{\Lambda,j} = \left(M_{\mathbf{H},\mathbf{u}}^{(j)\prime} \otimes I_K \right) V_{\mathbf{eu},j} \left(M_{\mathbf{H},\mathbf{u}}^{(j)} \otimes I_K \right).$$

Theorem A.5. We assume that Assumptions 1-8 are satisfied, and that $\min(N, T_j) \to \infty$, with (24) and (25). Then, under Assumptions 7 and 9, it holds that

$$\left(\frac{1}{T}\Sigma_{\Gamma,j} + \frac{1}{N}\Sigma_{\alpha\Gamma,j}\right)^{-1/2} \left(\widehat{\mathbf{\Gamma}}_{j} - \left(\widehat{\mathbf{H}}_{jj}^{\mathbf{\Gamma}}\left(\theta^{0}\right)\right)^{-1} \mathbf{\Gamma}_{j}^{0}\right) \xrightarrow{D^{*}} N\left(0, I_{P_{j}+1}\right).$$
(A.26)

Under Assumption 9(ii), if $\frac{T^{1/2}}{N} \rightarrow 0$, it holds that

$$\left(\frac{1}{T}\Sigma_{\Lambda,j}\right)^{-1/2} vec\left(\widehat{\Lambda}_{j} - \Lambda_{j}^{0}\widehat{\mathbf{H}}_{jj}\left(\theta^{0}\right)\right) \xrightarrow{D^{*}} N\left(0, I_{P_{j}K}\right),\tag{A.27}$$

for $j = \mathcal{D}, \mathcal{S}$, for almost all realizations of $\{R_{i,t}, 1 \leq i \leq N, 1 \leq t \leq T\}$.

In order to use the results in Theorem A.5, we propose the following estimators

$$\begin{split} \widehat{\Sigma}_{\Gamma,j} = & c_j^{-2} \left(\frac{\widehat{\mathbf{X}}_j' \widehat{\mathbf{X}}_j}{N} \right)^{-1} \left(\frac{\widehat{\mathbf{X}}_j' \widehat{\mathbf{B}}_j^{\widehat{P}_j}}{N} \right) \widehat{V}_{\mathbf{u},j} \left(\frac{\widehat{\mathbf{X}}_j' \widehat{\mathbf{B}}_j^{\widehat{P}_j}}{N} \right)' \left(\frac{\widehat{\mathbf{X}}_j' \widehat{\mathbf{X}}_j}{N} \right)^{-1} \\ \widehat{\Sigma}_{\alpha\Gamma,j} = & \widehat{\sigma}_{\alpha,j}^2 \left(\frac{\widehat{\mathbf{X}}_j' \widehat{\mathbf{X}}_j}{N} \right)^{-1}, \\ \widehat{\Sigma}_{\Lambda,j} = & \left(\left(\widehat{M}_{\mathbf{u}}^{(j)} \right)^{-1} \otimes I_K \right) \widehat{V}_{\mathbf{eu},j} \left(\left(\widehat{M}_{\mathbf{u}}^{(j)} \right)^{-1} \otimes I_K \right), \end{split}$$

whose consistency can be shown along the same lines as for the other estimators.

A.4.4 Asymptotics for the zero beta rate

The following lemma (which can be compared with Theorem 3 in Giglio and Xiu, 2021) characterizes the distribution of $\hat{\gamma}_{j,0}$ and the test for (27).

Theorem A.6. We assume that Assumptions 1-10 hold, and that $E(\alpha_{i,\mathcal{D}}\alpha_{k,\mathcal{S}}) = 0$ for all $1 \leq i, k \leq N$. Then, as $\min(N, T_j) \to \infty$ with $\frac{N}{T_j^2} \to 0$, it holds that

$$\frac{\widehat{\gamma}_{j,0} - \gamma_{j,0}}{\left(\sigma_{\alpha,j}^{2} \left(i_{N}^{\prime} \mathbf{M}_{B,j}^{0} i_{N}\right)^{-1}\right)^{1/2}} \xrightarrow{D^{*}} N\left(0,1\right),\tag{A.28}$$

for j = D, S, and

$$\frac{\left(\widehat{\pi}_{\mathcal{D}}\widehat{\gamma}_{\mathcal{D},0} + \widehat{\pi}_{\mathcal{S}}\widehat{\gamma}_{\mathcal{S},0}\right) - \left(\pi_{\mathcal{D}}\gamma_{\mathcal{D},0} + \pi_{\mathcal{S}}\gamma_{\mathcal{S},0}\right)}{\left(\pi_{\mathcal{D}}^{2}\sigma_{\alpha,\mathcal{D}}^{2}\left(i_{N}^{\prime}\mathbf{M}_{B,\mathcal{D}}^{0}i_{N}\right)^{-1} + \pi_{\mathcal{S}}^{2}\sigma_{\alpha,\mathcal{S}}^{2}\left(i_{N}^{\prime}\mathbf{M}_{B,\mathcal{S}}^{0}i_{N}\right)^{-1}\right)^{1/2}} \xrightarrow{D^{*}} N\left(0,1\right),$$
(A.29)

for almost all realizations of $\{R_{i,t}, 1 \leq i \leq N, 1 \leq t \leq T\}$, where $\mathbf{M}_{B,j}^0 = \mathbf{B}_j^0 \left(\mathbf{B}_j^{0'} \mathbf{B}_j^0\right)^{-1} \mathbf{B}_j^{0'}$ for $j = \mathcal{D}, \mathcal{S}$.

A.4.5 Testing for zero weighted intercept

We consider the following result, which can be used e.g. to test for

$$H_0: \pi_{\mathcal{D}}\gamma_{\mathcal{D},g} + \pi_{\mathcal{S}}\gamma_{\mathcal{S},g} = 0.$$

The result is a direct application of Assumption 11, and we therefore report it without proof. Define

$$\widetilde{\Sigma}_{\alpha,\gamma} = \pi_{\mathcal{D}}^2 \sigma_{\alpha,\mathcal{D}}^2 \Lambda_{\mathcal{D}} \left(\frac{B_{\mathcal{D}}' M_{iN} B_{\mathcal{D}}}{N} \right)^{-1} \Lambda_{\mathcal{D}}' + \pi_{\mathcal{S}}^2 \sigma_{\alpha,\mathcal{S}}^2 \Lambda_{\mathcal{S}} \left(\frac{B_{\mathcal{S}}' M_{iN} B_{\mathcal{S}}}{N} \right)^{-1} \Lambda_{\mathcal{S}}'$$

and

$$\begin{split} \widetilde{\Sigma} &= \pi_{\mathcal{D}}^{2} \Sigma_{\gamma,\mathcal{D}} + \pi_{\mathcal{S}}^{2} \Sigma_{\gamma,\mathcal{S}} + \pi_{\mathcal{D}} \pi_{\mathcal{S}} c_{\mathcal{D}}^{-1} c_{\mathcal{S}}^{-1} \left(\Lambda_{\mathcal{D}} \Pi'_{u,\mathcal{D};u,\mathcal{S}} \Lambda'_{\mathcal{D}} + \Lambda_{\mathcal{S}} \Pi_{u,\mathcal{D};u,\mathcal{S}} \Lambda'_{\mathcal{S}} \right) \\ &+ \pi_{\mathcal{D}} \pi_{\mathcal{S}} \left[\left(\gamma'_{\mathcal{D},1} \cdot M_{Hu}^{(\mathcal{D})'} \right) \otimes I_{K} \right] \Pi'_{ue,\mathcal{D}\mathcal{S}} \left[\left(\gamma_{\mathcal{S},1} \cdot M_{Hu}^{(\mathcal{D})} \right) \otimes I_{K} \right] \\ &+ \left(\pi_{\mathcal{D}} \pi_{\mathcal{S}} \left[\left(\gamma'_{\mathcal{D},1} \cdot M_{Hu}^{(\mathcal{D})'} \right) \otimes I_{K} \right] \Pi'_{ue,\mathcal{D}\mathcal{S}} \left[\left(\gamma_{\mathcal{S},1} \cdot M_{Hu}^{(\mathcal{D})} \right) \otimes I_{K} \right] \right)' \\ &+ \pi_{\mathcal{D}} \pi_{\mathcal{S}} c_{\mathcal{D}}^{-1} \Lambda_{\mathcal{D}} \Pi'_{ue,\mathcal{S};u,\mathcal{D}} \left[\left(\gamma_{\mathcal{S},1} \cdot M_{Hu}^{(\mathcal{D})} \right) \otimes I_{K} \right] + \left(\pi_{\mathcal{D}} \pi_{\mathcal{S}} c_{\mathcal{D}}^{-1} \Lambda_{\mathcal{D}} \Pi'_{ue,\mathcal{S};u,\mathcal{D}} \left[\left(\gamma_{\mathcal{S},1} \cdot M_{Hu}^{(\mathcal{D})} \right) \otimes I_{K} \right] \right)' \\ &+ \pi_{\mathcal{D}} \pi_{\mathcal{S}} c_{\mathcal{S}}^{-1} \left[\left(\gamma_{\mathcal{D},1} \cdot M_{Hu}^{(\mathcal{S})} \right) \otimes I_{K} \right]' \Pi'_{ue,\mathcal{D};u,\mathcal{S}} \Lambda'_{\mathcal{S}} + \left(\pi_{\mathcal{D}} \pi_{\mathcal{S}} c_{\mathcal{S}}^{-1} \left[\left(\gamma_{\mathcal{D},1} \cdot M_{Hu}^{(\mathcal{S})} \right) \otimes I_{K} \right]' \Pi'_{ue,\mathcal{D};u,\mathcal{S}} \Lambda'_{\mathcal{S}} \right)'. \end{split}$$

Theorem A.7. We assume that Assumptions 1-11 hold, and that $E(\alpha_{i,\mathcal{D}}\alpha_{k,\mathcal{S}}) = 0$ for all $1 \leq i, k \leq N$. Then, as $\min(N, T_j) \to \infty$ with $\frac{N}{T_j^2} \to 0$, it holds that

$$\left(\frac{1}{N}\widetilde{\Sigma}_{\alpha,\gamma} + \frac{1}{T}\widetilde{\Sigma}\right)^{-1/2} \left[\left(\widehat{\pi}_{\mathcal{D}}\widehat{\gamma}_{\mathcal{D},g} + \widehat{\pi}_{\mathcal{S}}\widehat{\gamma}_{\mathcal{S},g}\right) - \left(\pi_{\mathcal{D}}\gamma_{\mathcal{D},g} + \pi_{\mathcal{S}}\gamma_{\mathcal{S},g}\right)\right] \xrightarrow{d} N\left(0, I_{K}\right)$$

The long-run covariance matrices that appear in the theorem can be estimated as

$$\begin{split} \widehat{\Pi}'_{u,\mathcal{D};u,S} &= \frac{1}{T} \sum_{t=1}^{T} \left(\widehat{u}_{\mathcal{D},t} \widehat{u}'_{\mathcal{S},t} \widehat{d}_{\mathcal{D},t} \widehat{d}_{\mathcal{S},t} \right) + \sum_{k=1}^{h_m} \left(1 - \frac{k}{h_m + 1} \right) \left(\widehat{m}_{\Pi,k}^{(\mathcal{D}S)} + \widetilde{m}_{\Pi,k}^{(\mathcal{D}S)} \right) \\ &= \sum_{k=1}^{h_m} \left(1 - \frac{k}{h_m + 1} \right) \left(\widehat{m}_{\Pi,k}^{(\mathcal{D}S)} + \widetilde{m}_{\Pi,k}^{(\mathcal{D}S)} \right), \end{split}$$

with

$$\widehat{m}_{\Pi,k}^{(\mathcal{DS})} = \frac{1}{T} \sum_{t=k+1}^{T} \widehat{u}_{\mathcal{S},t} \widehat{u}_{\mathcal{S},t-k}' \widehat{d}_{\mathcal{S},t} \widehat{d}_{\mathcal{S},t-k},$$
$$\widetilde{m}_{\Pi,k}^{(\mathcal{DS})} = \frac{1}{T} \sum_{t=k+1}^{T} \widehat{u}_{\mathcal{S},t-k} \widehat{u}_{\mathcal{S},t}' \widehat{d}_{\mathcal{S},t-k} \widehat{d}_{\mathcal{S},t},$$

and

$$\begin{aligned} \widehat{\Pi}'_{ue,\mathcal{S};u,\mathcal{D}} &= \frac{1}{T} \sum_{t=1}^{T} \widehat{u}_{\mathcal{D},t} \widehat{d}_{\mathcal{D},t} vec \left(\widehat{e}_t \widehat{u}_{\mathcal{S},t} \widehat{d}_{\mathcal{S},t} \right)' + \sum_{k=1}^{h_m} \left(1 - \frac{k}{h_m + 1} \right) \left(\widehat{m}_{\Pi,ue,\mathcal{SD},k}^{(\mathcal{DS})} + \widetilde{m}_{\Pi,ue,\mathcal{SD},k}^{(\mathcal{DS})} \right) \\ &= \sum_{k=1}^{h_m} \left(1 - \frac{k}{h_m + 1} \right) \left(\widehat{m}_{\Pi,ue,\mathcal{SD},k}^{(\mathcal{DS})} + \widetilde{m}_{\Pi,ue,\mathcal{SD},k}^{(\mathcal{DS})} \right), \end{aligned}$$

where

$$\widehat{m}_{\Pi,ue,\mathcal{SD},k}^{(\mathcal{DS})} = \frac{1}{T} \sum_{t=k+1}^{T} \widehat{u}_{\mathcal{D},t} vec\left(\widehat{e}_{t-k}\widehat{u}_{\mathcal{S},t-k}\right)' \widehat{d}_{\mathcal{D},t}\widehat{d}_{\mathcal{S},t-k},$$
$$\widetilde{m}_{\Pi,ue,\mathcal{SD},k}^{(\mathcal{DS})} = \frac{1}{T} \sum_{t=k+1}^{T} \widehat{u}_{\mathcal{D},t-k} vec\left(\widehat{e}_{t}\widehat{u}_{\mathcal{S},t}\right)' \widehat{d}_{\mathcal{D},t-k}\widehat{d}_{\mathcal{S},t},$$

etc.

B Simulations

We evaluate the performance of (i) the Least Squares estimator of θ , and (ii) the sequential procedure to determine the number of common factors.ⁱ

Data are generated according to (5), that is

$$R_{i,t} = \mathbb{I} \left(\mathcal{D}_t \right) \left(\gamma_{\mathcal{D},0} + \alpha_{\mathcal{D},i} + \beta'_{\mathcal{D},i} \gamma_{\mathcal{D},1} + \beta'_{\mathcal{D},i} \mathbf{u}_{\mathcal{D},t} \right) + \mathbb{I} \left(\mathcal{S}_t \right) \left(\gamma_{\mathcal{S},0} + \alpha_{\mathcal{S},i} + \beta'_{\mathcal{S},i} \gamma_{\mathcal{S},1} + \beta'_{\mathcal{S},i} \mathbf{u}_{\mathcal{S},t} \right) + \sqrt{0.5} \epsilon_{i,t}.$$
(B.1)

In this data generating process (DGP), we generate all variables as having zero mean for simplicity. Where possible and relevant, we calibrate the values of parameters to values that arise from empirical analysis. In particular, as far as the downside risk is concerned, we set $\theta^0 = -0.03$, which is the value set in Farago and Tédongap (2018) and Lettau, Maggiori, and Weber (2014). Similarly, we generate z_t as i.i.d.N (μ_z, σ_z^2). Based on our data, we set $\sigma_z = 0.044$; also, in Section 6 we found that $P(z_t \leq \theta^0) \simeq 0.15$, which corresponds to a value $\mu_z = -0.015$. The rest of the DGP is specified as follows. For $j = \mathcal{D}, \mathcal{S}$, we generate $\mathbf{u}_{j,t}$ as i.i.d. N(0,1) for $1 \leq t \leq T$; similarly, we generate $\beta_{i,j}$ as i.i.d. N(0,1) for $1 \leq i \leq N$; finally, $\alpha_{j,i}$ is also i.i.d. N(0,1) for $1 \leq i \leq N$.ⁱⁱ The number of replications is set to 1,000.

The specifications of our estimators and tests are as follows. In the estimation of θ , we use a grid search on the interval [-0.08, -0.04], with steps of size 0.001. In the estimation of P_j , we use the algorithm described in Section 4.2, with: M = N, $\varepsilon = 0.01$ in (19), $n_S = 4$, and the nominal level of the individual tests set to $\alpha = \frac{0.05}{T}$.ⁱⁱⁱ

ⁱAs far as the latter is concerned, the results complement the ones in Trapani (2018), which considers the case of a linear model.

ⁱⁱWe note that we have carried out some limited experiments where, e.g., a vanishing proportion of the $\alpha_{j,i}$ s had nonzero mean, but results did not change. As far as the idiosyncratic innovation $\epsilon_{i,t}$ is concerned, we only consider the case of no cross sectional independence in our simulations; thus, we generate $\epsilon_{i,t}$ as *i.i.d.* N(0, 0.5).

ⁱⁱⁱWe point out that altering the choice of M within a reasonable range does not have any impact on the final results. Indeed, we tried M = N/2 and M = 2N, without noting any changes. This robustness is in line with the simulations in Trapani (2018).

Similarly, we have set ρ_j according to (19), with $\varepsilon = 0.01$; based on Theorem A.3, ε does not need to be too large, since its purpose is to smooth away a slowly varying sequence. Even in this case, as the theory prescribes, the impact of ρ_j is minimal; we have tried, as robustness check, $\varepsilon = 0.05$ and $\varepsilon = 0.1$, and results virtually do not change.

Results are reported for (N, T) = (57, 316) and (130, 666), which correspond to the sample sizes in our empirical applications. Further simulations, which reinforce the findings in this section, are available upon request. Similarly, we consider various number of common factors which mimic the findings in our empirical applications: $P_{\mathcal{D}}^0 \in \{1, 2\}$ and $P_{\mathcal{S}}^0 \in \{1, 2, 3, 4, 5, 6\}$.^{iv}

Consider the estimation of θ , whose performance, in terms of bias and RMSE, we report in Table B.1. Results show that both indicators decrease with T, as expected, and also with N, which is due to a second order improvement in the estimation of the factor structure. We note that $\hat{\theta}$ tends to be biased towards zero, with this vanishing as $(N, T) \to \infty$.

As far as estimation of the number of common factors is concerned, the results are in Table B.2, where we have reported two indicators: (i) the average estimated number of factors

$$\overline{\widehat{P}_j} = \frac{1}{1000} \sum_{r=1}^{1000} \widehat{P}_{j,r},$$

 $j = \mathcal{D}, \mathcal{S}$, and *(ii)* the percentage of times that $\hat{P}_{j,r} \neq P_j^0$, $j = \mathcal{D}, \mathcal{S}$. Results are very satisfactory, with the exception of the case $(P_{\mathcal{D}}^0, P_{\mathcal{S}}^0) = (2, 1)$, when (N, T) = (57, 316). This case is however unlikely to occur, since our experience indicates that the number of common factors in the downside regime tends to be lower than the number of common factors in the other regime. In all other cases, the estimated numbers of factors is always close to the true number of factors, and the simulation results provide comfort that the number of common factors is precisely identified by our procedure even in moderately-sized cross-sections of asset returns.

^{iv}Therefore we do not consider the case where there is no factor at all (i.e., either $P_{\mathcal{D}}^0 = 0$ or $P_{\mathcal{S}}^0 = 0$, or both), requiring that at least one factor should always be present in asset returns, which seems reasonable (see Lettau and Pelger (2020) for a related discussion).

Table B.1: Bias and RMSE for $\hat{\theta}$

Panel A: $\theta = -0.03$								
		Bi	as					
(N,T) = (57,316) $(N,T) = (130,666)$								
$P_{\mathcal{S}} \setminus P_{\mathcal{D}}$	1	2	$P_{\mathcal{S}} \setminus P_{\mathcal{D}}$	1	2			
1	0.001	0.000	1	0.000	0.000			
2	0.000	0.000	2	0.000	0.000			
3	0.000	0.000	3	0.000	0.000			
4	0.000	0.000	4	0.000	0.000			
5	0.000	0.000	5	0.000	0.000			
6	0.000	0.000	6	0.000	0.000			
		ΒN	ISE					
(N, 7)	T) = (57.3)	316)	(N,T)	=(130,	666)			
$P_S \setminus P_D$	1	2	$P_S \setminus P_T$	1	2			
	0.003	0.002		0.003	0.000			
2	0.000	0.000	2	0.000	0.000			
3	0.000	0.000	3	0.000	0.000			
4	0.000	0.000	4	0.000	0.000			
5	0.000	0.000	5	0.000	0.000			
6	0.000	0.000	6	0.000	0.000			
	F	Panel B:	$\theta = -0.06$					
(NI C		Bi	as (N T)	(190	ccc)			
$(1\mathbf{v}, 1$	(07, 0) = (07, 0)	010)	(N, I)	=(130, 1)	000)			
1	0.008	 0.006	1	0.014	2 0.010			
2	0.005	0.000	2	0.014	0.010			
2	0.000	0.000	3	0.004	0.001			
4	0.002	0.000		0.000	0.000			
5	0.000	0.001	5	0.000	0.000			
6	0.000	0.000	6	0.000	0.000			
0	0.000	0.000		0.000	0.000			
		RM	ISE					
(N, T)	T) = (57, 3)	B16)	(N,T)	=(130,	666)			
, ,	1	2	. ,	1	2			
1	0.009	0.008	1	0.014	0.012			
2	0.007	0.005	2	0.007	0.004			
3	0.004	0.002	3	0.001	0.000			
4	0.002	0.001	4	0.000	0.000			
5	0.001	0.000	5	0.000	0.000			
C	0.001	0.001	6	0.000	0.000			

The table contains bias and RMSE for $\widehat{\theta}.$

 Table B.2:
 Number of estimated common factors across regimes

0							
Mean							
	(N,T) = (57,3)	516)	(N,T) = (130,666)				
$P_{\mathcal{S}} \setminus P_{\mathcal{D}}$	1	2	$P_{\mathcal{S}} \setminus P_{\mathcal{D}}$	1	2		
1	0.996, 0.999	1.000, 1.599	1	0.998, 0.996	0.996, 1.984		
2	1.997, 0.999	1.992, 1.996	2	1.980, 0.996	1.990, 1.984		
3	3.000, 0.998	2.985, 1.951	3	2.976, 1.000	2.950, 1.980		
4	3.996.0.999	3.996, 1.980	4	3.966, 0.992	3.958, 1.980		
5	4.978, 0.999	4.985, 1.992	5	4.946, 0.994	4.924, 1.984		
6	5.975, 1.000	5.929, 1.883	6	5.924, 0.994	5.912, 1.980		
		Missed true	e value (%)	1			
	(N,T) = (57,3)	516)		(N,T) = (130, 0)	666)		
$P_{\mathcal{S}} \setminus P_{\mathcal{D}}$	1	2	$P_{\mathcal{S}} \setminus P_{\mathcal{D}}$	1	2		
1	0.400, 0.100	0.000, 40.00	1	0.200, 0.800	0.400, 1.000		
2	0.200, 0.100	0.200, 0.700	2	1.400, 0.400	0.600, 1.000		
3	0.000, 0.200	0.700, 4.800	3	1.000, 0.000	2.000, 1.400		
4	0.200, 0.100	0.200, 1.800	4	1.200, 0.800	2.000, 1.600		
5	0.700, 0.100	0.500, 0.700	5	2.400, 0.600	2.000, 1.000		
6	1.000, 0.000	4.400, 11.50	6	2.000, 0.600	2.200, 1.200		

The table contains the average numbers of estimated factors and the percentage of times the estimator is wrong.

C Technical Lemmas

We begin with a Borel-Cantelli type result, whose proof can be found in Barigozzi and Trapani (2021).

Lemma C.1. Consider a multi-index partial sums process U_{i_1,\ldots,i_h} , with $1 \le i_1 \le S_1$, $1 \le i_2 \le S_2$, etc... Assume that

$$\sum_{S_1} \cdots \sum_{S_h} \frac{1}{S_1 \cdot \ldots \cdot S_h} P\left(\max_{1 \le i_1 \le S_1, \ldots, 1 \le i_h \le S_h} |U_{i_1, \ldots, i_h}| > \epsilon L_{S_1, \ldots, S_h}\right) < \infty, \tag{C.2}$$

for some $\epsilon > 0$ and a sequence L_{S_1,\dots,S_h} defined as

$$L_{S_{1},...,S_{h}} = S_{1}^{d_{1}} \cdot ... \cdot S_{h}^{d_{h}} l_{1} (S_{1}) \cdot ... l_{h} (S_{h}),$$

where d_1 , d_2 , etc. are non-negative numbers and $l_1(\cdot)$, $l_2(\cdot)$, etc. are slowly varying functions in the sense of Karamata. Then it holds that

$$\lim \sup_{(S_1,\dots,S_h)\to\infty} \frac{|U_{S_1,\dots,S_h}|}{L_{S_1,\dots,S_h}} = 0 \ a.s.$$
(C.3)

Henceforth, we will extensively use the following notation:

$$\overline{v}_{N,T}\left(\epsilon\right) = \left(\ln N \ln T\right)^{2\frac{3+r+\epsilon}{r}},\tag{C.4}$$

where $\epsilon > 0$ and r is defined in Assumption 3(i). Recall (11)

$$S\left(\mathbf{B}^{P},\mathbf{U}^{P},\theta\right) = \frac{1}{NT}\sum_{i=1}^{N}\sum_{t=1}^{T}\left[\widetilde{R}_{i,t} - \left(\beta_{\mathcal{D},i}^{P_{\mathcal{D}}'}\mathbf{u}_{\mathcal{D},t}^{P_{\mathcal{D}}}d_{\mathcal{D},t}\left(\theta\right) + \beta_{\mathcal{S},i}^{P_{\mathcal{S}}'}\mathbf{u}_{\mathcal{S},t}^{P_{\mathcal{S}}}d_{\mathcal{S},t}\left(\theta\right)\right)\right]^{2},$$

which is the least squares loss function calculated at the triplet $(\mathbf{B}^{P}, \mathbf{U}^{P}, \theta)$; similarly, we define

$$S_{\mathbf{UB}}\left(\theta\right) = \frac{1}{NT} \sum_{i=1}^{N} \sum_{t=1}^{T} \left[\widetilde{R}_{i,t} - \left(\beta_{\mathcal{D},i}^{P_{\mathcal{D}}^{0}} \mathbf{u}_{\mathcal{D},t}^{P_{\mathcal{D}}^{0}} d_{\mathcal{D},t}\left(\theta\right) + \beta_{\mathcal{S},i}^{P_{\mathcal{S}}^{0}} \mathbf{u}_{\mathcal{S},t}^{P_{\mathcal{S}}^{0}} d_{\mathcal{S},t}\left(\theta\right) \right) \right]^{2},$$
(C.5)

which denotes the concentrated version of $S(\mathbf{B}^{P}, \mathbf{U}^{P}, \theta)$ when using the correct number of factors $P_{\mathcal{D}}^{0}$ and $P_{\mathcal{S}}^{0}$. Finally, when using $P_{\mathcal{D}}$ and $P_{\mathcal{S}}$ factors, with $P_{\mathcal{D}}^{0} \leq P_{\mathcal{D}} \leq P_{\max}$ and $P_{\mathcal{S}}^{0} \leq P_{\mathcal{S}} \leq P_{\max}$, we write

$$S_{\mathbf{uB}}^{(P)}(\theta) = \frac{1}{NT} \sum_{i=1}^{N} \sum_{t=1}^{T} \left[\widetilde{R}_{i,t} - \left(\widehat{\beta}_{\mathcal{D},i}^{P_{\mathcal{D}'}} \widehat{\mathbf{u}}_{\mathcal{D},t}^{P_{\mathcal{D}}} d_{\mathcal{D},t} \left(\theta \right) + \widehat{\beta}_{\mathcal{S},i}^{P_{\mathcal{S}'}} \widehat{\mathbf{u}}_{\mathcal{S},t}^{P_{\mathcal{S}}} d_{\mathcal{S},t} \left(\theta \right) \right) \right]^2,$$
(C.6)

to define the concentrated version of $S(\mathbf{B}^{P}, \mathbf{U}^{P}, \theta)$ in that case. We will also use the $P_m \times P_j$ projection matrices

$$\widehat{H}_{jm}^{P_{j,m}}\left(\theta\right) = \frac{\mathbf{U}_{m,m}^{0}\left(\theta^{0}\right)\mathbf{U}_{m,j}^{0}\left(\theta\right)'}{T} \frac{\mathbf{B}_{m}^{0'}\widehat{\mathbf{B}}_{j}^{P_{j}}\left(\theta\right)}{N} \left(\widehat{V}_{j}^{P_{j}}\left(\theta\right)\right)^{-1},\tag{C.7}$$

where $j, m = \mathcal{D}, \mathcal{S}, \mathbf{U}_{m,j}^{0}(\theta)$ is defined as the $P_m \times T$ matrix of regime-specific factors, $\mathbf{U}_m(\theta)$, multiplied by a $T \times T$ diagonal matrix whose entries are $d_{j,t}(\theta), 1 \leq t \leq T$; finally, $\hat{V}_j^{P_j}(\theta)$ is a $P_j \times P_j$ matrix containing the largest P_j eigenvalues of

$$\frac{1}{NT}\sum_{t=1}^{T}\widetilde{\mathbf{R}}_{t}\widetilde{\mathbf{R}}_{t}^{\prime}d_{j,t}\left(\theta\right).$$

Note that, for $j \neq m$, $\widehat{H}_{jm}^{P_{jm}}(\theta^0)$ reduces to a matrix of zeros. In order to make the notation lighter, the superscript $P_{j,m}$ is set equal to P_j when j = m.

We now present a few lemmas to show the strong consistency of $\hat{\theta}$.

Lemma C.2. We assume that Assumptions 1-4 are satisfied. Then it holds that

$$\frac{1}{N}\sum_{i=1}^{N}\left\|\widehat{\beta}_{j,i}^{P_{j}}\left(\theta^{0}\right)-\widehat{H}_{jj}^{P_{j}}\left(\theta^{0}\right)'\beta_{j,i}^{0}\right\|^{2}=o_{a.s.}\left(P_{j}C_{N,T}^{-2}\overline{v}_{N,T}\left(\epsilon\right)\right),$$

for every $\epsilon > 0$ and $j = \mathcal{D}, \mathcal{S}$.

Proof. This follows by adapting the passages in the proof of Theorem 3.2 in Massacci (2017).

Then we have

$$\frac{1}{N} \sum_{i=1}^{N} \left\| \widehat{\beta}_{j,i}^{P_{j}}\left(\theta^{0}\right) - \widehat{H}_{jj}^{P_{j}}\left(\theta^{0}\right)' \beta_{j,i}^{0} \right\|^{2} \tag{C.8}$$

$$\leq \left[\frac{4}{N^{2}} \sum_{i=1}^{N} \sum_{l=1}^{N} \sigma_{jil}^{2} \left(\theta^{0}\right) + \frac{1}{N} \left(\frac{1}{N^{2}} \sum_{l=1}^{N} \sum_{q=1}^{N} \left| \sum_{i=1}^{N} \varkappa_{jil} \left(\theta^{0}\right) \varkappa_{jiq} \left(\theta^{0}\right) \right|^{2} \right)^{1/2} + \frac{2}{N} \sum_{i=1}^{N} \left(\left\| \beta_{\mathcal{D},i}^{0} \right\|^{2} + \left\| \beta_{\mathcal{S},i}^{0} \right\|^{2} \right) \frac{1}{NT^{2}} \sum_{l=1}^{N} \left\| \sum_{t=1}^{T} d_{j,t} \left(\theta^{0}\right) \mathbf{u}_{j,t}^{0} e_{l,t} \right\|^{2} \right] \\
\times \left(\frac{1}{N} \sum_{i=1}^{N} \left\| \widehat{\beta}_{j,i}^{P_{j}} \left(\theta^{0}\right) \right\|^{2} \right) \\
= A_{N,T} \left(\frac{1}{N} \sum_{i=1}^{N} \left\| \widehat{\beta}_{j,i}^{P_{j}} \left(\theta^{0}\right) \right\|^{2} \right),$$

having defined

$$\sigma_{jil}\left(\theta^{0}\right) = \frac{1}{T}\sum_{t=1}^{T} E\left(d_{j,t}\left(\theta^{0}\right)e_{i,t}e_{l,t}\right),$$

$$\varkappa_{jil}\left(\theta^{0}\right) = \frac{1}{T}\sum_{t=1}^{T} d_{j,t}\left(\theta^{0}\right)e_{i,t}e_{l,t} - \sigma_{jil}\left(\theta^{0}\right).$$

By the proof of Theorem 3.2 in Massacci (2017), it follows immediately that $E|A_{N,T}| \leq c_0 C_{N,T}^{-2}$. Thus, using the maximal inequality for multi-parameter processes in Corollary 4 of it follows that

$$E \max_{1 \le i \le N, 1 \le t \le T} |A_{i,t}| \le c_0 C_{N,T}^{-2} \ln N \ln T.$$

Lemma C.1 and Markov inequality immediately yield $A_{N,T} = o_{a.s.} \left(C_{N,T}^{-2} (\ln N)^{2+\epsilon} (\ln T)^{2+\epsilon} \right)$ for every $\epsilon > 0$. Also,

$$\frac{1}{N}\sum_{i=1}^{N}\left\|\widehat{\beta}_{j,i}^{P_{j}}\left(\theta^{0}\right)\right\|^{2}=P_{j}$$

by construction, since $\hat{\mathbf{B}}_{j}^{P_{j}}(\theta^{0})' \hat{\mathbf{B}}_{j}^{P_{j}}(\theta^{0}) = NI_{P_{j}}$. The desired result follows by putting everything together.

Lemma C.3. We assume that Assumptions 1-5 are satisfied. Then it holds that

$$\left|\widehat{\theta} - \theta^{0}\right| = o_{a.s.}\left(1\right),$$

for all $P_j^0 \leq P_j \leq P_{\max}, j = \mathcal{D}, \mathcal{S}$.

Proof. The desired result follows upon showing

$$S_{\mathbf{UB}}^{(P)}(\theta) - S_{\mathbf{UB}}^{(P_0)}(\theta^0) = c_0 \left| \theta - \theta^0 \right| + o_{a.s.}(1), \qquad (C.9)$$

with $c_0 > 0$, for every $\theta \neq \theta^0$ - note that $S_{\mathbf{UB}}^{(P)}(\theta)$ is defined in (C.6). We being by considering the identity

$$\begin{split} S_{\mathbf{UB}}^{(P)}\left(\theta\right) &- S_{\mathbf{UB}}^{(P)}\left(\theta^{0}\right) \\ = & S_{\mathbf{UB}}^{(P)}\left(\theta\right) - S_{\mathbf{B}}^{(P)}\left(\mathbf{B}_{\mathcal{D}}^{0}\widehat{H}_{\mathcal{DD}}^{P_{\mathcal{D}}}\left(\theta\right) + \mathbf{B}_{\mathcal{S}}^{0}\widehat{H}_{\mathcal{SD}}^{P_{\mathcal{S},\mathcal{D}}}\left(\theta\right), \mathbf{B}_{\mathcal{D}}^{0}\widehat{H}_{\mathcal{DS}}^{P_{\mathcal{D},\mathcal{S}}}\left(\theta\right) + \mathbf{B}_{\mathcal{S}}^{0}\widehat{H}_{\mathcal{SS}}^{P_{\mathcal{S}}}\left(\theta\right), \theta^{0}\right) \\ &+ S_{\mathbf{B}}^{(P)}\left(\mathbf{B}_{\mathcal{D}}^{0}\widehat{H}_{\mathcal{DD}}^{P_{\mathcal{D}}}\left(\theta\right) + \mathbf{B}_{\mathcal{S}}^{0}\widehat{H}_{\mathcal{SD}}^{P_{\mathcal{S},\mathcal{D}}}\left(\theta\right), \mathbf{B}_{\mathcal{D}}^{0}\widehat{H}_{\mathcal{DS}}^{P_{\mathcal{D},\mathcal{S}}}\left(\theta\right) + \mathbf{B}_{\mathcal{S}}^{0}\widehat{H}_{\mathcal{SS}}^{P_{\mathcal{S}}}\left(\theta\right), \theta^{0}\right) \\ &- S_{\mathbf{B}}^{(P)}\left(\mathbf{B}_{\mathcal{D}}^{0}\widehat{H}_{\mathcal{DD}}^{P_{\mathcal{D}}}\left(\theta\right), \mathbf{B}_{\mathcal{S}}^{0}\widehat{H}_{\mathcal{SS}}^{P_{\mathcal{S}}}\left(\theta\right), \theta^{0}\right) + S_{\mathbf{B}}^{(P)}\left(\mathbf{B}_{\mathcal{D}}^{0}\widehat{H}_{\mathcal{DD}}^{P_{\mathcal{D}}}\left(\theta\right), \mathbf{B}_{\mathcal{S}}^{0}\widehat{H}_{\mathcal{SS}}^{P_{\mathcal{S}}}\left(\theta\right), \theta^{0}\right) - S_{\mathbf{UB}}^{(P)}\left(\theta^{0}\right), \end{split}$$

where $S_{\mathbf{B}}^{(P)}$ denotes the loss function $S\left(\mathbf{B}^{P}, \mathbf{U}^{P}, \theta\right)$ concentrated at \mathbf{B}^{P} . By construction, it holds that

$$S_{\mathbf{B}}^{(P)}\left(\mathbf{B}_{\mathcal{D}}^{0}\widehat{H}_{\mathcal{DD}}^{P_{\mathcal{D}}}\left(\theta\right),\mathbf{B}_{\mathcal{S}}^{0}\widehat{H}_{\mathcal{SS}}^{P_{\mathcal{S}}}\left(\theta\right),\theta^{0}\right)=S_{\mathbf{B}}^{(P)}\left(\mathbf{B}_{\mathcal{D}}^{0},\mathbf{B}_{\mathcal{S}}^{0},\theta^{0}\right).$$

Following the proof of Lemma A.2 and Theorem 3.3 in Massacci (2017), it can be shown by standard arguments that

$$E \left| S_{\mathbf{B}}^{(P)} \left(\mathbf{B}_{\mathcal{D}}^{0}, \mathbf{B}_{\mathcal{S}}^{0}, \theta^{0} \right) - S_{\mathbf{UB}}^{(P)} \left(\theta^{0} \right) \right|^{2} \leq c_{0} C_{N,T}^{-2},$$

$$E \left| S_{\mathbf{UB}}^{(P)} \left(\theta \right) - S_{\mathbf{B}}^{(P)} \left(\mathbf{B}_{\mathcal{D}}^{0} \widehat{H}_{\mathcal{DD}}^{P_{\mathcal{D}}} \left(\theta \right) + \mathbf{B}_{\mathcal{S}}^{0} \widehat{H}_{\mathcal{SD}}^{P_{\mathcal{S},\mathcal{D}}} \left(\theta \right), \mathbf{B}_{\mathcal{D}}^{0} \widehat{H}_{\mathcal{DS}}^{P_{\mathcal{S},\mathcal{S}}} \left(\theta \right) + \mathbf{B}_{\mathcal{S}}^{0} \widehat{H}_{\mathcal{SS}}^{P_{\mathcal{S}}} \left(\theta \right), \theta^{0} \right) \right|^{2} \leq c_{0} C_{N,T}^{-2};$$

then, using again Corollary 4 in Moricz (1983), it follows that

$$E \max_{1 \le t \le T, 1 \le i \le N} \left| S_{\mathbf{B}, it}^{(P)} \left(\mathbf{B}_{\mathcal{D}}^{0}, \mathbf{B}_{\mathcal{S}}^{0}, \theta^{0} \right) - S_{\mathbf{UB}, it}^{(P)} \left(\theta^{0} \right) \right|^{2} \le c_{0} C_{N, T}^{-2} \left(\ln N \right) \left(\ln T \right),$$

and

$$E \max_{1 \le t \le T, 1 \le i \le N} \left| S_{\mathbf{UB}, it}^{(P)}\left(\theta\right) - S_{\mathbf{B}, it}^{(P)}\left(\mathbf{B}_{\mathcal{D}}^{0}\widehat{H}_{\mathcal{DD}}^{P_{\mathcal{D}}}\left(\theta\right) + \mathbf{B}_{\mathcal{S}}^{0}\widehat{H}_{\mathcal{SD}}^{P_{\mathcal{S}}, \mathcal{D}}\left(\theta\right), \mathbf{B}_{1}^{0}\widehat{H}_{\mathcal{DS}}^{P_{\mathcal{D}}, \mathcal{S}}\left(\theta\right) + \mathbf{B}_{\mathcal{S}}^{0}\widehat{H}_{\mathcal{SS}}^{P_{\mathcal{S}}}\left(\theta\right), \theta^{0} \right) \right|^{2} \le c_{0}C_{N,T}^{-2}\left(\ln N\right)\left(\ln T\right),$$

where

$$S_{\mathbf{UB},it}^{(P)}\left(\theta\right) = \frac{1}{NT} \sum_{i'=1}^{i} \sum_{t'=1}^{t} \left[\widetilde{R}_{i,t} - \left(\widehat{\beta}_{\mathcal{D},i}' \widehat{\mathbf{u}}_{\mathcal{D},t} d_{\mathcal{D},t}\left(\theta\right) + \widehat{\beta}_{\mathcal{S},i}' \widehat{\mathbf{u}}_{\mathcal{S},t} d_{\mathcal{S},t}\left(\theta\right) \right) \right]^{2}$$

and $S_{{\bf B},it}$ defined similarly. Hence, Lemma C.1 yields

$$S_{\mathbf{UB}}^{(P)}(\theta) - S_{\mathbf{B}}^{(P)} \left(\mathbf{B}_{\mathcal{D}}^{0} \widehat{H}_{\mathcal{DD}}^{P_{\mathcal{D}}}(\theta) + \mathbf{B}_{\mathcal{S}}^{0} \widehat{H}_{\mathcal{SD}}^{P_{\mathcal{S},\mathcal{D}}}(\theta) , \mathbf{B}_{\mathcal{D}}^{0} \widehat{H}_{\mathcal{DS}}^{P_{\mathcal{D},\mathcal{S}}}(\theta) + \mathbf{B}_{\mathcal{S}}^{0} \widehat{H}_{\mathcal{SS}}^{P_{\mathcal{S}}}(\theta) , \theta^{0} \right)$$

= $o_{a.s.} \left(C_{N,T}^{-1} \left(\ln N \right)^{1+\epsilon} \left(\ln T \right)^{1+\epsilon} \right).$

Similarly, by using Lemma A.3 in Massacci (2017) and the same arguments as above, it follows that there exists a $c_0 > 0$ such that

$$S_{\mathbf{B}}^{(P)} \left(\mathbf{B}_{\mathcal{D}}^{0} \widehat{H}_{\mathcal{DD}}^{P_{\mathcal{D}}} \left(\theta \right) + \mathbf{B}_{\mathcal{S}}^{0} \widehat{H}_{\mathcal{SD}}^{P_{2,\mathcal{D}}} \left(\theta \right), \mathbf{B}_{\mathcal{D}}^{0} \widehat{H}_{\mathcal{DS}}^{P_{\mathcal{D},\mathcal{S}}} \left(\theta \right) + \mathbf{B}_{\mathcal{S}}^{0} \widehat{H}_{\mathcal{SS}}^{P_{\mathcal{S}}} \left(\theta \right), \theta^{0} \right) - S_{\mathbf{B}}^{(P)} \left(\mathbf{B}_{\mathcal{D}}^{0}, \mathbf{B}_{\mathcal{S}}^{0}, \theta^{0} \right) \\ = c_{0} \left| \theta - \theta^{0} \right| + o_{a.s.} \left(1 \right).$$

Putting everything together, we have

$$S_{\mathbf{UB}}^{(P)}\left(\theta\right) - S_{\mathbf{UB}}^{(P)}\left(\theta^{0}\right) = c_{0} + o_{a.s.}\left(1\right).$$

Then (C.9) follows if we show that

$$S_{\mathbf{UB}}^{(P)}\left(\theta^{0}\right) - S_{\mathbf{UB}}^{(P_{0})}\left(\theta^{0}\right) = o_{a.s.}\left(1\right).$$
(C.10)

It holds that

$$\left|S_{\mathbf{UB}}^{(P)}\left(\theta^{0}\right) - S_{\mathbf{UB}}^{(P_{0})}\left(\theta^{0}\right)\right| \leq \left|S_{\mathbf{UB}}^{1}\left(\theta^{0}\right)\right| + \left|S_{\mathbf{UB}}^{2}\left(\theta^{0}\right)\right|,$$

where

$$S_{\mathbf{UB}}^{1}\left(\theta^{0}\right) = \frac{2}{NT} \sum_{t=1}^{T} \left(d_{\mathcal{D},t}\left(\theta^{0}\right) \mathbf{u}_{\mathcal{D},t}^{0\prime} \widehat{H}_{\mathcal{DD}}^{P_{\mathcal{D}}+}\left(\theta^{0}\right)' \left(\widehat{\mathbf{B}}_{\mathcal{D}}^{P_{\mathcal{D}}}\left(\theta^{0}\right) - \mathbf{B}_{\mathcal{D}}^{0} \widehat{H}_{\mathcal{DD}}^{P_{\mathcal{D}}}\left(\theta^{0}\right) \right)' + d_{\mathcal{S},t}\left(\theta^{0}\right) \mathbf{u}_{\mathcal{S},t}^{0\prime} \widehat{H}_{\mathcal{SS}}^{P_{\mathcal{S}}+}\left(\theta^{0}\right)' \left(\widehat{\mathbf{B}}_{\mathcal{S}}^{P_{\mathcal{S}}}\left(\theta^{0}\right) - \mathbf{B}_{\mathcal{S}}^{0} \widehat{H}_{\mathcal{SS}}^{P_{\mathcal{S}}}\left(\theta^{0}\right) \right)' \right) \mathbf{e}_{t},$$

$$S_{\mathbf{UB}}^{2}\left(\theta^{0}\right) = \frac{1}{NT} \sum_{t=1}^{T} \left(d_{\mathcal{D},t}\left(\theta^{0}\right) \mathbf{u}_{\mathcal{D},t}^{0\prime} \widehat{H}_{\mathcal{DD}}^{P_{\mathcal{D}}+}\left(\theta^{0}\right)' \left(\widehat{\mathbf{B}}_{\mathcal{D}}^{P_{\mathcal{D}}}\left(\theta^{0}\right) - \mathbf{B}_{\mathcal{D}}^{0} \widehat{H}_{\mathcal{DD}}^{P_{\mathcal{D}}}\left(\theta^{0}\right) \right)' \right. \\ \left. \times \left(\widehat{\mathbf{B}}_{\mathcal{D}}^{P_{\mathcal{D}}}\left(\theta^{0}\right) - \mathbf{B}_{\mathcal{D}}^{0} \widehat{H}_{\mathcal{DD}}^{P_{\mathcal{D}}}\left(\theta^{0}\right)\right) \widehat{H}_{\mathcal{DD}}^{P_{\mathcal{D}}+}\left(\theta^{0}\right) \mathbf{u}_{\mathcal{D},t}^{0} \right. \\ \left. + d_{\mathcal{S},t}\left(\theta^{0}\right) \mathbf{u}_{\mathcal{S},t}^{0\prime} \widehat{H}_{\mathcal{SS}}^{P_{\mathcal{S}}+}\left(\theta^{0}\right)' \left(\widehat{\mathbf{B}}_{\mathcal{S}}^{P_{\mathcal{S}}}\left(\theta^{0}\right) - \mathbf{B}_{\mathcal{S}}^{0} \widehat{H}_{\mathcal{SS}}^{P_{\mathcal{S}}}\left(\theta^{0}\right)\right)' \right. \\ \left. + \left(\widehat{\mathbf{B}}_{\mathcal{S}}^{P_{\mathcal{S}}}\left(\theta^{0}\right) - \mathbf{B}_{\mathcal{S}}^{0} \widehat{H}_{\mathcal{SS}}^{P_{\mathcal{S}}}\left(\theta^{0}\right)\right) \widehat{H}_{\mathcal{SS}}^{P_{\mathcal{S}}+}\left(\theta^{0}\right) \mathbf{u}_{2,t}^{0}\right). \right.$$

We now estimate the order of magnitude of $S_{\mathbf{UB}}^{1}(\theta^{0})$ and $S_{\mathbf{UB}}^{2}(\theta^{0})$. We have

$$\begin{split} \left| S_{\mathbf{UB}}^{1} \left(\theta^{0} \right) \right| &\leq 2T^{-1/2} P_{\mathcal{D}}^{0} \left\| \widehat{H}_{\mathcal{D}\mathcal{D}}^{P_{\mathcal{D}}+} \left(\theta^{0} \right) \right\| \left(\frac{1}{N} \sum_{i=1}^{N} \left\| \widehat{\beta}_{\mathcal{D},i}^{P_{\mathcal{D}}} \left(\theta^{0} \right) - \widehat{H}_{\mathcal{D}\mathcal{D}}^{P_{\mathcal{D}}} \left(\theta^{0} \right)' \beta_{\mathcal{D},i}^{0} \right\|^{2} \right)^{1/2} \\ &\left(\frac{1}{N} \sum_{i=1}^{N} \left\| \frac{1}{\sqrt{T}} \sum_{t=1}^{T} d_{\mathcal{D},t} \left(\theta^{0} \right) \mathbf{u}_{\mathcal{D},t}^{0} e_{i,t} \right\|^{2} \right)^{1/2} \\ &+ 2T^{-1/2} P_{\mathcal{S}}^{0} \left\| \widehat{H}_{\mathcal{S}\mathcal{S}}^{P_{\mathcal{S}}+} \left(\theta^{0} \right) \right\| \left(\frac{1}{N} \sum_{i=1}^{N} \left\| \widehat{\beta}_{\mathcal{S},i}^{P_{\mathcal{S}}} \left(\theta^{0} \right) - \widehat{H}_{\mathcal{S}\mathcal{S}}^{P_{\mathcal{S}}} \left(\theta^{0} \right)' \beta_{\mathcal{S},i}^{0} \right\|^{2} \right)^{1/2} \\ &\left(\frac{1}{N} \sum_{i=1}^{N} \left\| \frac{1}{\sqrt{T}} \sum_{t=1}^{T} d_{\mathcal{S},t} \left(\theta^{0} \right) \mathbf{u}_{\mathcal{S},t}^{0} e_{i,t} \right\|^{2} \right)^{1/2}. \end{split}$$

By construction, note that

$$\left\|\widehat{H}_{jj}^{P_j+}\left(\theta^0\right)\right\| \le c_0 P_j^{1/2} \tag{C.11}$$

for $j = \mathcal{D}, \mathcal{S}$. Also, Assumption 4(v) entails that, by the same logic as in the previous passages

that

$$\frac{1}{N}\sum_{i=1}^{N}\left\|\frac{1}{\sqrt{T}}\sum_{t=1}^{T}d_{j,t}\left(\theta^{0}\right)\mathbf{u}_{j,t}^{0}e_{i,t}\right\|^{2}=o_{a.s.}\left(\overline{v}_{N,T}\left(\epsilon\right)\right),$$

for every $\epsilon > 0$ and $j = \mathcal{D}, \mathcal{S}$. Thus, using Lemma C.2, we finally have

$$\left|S_{\mathbf{UB}}^{1}\left(\theta^{0}\right)\right| = o_{a.s.}\left(\overline{P}C_{N,T}^{-1}T^{-1/2}\overline{v}_{N,T}\left(\epsilon\right)\right),$$

having set $\overline{P} = \max \{P_{\mathcal{D}}, P_{\mathcal{S}}\}$; this term is therefore $o_{a.s.}(1)$ by (20). Similarly, it holds that (see Massacci (2017))

$$\begin{split} \left| S_{\mathbf{UB}}^{2} \left(\theta^{0} \right) \right| &\leq \frac{1}{T} \sum_{t=1}^{T} \left\| \mathbf{u}_{\mathcal{D},t}^{0} \right\|^{2} \left\| \widehat{H}_{\mathcal{DD}}^{P_{\mathcal{D}}+} \left(\theta^{0} \right) \right\|^{2} \frac{1}{N} \sum_{i=1}^{N} \left\| \widehat{\beta}_{\mathcal{D},i}^{P_{\mathcal{D}}} \left(\theta^{0} \right) - \widehat{H}_{\mathcal{DD}}^{P_{\mathcal{D}}} \left(\theta^{0} \right)' \beta_{\mathcal{D},i}^{0} \right\|^{2} \\ &+ \frac{1}{T} \sum_{t=1}^{T} \left\| \mathbf{u}_{\mathcal{S},t}^{0} \right\|^{2} \left\| \widehat{H}_{\mathcal{SS}}^{P_{\mathcal{S}}+} \left(\theta^{0} \right) \right\|^{2} \frac{1}{N} \sum_{i=1}^{N} \left\| \widehat{\beta}_{\mathcal{S},i}^{P_{\mathcal{S}}} \left(\theta^{0} \right) - \widehat{H}_{\mathcal{SS}}^{P_{\mathcal{S}}} \left(\theta^{0} \right)' \beta_{\mathcal{S},i}^{0} \right\|^{2}. \end{split}$$

Using (C.11) and Lemma C.2, and noting that Assumptions 3(i) and 5(i) entail

$$\frac{1}{T} \sum_{t=1}^{T} \left\| \mathbf{u}_{j,t}^{0} \right\|^{2} = O_{a.s.} (1) \, ,$$

for $j = \mathcal{D}, \mathcal{S}$, we have

$$\left|S_{\mathbf{UB}}^{2}\left(\theta^{0}\right)\right|=o_{a.s.}\left(\overline{P}^{2}C_{N,T}^{-2}\overline{v}_{N,T}\left(\epsilon\right)\right),$$

which again is $o_{a.s.}(1)$ by (20). Thus, (C.10) follows. Putting all together, the desired result obtains.

In the next two lemmas, we define the set

$$B_{N,T} = \left(\frac{\overline{v}_{N,T}\left(\epsilon\right)}{T}, c_B\right),\tag{C.12}$$
where $0 < c_0 < \infty$ and $\overline{v}_{N,T}$ is defined in (C.4), and the functions

$$\omega^{0}(\eta,\theta) = \frac{1}{N^{\eta}T} \sum_{i=1}^{N} \sum_{t=1}^{T} \left| d_{\mathcal{S},t}(\theta) - d_{\mathcal{S},t}\left(\theta^{0}\right) \right| \left(\delta_{i}^{0'} \underline{\mathbf{u}}_{t}^{0}\right)^{2}, \qquad (C.13)$$

$$h^{0}(\eta,\theta) = \frac{1}{N^{\eta}T} \sum_{i=1}^{N} \sum_{t=1}^{T} d_{\mathcal{S},t}(\theta) \,\delta_{i}^{0'} \underline{\mathbf{u}}_{t}^{0} e_{i,t}, \qquad (C.14)$$

where $\underline{\mathbf{u}}_t^0$ and δ_i^0 are defined in (A.1) and (A.2) respectively.

Lemma C.4. We assume that Assumptions 3 and 5 are satisfied. Then there exist two random variables N_0 and T_0 , and a positive, finite constant c_0 , such that for $N \ge N_0$ and $T \ge T_0$, it holds that

$$\inf_{|\theta-\theta^0|\in B_{N,T}}\frac{\omega^0\left(\eta,\theta\right)}{|\theta-\theta^0|}\geq c_0$$

Proof. When possible, we let $\eta = 1$ to avoid a burdensome notation, and we consider the case $\theta - \theta^0 > 0$ only (the opposite case can be shown by symmetry). Define

$$\overline{\omega}_{N,T}^{0} = \omega^{0}\left(\eta,\theta\right) - E\omega^{0}\left(\eta,\theta\right),$$

and note, to start with, that, by Assumption 5(iii), there exist two positive, finite constants c_0 and c_1 such that

$$c_0 \left| \theta - \theta^0 \right| \le E \omega^0 \left(\eta, \theta \right) \le c_1 \left| \theta - \theta^0 \right|,$$

so that

$$\inf_{|\theta-\theta^0|\in B_{N,T}} \frac{E\omega^0\left(\eta,\theta\right)}{|\theta-\theta^0|} \ge c_0.$$
(C.15)

We now show that

$$\sup_{|\theta-\theta^0|\in B_{N,T}} \frac{\left|\overline{\omega}_{N,T}^0\right|}{|\theta-\theta^0|} = o_{a.s.} (1).$$
(C.16)

Consider the construction $\theta_j = \theta^0 + c^j \overline{v}_{N,T}(\epsilon) / T$, where $1 < c < \infty$, j = 0, ..., Q and Q is defined such that $c^{Q-1} \overline{v}_{N,T}(\epsilon) / T \le c_B$ and $c^Q \overline{v}_{N,T}(\epsilon) / T > c_B$ where c_B is defined in (C.12), and let

$$\overline{\omega}_{\widetilde{i},\widetilde{t}}^{0}(\theta_{j}) = \sum_{i=1}^{\widetilde{i}} \sum_{t=1}^{\widetilde{t}} \left(\left| d_{\mathcal{S},t}\left(\theta_{j}\right) - d_{\mathcal{S},t}\left(\theta^{0}\right) \right| \left(\delta_{i}^{0\prime} \underline{\mathbf{u}}_{t}^{0} \right)^{2} - E\left(\left| d_{\mathcal{S},t}\left(\theta_{j}\right) - d_{\mathcal{S},t}\left(\theta^{0}\right) \right| \left(\delta_{i}^{0\prime} \underline{\mathbf{u}}_{t}^{0} \right)^{2} \right) \right).$$

Then, standard calculations based on the modulus of continuity yield

$$\sup_{\substack{|\theta-\theta^{0}|\in B_{N,T}|\\ \theta-\theta^{0}|\in B_{N,T}}} \frac{\overline{\omega}_{N,T}^{0}}{|\theta-\theta^{0}|} \tag{C.17}$$

$$\leq 2 \sup_{0\leq j\leq Q} \frac{\frac{1}{NT}\sum_{i=1}^{N}\sum_{t=1}^{T} \left(\delta_{i}^{0'}\underline{\mathbf{u}}_{t}^{0}\right)^{2} \left(|d_{\mathcal{S},t}\left(\theta_{j}\right) - d_{\mathcal{S},t}\left(\theta^{0}\right)| - E \left|d_{\mathcal{S},t}\left(\theta_{j}\right) - d_{\mathcal{S},t}\left(\theta^{0}\right)\right|\right)}{|\theta_{j} - \theta^{0}|} \\
+ 2 \sup_{0\leq j\leq Q} \sup_{\theta_{j}\leq \theta\leq \theta_{j+1}} \frac{\frac{\ln nT}{NT^{2}}\sum_{i=1}^{N}\sum_{t=1}^{T} \left(\delta_{i}^{0'}\underline{\mathbf{u}}_{t}^{0}\right)^{2} \left(|d_{\mathcal{S},t}\left(\theta\right) - d_{\mathcal{S},t}\left(\theta_{j}\right)| - E \left|d_{\mathcal{S},t}\left(\theta\right) - d_{\mathcal{S},t}\left(\theta_{j}\right)\right|\right)}{|\theta_{j} - \theta^{0}|} \\
= 2 \sup_{0\leq j\leq Q} \frac{\overline{\omega}^{0}\left(\theta_{j},\theta^{0}\right)}{|\theta_{j} - \theta^{0}|} + 2 \sup_{0\leq j\leq Q} \sup_{\theta_{j}\leq \theta\leq \theta_{j+1}} \frac{\overline{\omega}^{0}\left(\theta,\theta_{j}\right)}{|\theta_{j} - \theta^{0}|} = I + II.$$

We begin with I. Let $\delta_{N,T} = NT$; it holds that

$$\sum_{N=1}^{\infty} \sum_{T=1}^{\infty} \frac{1}{NT} P \left[\max_{1 \leq \widetilde{i} \leq N, 1 \leq \widetilde{t} \leq T} \sup_{0 \leq j \leq Q-1} \frac{\overline{\omega}_{\widetilde{i},\widetilde{t}}^{0}(\theta_{j})}{|\theta_{j} - \theta^{0}|} > \delta_{N,T} \right]$$

$$\leq \sum_{j=0}^{\infty} \sum_{N=1}^{\infty} \sum_{T=1}^{\infty} \frac{1}{NT} P \left[\max_{1 \leq \widetilde{i} \leq N, 1 \leq \widetilde{t} \leq T} \frac{\overline{\omega}_{\widetilde{i},\widetilde{t}}^{0}(\theta_{j})}{|\theta_{j} - \theta^{0}|} > \delta_{N,T} \right]$$

$$\leq \sum_{j=0}^{\infty} \sum_{N=1}^{\infty} \sum_{T=1}^{\infty} \frac{\delta_{N,T}^{-2}}{NT} E \left(\max_{1 \leq \widetilde{i} \leq N, 1 \leq \widetilde{t} \leq T} \left| \frac{\overline{\omega}_{\widetilde{i},\widetilde{t}}^{0}(\theta_{j})}{|\theta_{j} - \theta^{0}|} \right|^{2} \right).$$

Also, consider the scalar case for simplicity and note that

$$E\left(NT\overline{\omega}_{N,T}^{0}\left(\theta_{j}\right)\right)^{2}$$

$$= E\left[\sum_{i=1}^{N}\left(\delta_{i}^{0}\right)^{2}\sum_{t=1}^{T}\left(\left|d_{\mathcal{S},t}\left(\theta_{j}\right)-d_{\mathcal{S},t}\left(\theta^{0}\right)\right|\left(\underline{\mathbf{u}}_{t}^{0}\right)^{2}-E\left|d_{\mathcal{S},t}\left(\theta_{j}\right)-d_{\mathcal{S},t}\left(\theta^{0}\right)\right|\left(\underline{\mathbf{u}}_{t}^{0}\right)^{2}\right)\right]^{2}$$

$$\leq c_{0}\left(\sum_{i=1}^{N}\left(\delta_{i}^{0}\right)^{2}\right)^{2}E\left[\sum_{t=1}^{T}\left(\left|\left|d_{\mathcal{S},t}\left(\theta_{j}\right)-d_{\mathcal{S},t}\left(\theta^{0}\right)\right|\left(\underline{\mathbf{u}}_{t}^{0}\right)^{2}\right]-E\left[\left|d_{\mathcal{S},t}\left(\theta_{j}\right)-d_{\mathcal{S},t}\left(\theta^{0}\right)\right|\left(\underline{\mathbf{u}}_{t}^{0}\right)^{2}\right]\right)\right]^{2}$$

$$\leq c_{0}N^{2}E\left[\sum_{t=1}^{T}\left(\left|\left|d_{\mathcal{S},t}\left(\theta_{j}\right)-d_{\mathcal{S},t}\left(\theta^{0}\right)\right|\left(\underline{\mathbf{u}}_{t}^{0}\right)^{2}\right]-E\left[\left|d_{\mathcal{S},t}\left(\theta_{j}\right)-d_{\mathcal{S},t}\left(\theta^{0}\right)\right|\left(\underline{\mathbf{u}}_{t}^{0}\right)^{2}\right]\right)\right]^{2}.$$

By Assumption 5(*i*), $|d_{\mathcal{S},t}(\theta_j) - d_{\mathcal{S},t}(\theta^0)| (\underline{\mathbf{u}}_t^0)^2$ is a strictly stationary, ρ -mixing sequence with mixing numbers ρ_m such that $\sum_{m=1}^{\infty} \rho_m^{1/2} < \infty$ - we refer to the proof of Lemma C.11 for details

on how to show this. Thus, Lemma 1 in Peligrad (1987) entails

$$E\left[\sum_{t=1}^{T}\left(\left[\left|d_{\mathcal{S},t}\left(\theta_{j}\right)-d_{\mathcal{S},t}\left(\theta^{0}\right)\right|\left(\underline{\mathbf{u}}_{t}^{0}\right)^{2}\right]-E\left[\left|d_{\mathcal{S},t}\left(\theta_{j}\right)-d_{\mathcal{S},t}\left(\theta^{0}\right)\right|\left(\underline{\mathbf{u}}_{t}^{0}\right)^{2}\right]\right)\right]^{2}$$

$$\leq c_{0}TE\left[\left(\left[\left|d_{\mathcal{S},t}\left(\theta_{j}\right)-d_{\mathcal{S},t}\left(\theta^{0}\right)\right|\left(\underline{\mathbf{u}}_{t}^{0}\right)^{2}\right]-E\left[\left|d_{\mathcal{S},t}\left(\theta_{j}\right)-d_{\mathcal{S},t}\left(\theta^{0}\right)\right|\left(\underline{\mathbf{u}}_{t}^{0}\right)^{2}\right]\right)\right]^{2}$$

$$\leq c_{0}TE\left(\left|d_{\mathcal{S},t}\left(\theta_{j}\right)-d_{\mathcal{S},t}\left(\theta^{0}\right)\right|^{2}\left(\underline{\mathbf{u}}_{t}^{0}\right)^{4}\right).$$

Also, note that, by Assumption 5(ii)

$$E\left(\left|d_{\mathcal{S},t}\left(\theta_{j}\right)-d_{\mathcal{S},t}\left(\theta^{0}\right)\right|^{2}\left(\underline{\mathbf{u}}_{t}^{0}\right)^{4}\right)$$

= $E\left(\left|d_{\mathcal{S},t}\left(\theta_{j}\right)-d_{\mathcal{S},t}\left(\theta^{0}\right)\right|^{2}E\left[\left(\underline{\mathbf{u}}_{t}^{0}\right)^{4}|z_{t}\right]\right)\leq c_{0}\left|\theta_{j}-\theta^{0}\right|$

Thus,

$$E\left(NT\overline{\omega}_{N,T}^{0}\left(\theta_{j}\right)\right)^{2} \leq c_{0}\left|\theta_{j}-\theta^{0}\right|N^{2}T;$$

using again Corollary 4 in Moricz (1983), this entails that

$$E\left(\max_{1\leq \widetilde{i}\leq N, 1\leq \widetilde{t}\leq T} \left|\frac{\overline{\omega}_{\widetilde{i},\widetilde{t}}^{0}(\theta_{j})}{\theta_{j}-\theta^{0}}\right|^{2}\right)\leq c_{0}\left|\theta_{j}-\theta^{0}\right|^{-1}N^{2}T\left(\ln N\right)\left(\ln T\right).$$

Therefore

$$\sum_{N=1}^{\infty} \sum_{T=1}^{\infty} \frac{1}{NT} P\left(\max_{1 \le \widetilde{i} \le N, 1 \le \widetilde{t} \le T} \sup_{0 \le j \le Q-1} \frac{\overline{\omega}_{\widetilde{i},\widetilde{t}}^{0}(\theta_{j})}{|\theta_{j} - \theta^{0}|} > \delta_{N,T}\right)$$

$$\leq c_{0} \left(\sum_{j=0}^{\infty} c^{-j}\right) \sum_{N=1}^{\infty} \sum_{T=1}^{\infty} \frac{\delta_{N,T}^{-2}}{NT} (\overline{v}_{N,T}(\epsilon))^{-1} N^{2} T^{2} (\ln N) (\ln T) < \infty,$$
(C.18)

so that finally this entails

$$\sup_{0 \le j \le Q-1} \frac{\overline{\omega}_{NT}^{0}\left(\theta_{j}\right)}{\left|\theta_{j} - \theta^{0}\right|} = o_{a.s.}\left(1\right)$$

We now turn to *II*. We now turn to *II*; as in Hansen (2000), we consider the construction $\theta_{j+1} = \theta_j + \delta_j$, where $\delta_j = T^{-1}c^{j-1}(c-1)\overline{v}_{NT}$; letting $m_j = T\delta_j = c^{j-1}(c-1)\overline{v}_{NT}$, we also define

 $\theta_k = \theta_j + T^{-1} (k-1), \le 1 \le k \le m_j$. We rely on the inequality

$$\sup_{\theta_j \le \theta \le \theta_{j+1}} \frac{1}{NT} \sum_{i=1}^{\widetilde{i}} \sum_{t=1}^{\widetilde{t}} \left(\delta_i^{0'} \underline{\mathbf{u}}_t^0 \right)^2 \left(|d_{\mathcal{S},t} \left(\theta \right) - d_{\mathcal{S},t} \left(\theta_j \right)| - E \left| d_{\mathcal{S},t} \left(\theta \right) - d_{\mathcal{S},t} \left(\theta_j \right) \right| \right)$$
(C.19)

$$\leq \max_{2 \leq k \leq m_j+1} \frac{1}{NT} \left| \sum_{i=1}^{n'} \sum_{t=1}^{t'} \left(\delta_i^{0'} \underline{\mathbf{u}}_t^0 \right)^2 d_{2t} \left(\theta_k, \theta_j \right) \right| + \max_{1 \leq k \leq m_j} \frac{1}{NT} \left| \sum_{i=1}^{n'} \sum_{t=1}^{t'} \left(\delta_i^{0'} \underline{\mathbf{u}}_t^0 \right)^2 d_{2t} \left(\theta_{k+1}, \theta_k \right) \right| = a+b,$$

where

$$d_{2t} \left(\theta_{k}, \theta_{j}\right) = \left| d_{\mathcal{S},t} \left(\theta_{k}\right) - d_{\mathcal{S},t} \left(\theta_{j}\right) \right| - E \left| d_{\mathcal{S},t} \left(\theta_{k}\right) - d_{\mathcal{S},t} \left(\theta_{j}\right) \right|,$$

$$d_{2t} \left(\theta_{k+1}, \theta_{k}\right) = \left| d_{\mathcal{S},t} \left(\theta_{k+1}\right) - d_{\mathcal{S},t} \left(\theta_{k}\right) \right| - E \left| d_{\mathcal{S},t} \left(\theta_{k+1}\right) - d_{\mathcal{S},t} \left(\theta_{k}\right) \right|.$$

We begin by considering a

$$\begin{split} &\sum_{N=1}^{\infty} \sum_{T=1}^{\infty} \frac{1}{NT} P\left(\max_{1 \le n' \le N, 1 \le t' \le T} \sup_{0 \le j \le Q} \max_{2 \le k \le m_j + 1} \frac{\frac{1}{NT} \left| \sum_{i=1}^{n'} \sum_{t=1}^{t'} \left(\delta_i^{0'} \underline{\mathbf{u}}_t^0 \right)^2 d_{2t} \left(\theta_k, \theta_j \right) \right|}{|\theta_j - \theta^0|} > \varepsilon \right) \\ &\le \sum_{j=0}^{\infty} \sum_{N=1}^{\infty} \sum_{T=1}^{\infty} \frac{1}{NT} P\left(\max_{1 \le n' \le N, 1 \le t' \le T} \max_{2 \le k \le m_j + 1} \left| \sum_{i=1}^{n'} \sum_{t=1}^{t'} \left(\delta_i^{0'} \underline{\mathbf{u}}_t^0 \right)^2 d_{2t} \left(\theta_k, \theta_j \right) \right| > \varepsilon NT \left| \theta_j - \theta^0 \right| \right) \\ &\le \sum_{j=0}^{\infty} \sum_{k=2}^{m_j + 1} \sum_{N=1}^{\infty} \sum_{T=1}^{\infty} \frac{1}{NT} P\left(\max_{1 \le n' \le N, 1 \le t' \le T} \left| \sum_{i=1}^{n'} \sum_{t=1}^{t'} \left(\delta_i^{0'} \underline{\mathbf{u}}_t^0 \right)^2 d_{2t} \left(\theta_k, \theta_j \right) \right| > \varepsilon NT \left| \theta_j - \theta^0 \right| \right) \\ &\le \sum_{j=0}^{\infty} \sum_{k=2}^{m_j + 1} \sum_{N=1}^{\infty} \sum_{T=1}^{\infty} \frac{\varepsilon^{-(2+r)} \left(NT \right)^{-(2+r)} T^{2+r} c^{-(2+r)j}}{\overline{v}_{N,T}^{2+r} \left(\epsilon \right) NT} \\ &\times E \left(\max_{1 \le n' \le N, 1 \le t' \le T} \left| \sum_{i=1}^{n'} \sum_{t=1}^{t'} \left(\delta_i^{0'} \underline{\mathbf{u}}_t^0 \right)^2 d_{2t} \left(\theta_k, \theta_j \right) \right|^{2+r} \right) \right) \end{split}$$

for some r > 0 such that $E \|\underline{\mathbf{u}}_t^0\|^{4+2r} \leq c_0 T^{2+r}$. Note now that, as in the above, $(\delta_i^{0'} \underline{\mathbf{u}}_t^0)^2 d_{2t} (\theta_k, \theta_j)$ is a strictly stationary, ρ -mixing sequence with mixing numbers ρ_m such that $\sum_{m=1}^{\infty} \rho_m^{1/2} < \infty$;

thus, using Burkholder's inequality (see Theorem 1.1 in Shao, 1995), we obtain

$$E \left| \sum_{i=1}^{N} \sum_{t=1}^{T} \left(\delta_{i}^{0'} \underline{\mathbf{u}}_{t}^{0} \right)^{2} d_{2t} \left(\theta_{k}, \theta_{j} \right) \right|^{2+r}$$

$$\leq c_{0} N^{2+r} T^{1+r/2} \left(E \left\| \left\| \underline{\mathbf{u}}_{t}^{0} \right\|^{2} d_{2t} \left(\theta_{k}, \theta_{j} \right) \right\|^{2} \right)^{1+r/2} \leq c_{1} N^{2+r} T^{1+r/2} \left| E \left\| \underline{\mathbf{u}}_{t}^{0} \right\|^{4} d_{2t}^{2} \left(\theta_{k}, \theta_{j} \right) \right|^{1+r/2}$$

$$\leq c_{0} N^{2+r} T^{1+r/2} \left| \theta_{k} - \theta_{j} \right|^{1+r/2} \leq c_{1} N^{2+r} \overline{v}_{NT}^{1+r/2} \left(\epsilon \right) c^{j(1+r/2)}.$$

Applying Corollary 4 in Moricz (1983), it follows that

$$\begin{split} \sum_{N=1}^{\infty} \sum_{T=1}^{\infty} \frac{1}{NT} P \left(\max_{1 \le n' \le N, 1 \le t' \le T} \sup_{0 \le j \le Q} \max_{2 \le k \le m_j + 1} \frac{\frac{1}{NT} \left| \sum_{i=1}^{n'} \sum_{t=1}^{t'} \left(\delta_i^{0'} \underline{\mathbf{u}}_t^0 \right)^2 d_{2t} \left(\theta_k, \theta_j \right) \right|}{|\theta_j - \theta^0|} > \varepsilon \right) \\ &\le c_0 \sum_{j=0}^{\infty} \sum_{k=2}^{m_j+1} \sum_{N=1}^{\infty} \sum_{T=1}^{\infty} \frac{e^{-(2+r)} \left(NT \right)^{-(2+r)} T^{2+r} e^{-(2+r)j}}{\overline{v}_{N,T}^{2+r} \left(\epsilon \right) NT} N^{2+r} \overline{v}_{NT}^{1+r/2} \left(\epsilon \right) c^{j(1+r/2)} \left(\ln N \ln T \right)^{2+r} \\ &\le c_0 \sum_{j=0}^{\infty} \sum_{N=1}^{\infty} \sum_{T=1}^{\infty} \frac{e^{-(1+r/2)j}}{\overline{v}_{N,T}^{1+r/2} \left(\epsilon \right) NT} \left(\ln N \ln T \right)^{2+r} m_j \\ &\le c_0 \left(\sum_{j=0}^{\infty} e^{-rj/2} \right) \sum_{N=1}^{\infty} \sum_{T=1}^{\infty} \frac{\left(\ln N \ln T \right)^{2+r}}{\overline{v}_{NT}^{r/2} \left(\epsilon \right) NT} < \infty. \end{split}$$

This entails that $a = o_{a.s.}$ (1). Similarly, considering b

$$\begin{split} &\sum_{N=1}^{\infty}\sum_{T=1}^{\infty}\frac{1}{NT}P\left(\max_{1\leq n'\leq N, 1\leq t'\leq T}\sup_{0\leq j\leq Q}\max_{1\leq k\leq m_{j}}\frac{\frac{1}{NT}\left|\sum_{i=1}^{n'}\sum_{t=1}^{t'}\left(\delta_{i}^{0'}\underline{\mathbf{u}}_{t}^{0}\right)^{2}d_{2t}\left(\theta_{k+1},\theta_{k}\right)\right|}{\left|\theta_{j}-\theta^{0}\right|}>\varepsilon\right)\\ &\leq &c_{0}\sum_{j=0}^{\infty}\sum_{k=2}^{m_{j}+1}\sum_{N=1}^{\infty}\sum_{T=1}^{\infty}\frac{1}{NT}P\left(\max_{1\leq n'\leq N, 1\leq t'\leq T}\left|\sum_{i=1}^{n'}\sum_{t=1}^{t'}\left(\delta_{i}^{0'}\underline{\mathbf{u}}_{t}^{0}\right)^{2}d_{2t}\left(\theta_{k+1},\theta_{k}\right)\right|>NT\left|\theta_{j}-\theta^{0}\right|\varepsilon\right)\\ &\leq &c_{0}\sum_{j=0}^{\infty}\sum_{k=2}^{m_{j}+1}\sum_{N=1}^{\infty}\sum_{T=1}^{\infty}c^{-2j}\frac{\varepsilon^{-2}\left(NT\right)^{-2}T^{2}}{\overline{v}_{N,T}^{2}\left(\epsilon\right)NT}E\max_{1\leq n'\leq N, 1\leq t'\leq T}\left|\sum_{i=1}^{n'}\sum_{t=1}^{t'}\left(\delta_{i}^{0'}\underline{\mathbf{u}}_{t}^{0}\right)^{2}d_{2t}\left(\theta_{k+1},\theta_{k}\right)\right|^{2}.\end{split}$$

Also, using again Theorem 1.1 in Shao (1995)

$$\begin{split} E \left| \sum_{i=1}^{N} \sum_{t=1}^{T} \left(\delta_{i}^{0'} \underline{\mathbf{u}}_{t}^{0} \right)^{2} d_{2t} \left(\theta_{k+1}, \theta_{k} \right) \right|^{2} \\ \leq c_{0} N^{2} T E \left\| \underline{\mathbf{u}}_{t}^{0} \right\|^{4} d_{2t}^{2} \left(\theta_{k+1}, \theta_{k} \right) \leq c_{1} N^{2} T \left| \theta_{k+1} - \theta_{k} \right| \\ \leq c_{1} N^{2} T \frac{\delta_{j}}{m_{j}} = c_{1} N^{2}, \end{split}$$

so that Corollary 4 in Moricz (1983) yields

$$\sum_{N=1}^{\infty} \sum_{T=1}^{\infty} \frac{1}{NT} P\left(\max_{1 \le n' \le N, 1 \le t' \le T} \sup_{0 \le j \le Q} \max_{1 \le k \le m_j} \frac{\frac{\ln \ln T}{NT^2} \left| \sum_{i=1}^{n'} \sum_{t=1}^{t'} (\Delta_i f_t)^2 d_{2t} \left(\theta_{k+1}, \theta_k\right) \right|}{|\theta_j - \theta^0|} > \varepsilon \right)$$

$$\leq c_0 \sum_{j=0}^{\infty} \sum_{k=2}^{m_j+1} \sum_{N=1}^{\infty} \sum_{T=1}^{\infty} c^{-2j} \frac{\varepsilon^{-2}}{\overline{v}_{N,T}^2 \left(\epsilon\right) NT} \left(\ln N \ln T\right)^2$$

$$\leq c_0 \left(\sum_{j=0}^{\infty} c^{-j} \right) \sum_{N=1}^{\infty} \sum_{T=1}^{\infty} \frac{\varepsilon^{-2}}{\overline{v}_{NT} NT} \left(\ln N \ln T\right)^2 < \infty,$$

so that $b = o_{a.s.}(1)$. Putting all together, the final result now obtains.

Lemma C.5. We assume that Assumptions 3-5 are satisfied. Then there exist two random variables N_0 and T_0 , and a positive, finite constant c_0 , such that for $N \ge N_0$ and $T \ge T_0$, it holds that

$$\sup_{|\theta-\theta^{0}|\in B_{N,T}}\frac{\left|h^{0}\left(\eta,\theta\right)-h^{0}\left(\eta,\theta^{0}\right)\right|}{\left|\theta-\theta^{0}\right|}=o_{a.s.}\left(1\right).$$

Proof. The proof is very similar to the one of the previous lemma, and we only report the main passages to save space. Again, set $\eta = 1$ and write $h^0(\eta, \theta) = h^0(\theta)$ (and $h^0(\eta, \theta^0) = h^0(\theta^0)$) for short; we have

$$\sup_{\theta \in V_{c}(\bar{v}_{NT})} \frac{h^{0}(\theta)}{|\theta - \theta^{0}|} \leq 2 \sup_{0 \leq j \leq Q} \frac{\frac{1}{NT} \sum_{i=1}^{N} \sum_{t=1}^{T} e_{i,t} \delta_{i}^{0'} \underline{\mathbf{u}}_{t}^{0} [d_{\mathcal{S},t}(\theta_{j}) - d_{\mathcal{S},t}(\theta^{0})]}{|\theta_{j} - \theta^{0}|}$$

$$+ 2 \sup_{0 \leq j \leq Q} \sup_{\theta_{j} \leq \theta \leq \theta_{j+1}} \frac{\frac{1}{NT} \sum_{i=1}^{N} \sum_{t=1}^{T} e_{i,t} \delta_{i}^{0'} \underline{\mathbf{u}}_{t}^{0} [d_{\mathcal{S},t}(\theta) - d_{\mathcal{S},t}(\theta_{j})]}{|\theta_{j} - \theta^{0}|}$$

$$= 2 \sup_{0 \leq j \leq Q} \frac{h^{0}(\theta_{j}, \theta^{0})}{|\theta_{j} - \theta^{0}|} + 2 \sup_{0 \leq j \leq Q} \sup_{\theta_{j} \leq \theta \leq \theta_{j+1}} \frac{h^{0}(\theta, \theta_{j})}{|\theta_{j} - \theta^{0}|} = I + II.$$
(C.20)

Let

$$h_{n',t'}^{0}\left(\theta_{j},\theta^{0}\right) = \frac{1}{NT}\sum_{i=1}^{n'}\sum_{t=1}^{t'}e_{i,t}\delta_{i}^{0'}\mathbf{\underline{u}}_{t}^{0}\left[d_{\mathcal{S},t}\left(\theta_{j}\right) - d_{\mathcal{S},t}\left(\theta^{0}\right)\right].$$

It holds that

$$\begin{split} &\sum_{N=1}^{\infty} \sum_{T=1}^{\infty} \frac{1}{NT} P\left(\max_{1 \le n' \le N, 1 \le t' \le T} \sup_{0 \le j \le Q} \frac{h_{n',t'}^{0}\left(\theta_{j},\theta^{0}\right)}{|\theta_{j}-\theta^{0}|} > N^{-1/2} \right) \\ &\le &\sum_{j=0}^{\infty} \sum_{N=1}^{\infty} \sum_{T=1}^{\infty} \frac{1}{NT} P\left(\max_{1 \le n' \le N, 1 \le t' \le T} h_{n',t'}^{0}\left(\theta_{j},\theta^{0}\right) > N^{-1/2} \left|\theta_{j}-\theta^{0}\right| \right) \\ &\le &\sum_{j=0}^{\infty} c^{-2j} \sum_{N=1}^{\infty} \sum_{T=1}^{\infty} \frac{N}{NT} \frac{T^{2}}{\overline{v}_{NT}^{2}\left(\epsilon\right)} E\left| \max_{1 \le n' \le N, 1 \le t' \le T} h_{n',t'}^{0}\left(\theta_{j},\theta^{0}\right) \right|^{2}. \end{split}$$

Also, using Assumption 4(v) it can be shown after some algebra that $E |h^0(\theta_j, \theta^0)|^2 \le c_0 (NT)^{-1} |\theta_j - \theta^0|$; hence

$$\sum_{N=1}^{\infty} \sum_{T=1}^{\infty} \frac{1}{NT} P\left(\max_{1 \le n' \le N, 1 \le t' \le T} \sup_{0 \le j \le Q} \frac{h_{n',t'}^0\left(\theta_j, \theta^0\right)}{|\theta_j - \theta^0|} > \varepsilon\right) \le c_0 \sum_{N=1}^{\infty} \sum_{T=1}^{\infty} \frac{1}{\overline{v}_{NT}\left(\epsilon\right) NT} \left(\ln N \ln T\right)^2 < \infty.$$

Thus, in (D.12), $I = O_{a.s.} (N^{-1/2})$. We now turn to II. We have

$$\sum_{N=1}^{\infty} \sum_{T=1}^{\infty} \frac{1}{NT} P\left(\max_{1 \le n' \le N, 1 \le t' \le T} \sup_{0 \le j \le Q} \sup_{\theta_j \le \theta \le \theta_{j+1}} \frac{h_{n',t'}^0\left(\theta,\theta_j\right)}{|\theta_j - \theta^0|} \ge NT\right)$$

$$\leq \sum_{j=0}^{\infty} \sum_{N=1}^{\infty} \sum_{T=1}^{\infty} \frac{1}{NT} P\left(\max_{1 \le n' \le N, 1 \le t' \le T} \max_{2 \le k \le m_j+1} h_{n',t'}^0\left(\theta_k,\theta_j\right) \ge NT \left|\theta_j - \theta^0\right|\right)$$

$$+ \sum_{j=0}^{\infty} \sum_{N=1}^{\infty} \sum_{T=1}^{\infty} \frac{1}{NT} P\left(\max_{1 \le n' \le N, 1 \le t' \le T} \max_{1 \le k \le m_j} h_{n',t'}^0\left(\theta_{k+1},\theta_k\right) \ge NT \left|\theta_j - \theta^0\right|\right)$$

having defined

$$\begin{aligned} h_{n',t'}^{0}\left(\theta,\theta_{j}\right) &= \sum_{i=1}^{n'} \sum_{t=1}^{t'} e_{i,t} \delta_{i}^{0'} \underline{\mathbf{u}}_{t}^{0} \widetilde{d}_{2t}^{'}\left(\theta,\theta_{j}\right), \\ h_{n',t'}^{0}\left(\theta_{k},\theta_{j}\right) &= \sum_{i=1}^{n'} \sum_{t=1}^{t'} e_{i,t} \delta_{i}^{0'} \underline{\mathbf{u}}_{t}^{0} \widetilde{d}_{2t}^{'}\left(\theta_{k},\theta_{j}\right), \\ h_{n',t'}^{0}\left(\theta_{k+1},\theta_{k}\right) &= \sum_{i=1}^{n'} \sum_{t=1}^{t'} e_{i,t} \delta_{i}^{0'} \underline{\mathbf{u}}_{t}^{0} \widetilde{d}_{2t}^{'}\left(\theta_{k+1},\theta_{k}\right), \end{aligned}$$

XVIII

with $\tilde{d}'_{2t}(\theta, \theta_j) = d_{\mathcal{S},t}(\theta) - d_{\mathcal{S},t}(\theta_j)$, and $\tilde{d}'_{2t}(\theta_k, \theta_j)$ and $\tilde{d}'_{2t}(\theta_{k+1}, \theta_k)$ defined similarly. We have

$$\sum_{j=0}^{\infty} \sum_{N=1}^{\infty} \sum_{T=1}^{\infty} \frac{1}{NT} P\left(\max_{1 \le n' \le N, 1 \le t' \le T} \max_{2 \le k \le m_j+1} h_{n',t'}^0\left(\theta_k, \theta_j\right) \ge NT \left|\theta_j - \theta^0\right|\right)$$

$$\leq \sum_{j=0}^{\infty} \sum_{k=2}^{m_j+1} \sum_{N=1}^{\infty} \sum_{T=1}^{\infty} \frac{1}{NT} P\left(\max_{1 \le n' \le N, 1 \le t' \le T} h_{n',t'}^0\left(\theta_k, \theta_j\right) \ge NT \left|\theta_j - \theta^0\right|\right)$$

$$\leq \sum_{j=0}^{\infty} \sum_{k=2}^{m_j+1} \sum_{N=1}^{\infty} \sum_{T=1}^{\infty} \frac{(NT)^{-(2+r)}}{NT} \frac{c^{-(2+r)j}T^{2+r}}{\overline{v}_{NT}^{2+r}\left(\epsilon\right)} E\left|\max_{1 \le n' \le N, 1 \le t' \le T} h_{n',t'}^0\left(\theta_k, \theta_j\right)\right|^{2+r}.$$

Also, using again Theorem 1.1 in Shao (1995), it can be shown that

$$E\left|\sum_{i=1}^{N}\sum_{t=1}^{T}e_{i,t}\delta_{i}^{0'}\underline{\mathbf{u}}_{t}^{0}\widetilde{d}_{2t}^{\prime}\left(\theta_{k},\theta_{j}\right)\right|^{2+r} \leq c_{0}N^{2+r}c^{j(1+r/2)}\overline{v}_{NT}^{1+r/2}.$$

Thus, by Corollary 4 in Moricz (1983)

$$\sum_{j=0}^{\infty} \sum_{N=1}^{\infty} \sum_{T=1}^{\infty} \frac{1}{NT} P\left(\max_{1 \le n' \le N, 1 \le t' \le T} \max_{2 \le k \le m_j+1} h_{n',t'}^0(\theta_k, \theta_j) \ge NT \left| \theta_j - \theta^0 \right| \right)$$
$$\leq c_0 \sum_{j=0}^{\infty} \sum_{k=2}^{m_j+1} \sum_{N=1}^{\infty} \sum_{T=1}^{\infty} \frac{1}{NT} \frac{c^{-jr/2}}{\overline{v}_{NT}^{r/2}(\epsilon)} \left(\ln N \ln T\right)^{2+r}$$

Similarly

$$\sum_{j=0}^{\infty} \sum_{N=1}^{\infty} \sum_{T=1}^{\infty} \frac{1}{NT} P\left(\max_{1 \le n' \le N, 1 \le t' \le T} \max_{1 \le k \le m_j} h_{n',t'}^0(\theta_{k+1}, \theta_k) \ge NT \left| \theta_j - \theta^0 \right| \right)$$

$$\leq \sum_{j=0}^{\infty} \sum_{k=1}^{m_j} \sum_{N=1}^{\infty} \sum_{T=1}^{\infty} \frac{1}{NT} P\left(\max_{1 \le n' \le N, 1 \le t' \le T} h_{n',t'}^0(\theta_{k+1}, \theta_k) \ge NT \left| \theta_j - \theta^0 \right| \right)$$

$$\leq \sum_{j=0}^{\infty} \sum_{k=1}^{m_j} \sum_{N=1}^{\infty} \sum_{T=1}^{\infty} \frac{(NT)^{-2}}{NT} \frac{c^{-2j}T^2}{\overline{v}_{NT}^2(\epsilon)} E \left| \max_{1 \le n' \le N, 1 \le t' \le T} h_{n',t'}^0(\theta_{k+1}, \theta_k) \right|^2,$$

with, after some algebra

$$E\left|\sum_{i=1}^{N}\sum_{t=1}^{T}e_{i,t}\delta_{i}^{0\prime}\underline{\mathbf{u}}_{t}^{0}\widetilde{d}_{2t}^{\prime}\left(\theta_{k+1},\theta_{k}\right)\right|^{2}\leq c_{0}N.$$

Thus

$$\sum_{j=0}^{\infty} \sum_{N=1}^{\infty} \sum_{T=1}^{\infty} \frac{1}{NT} P\left(\max_{1 \le n' \le N, 1 \le t' \le T} \max_{2 \le k \le m_j+1} h_{n',t'}^0(\theta_k, \theta_j) \ge N^{1/2} T \left| \theta_j - \theta^0 \right| \right)$$

$$\le \left(\sum_{j=0}^{\infty} c^{-j} \right) \sum_{N=1}^{\infty} \sum_{T=1}^{\infty} \frac{(NT^2)^{-1}}{NT} \frac{T^2}{\overline{v}_{NT}^2(\epsilon)} N \left(\ln N \ln T\right)^2 < \infty.$$

The desired result now follows.

Lemma C.6. We assume that Assumptions 1-4 are satisfied. Then it holds that, for every $P_j \ge P_j^0$

$$\left\|\widehat{\beta}_{j,i}^{P_{j'}}\widehat{\mathbf{u}}_{j,t}\widehat{d}_{j,t} - \beta_{j,i}^{0\prime}\mathbf{u}_{j,t}^{0}d_{j,t}\left(\theta^{0}\right)\right\| = o_{a.s.}\left(1\right),\tag{C.21}$$

for $j = \mathcal{D}, \mathcal{S}$ and all $1 \leq i \leq N, 1 \leq t \leq T$. Also, $S\left(\mathbf{B}^{P}, \mathbf{U}^{P}, \theta\right)$ is continuous in (\mathbf{B}, \mathbf{U}) .

Proof. Using the results in the proof of Lemma A.8 in Massacci (2017), it can be shown that $E \left\| \widehat{\beta}_{j,i}^{P_j} - \widehat{H}_{jj}^{P_j} (\theta^0) \beta_{j,i}^0 \right\|^2 \leq c_0 C_{N,T}^{-2}$. Thus, using the same logic as above, Lemma C.1 entails that for all $1 \leq i \leq N$, $\left\| \widehat{\beta}_{j,i}^{P_j} - \widehat{H}_{jj}^{P_j} (\theta^0) \beta_{j,i}^0 \right\| = o_{a.s.} \left(C_{N,T}^{-1} (\ln N)^{1+\epsilon} (\ln T)^{1+\epsilon} \right)$. With the same logic, by Corollary 3.2 in Massacci (2017), it follows that, for all $1 \leq t \leq T$, $\left\| \widehat{\mathbf{u}}_{j,t} \widehat{d}_{j,t} - \left(\widehat{H}_{jj}^{P_j} (\theta^0)' \right)^{-1} \mathbf{u}_{j,t}^0 d_{j,t} (\theta^0) \right\| = o_{a.s.} \left(C_{N,T}^{-1} (\ln N)^{1+\epsilon} (\ln T)^{1+\epsilon} \right)$. Equation (C.21) now follows immediately. As a consequence, we can assume that

$$\lim_{N,T\to\infty}\widehat{\beta}_{j,i}^{P_j\prime}\widehat{\mathbf{u}}_{j,t}\widehat{d}_{j,t} = \beta_{j,i}^{0\prime}\mathbf{u}_{j,t}^0d_{j,t}\left(\theta^0\right).$$
(C.22)

Equation (C.22) and elementary algebra now give $\lim_{N,T\to\infty} S\left(\widehat{\mathbf{B}}^{P}, \widehat{\mathbf{U}}^{P}, \theta\right) = S\left(\mathbf{B}^{0}, \mathbf{U}^{0}, \theta\right)$, which proves the last statement of the lemma.

We are now ready to derive a strong rate for $\hat{\theta}$.

Lemma C.7. We assume that Assumptions 1-5 are satisfied. Then it holds that

$$\widehat{\theta} - \theta^0 = o_{a.s.} \left(T^{-1} \overline{v}_{N,T} \left(\epsilon \right) \right),$$

for every $\epsilon > 0$.

Proof. Lemma C.3 entails that we can focus on the event $|\hat{\theta} - \theta^0| < c_B$. Further, after some algebra, Lemma C.6 entails that there exists two random variables N_0 and T_0 such that, for $N \geq N_0$ and $T \geq T_0$, we have

$$\frac{S\left(\widehat{\mathbf{B}}^{P}, \widehat{\mathbf{U}}^{P}, \theta\right) - S\left(\widehat{\mathbf{B}}^{P}, \widehat{\mathbf{U}}^{P}, \theta^{0}\right)}{|\theta - \theta^{0}|} \\ \geq \inf_{|\theta - \theta_{0}| \in B_{N,T}} \frac{E\omega^{0}\left(\eta, \theta\right)}{|\theta - \theta^{0}|} - \sup_{|\theta - \theta_{0}| \in B_{N,T}} \frac{\left|\overline{\omega}_{N,T}^{0}\right|}{|\theta - \theta^{0}|} - 2\sup_{|\theta - \theta_{0}| \in B_{N,T}} \frac{\left|h^{0}\left(\eta, \theta\right) - h^{0}\left(\eta, \theta^{0}\right)\right|}{|\theta - \theta^{0}|}.$$

By using Lemmas C.4 and C.5, it follows that, for all θ such that $|\theta - \theta_0| \in B_{N,T}$

$$S\left(\widehat{\mathbf{B}}^{P}, \widehat{\mathbf{U}}^{P}, \theta\right) - S\left(\widehat{\mathbf{B}}^{P}, \widehat{\mathbf{U}}^{P}, \theta^{0}\right) \ge c_{0} > 0 \text{ a.s.}$$

Since, by definition, $S\left(\widehat{\mathbf{B}}^{P}, \widehat{\mathbf{U}}^{P}, \widehat{\theta}\right) \leq S\left(\widehat{\mathbf{B}}^{P}, \widehat{\mathbf{U}}^{P}, \theta^{0}\right)$, this entails that $\left|\widehat{\theta} - \theta^{0}\right| \notin B_{N,T}$, whence the desired result.

Lemma C.8. We assume that Assumptions 1-5 are satisfied. Then it holds that

$$\frac{T_j - \hat{T}_j}{\hat{T}_j} = o_{a.s.} \left(T_j^{-1} \left(\ln T \right)^{1+\epsilon'} \overline{v}_{N,T} \left(\epsilon \right) \right),$$

for $j = \mathcal{D}, \mathcal{S}$ and every $\epsilon, \epsilon' > 0$.

Proof. We show the lemma for $j = \mathcal{D}$. Recall that $T_{\mathcal{D}} = \sum_{t=1}^{T} d_{\mathcal{D},t}(\theta^0)$ and $\widehat{T}_{\mathcal{D}} = \sum_{t=1}^{T} \widehat{d}_{\mathcal{D},t}$. It holds that

$$T_{\mathcal{D}} - \widehat{T}_{\mathcal{D}} \le \sum_{t=1}^{T} \left| d_{\mathcal{D},t} \left(\theta^{0} \right) - \widehat{d}_{\mathcal{D},t} \right|$$

Denoting the density of z_t by $f_t(z) = f(z)$, Assumption 5(*iii*) entails

$$E\left|\widehat{d}_{\mathcal{D},t} - d_{\mathcal{D},t}\left(\theta^{0}\right)\right| = \left|\int_{\widehat{\theta}}^{\theta^{0}} f\left(z\right) dz\right| \le c_{0}\left|\widehat{\theta} - \theta^{0}\right|,\tag{C.23}$$

so that

$$T_{\mathcal{D}} - \widehat{T}_{\mathcal{D}} \le c_0 T \left| \widehat{\theta} - \theta^0 \right| \left(\ln T \right)^{1+\epsilon},$$

having used Lemma C.1. The desired result follows immediately from Lemma C.7.

Lemma C.9. We assume that Assumptions 3-5 are satisfied. Then it holds that

$$E\left|\tilde{R}_{i,t}\right|^{4+\epsilon} < \infty \tag{C.24}$$

for some $\epsilon > 0$, and

$$E \max_{1 \le t \le T} \left| \sum_{s=1}^{t} \widetilde{R}_{h,s} \widetilde{R}_{l,s} d_{\mathcal{D},s} \left(\theta^{0} \right) - E \left[\widetilde{R}_{h,s} \widetilde{R}_{l,s} d_{\mathcal{D},s} \left(\theta^{0} \right) \right] \right|^{2} \le c_{1} T, \qquad (C.25)$$

$$E \max_{1 \le t \le T} \left| \sum_{s=1}^{t} \widetilde{R}_{h,s} \widetilde{R}_{l,s} d_{\mathcal{S},s} \left(\theta^{0} \right) - E \left[\widetilde{R}_{h,s} \widetilde{R}_{l,s} d_{\mathcal{S},s} \left(\theta^{0} \right) \right] \right|^{2} \le c_{2}T,$$
(C.26)

for all $1 \leq h, l \leq N$.

Proof. Equation (C.24) is an immediate consequence of Assumptions 3(i) and 4(i). We show only (C.25); (C.26) follows from exactly the same arguments. To begin with, note that Assumption 5(i) entails that $\tilde{R}_{h,s}\tilde{R}_{l,s}d_{1,s}(\theta^0)$ is also a strictly stationary, ρ -mixing sequence with mixing numbers ρ_m such that $\sum_{m=1}^{\infty} \rho_m^{1/2} < \infty$ - we refer to the proof of Lemma C.11 for details. Using (C.24), it follows that

$$E\left|\widetilde{R}_{h,s}\widetilde{R}_{l,s}d_{\mathcal{D},s}\left(\theta^{0}\right)\right|^{2} \leq \left(E\left|\widetilde{R}_{h,s}\right|^{4}\right)^{1/2} \left(E\left|\widetilde{R}_{l,s}\right|^{4}\right)^{1/2} < \infty.$$

Note also that

$$\sum_{m=1}^{\infty} \rho_m^{1/2} < \infty \Rightarrow \sum_{m=1}^{\infty} \frac{\rho_m}{m} < \infty \Rightarrow \sum_{m=1}^{\infty} \rho_m \left(2^m\right) < \infty.$$

Hence, Corollary 1.1 in Shao (1995) and the strict stationarity of $\tilde{R}_{h,s}\tilde{R}_{l,s}d_{\mathcal{D},s}(\theta^0)$ immediately yield (C.25).

Lemma C.10. We assume that Assumptions 1-6 are satisfied. Then it holds that

$$\begin{split} &\lim \sup_{\min(N,T)\to\infty} \frac{1}{N-p} \sum_{h=p+1}^{N} \widehat{g}_{j}^{(h)} = \overline{g}_{j} < \infty, \\ &\lim \inf_{\min(N,T)\to\infty} \frac{1}{N-p} \sum_{h=p+1}^{N} \widehat{g}_{j}^{(h)} = \underline{g}_{j} > 0, \end{split}$$

for $j = \mathcal{D}, \mathcal{S}$ and every $0 \le p \le P_{\max}$.

Proof. Note that

$$\sum_{h=p+1}^{N} \widehat{g}_{j}^{(h)} - \sum_{h=p+1}^{N} g_{j}^{(h)}$$

$$= \sum_{h=1}^{N} \widehat{g}_{j}^{(h)} - \sum_{h=1}^{N} g_{j}^{(h)} - \sum_{h=1}^{p} \widehat{g}_{j}^{(h)} + \sum_{h=1}^{p} g_{j}^{(h)}$$

$$= tr\left(\widehat{\Sigma}_{j}\right) - tr\left(\Sigma_{j}\right) - \left(\sum_{h=1}^{p} \left(\widehat{g}_{j}^{(h)} - g_{j}^{(h)}\right)\right).$$

Using Lemma A.2, it is easy to see that

$$\sum_{h=1}^{p} \left(\widehat{g}_{j}^{(h)} - g_{j}^{(h)} \right) = o_{a.s.} \left(\frac{pNT^{1/2}}{T\pi_{j}} \left(\ln N \right)^{1+\epsilon} \left(\ln T \right)^{1/2+\epsilon} \right).$$
(C.27)

Also, the same passages as in the proof of Lemma A.1 in Trapani (2018) yield

$$tr\left(\widehat{\Sigma}_{j}\right) - tr\left(\Sigma_{j}\right) = o_{a.s.}\left(\frac{N\left(\ln N\ln T\right)^{\frac{1+\epsilon}{2}}}{T^{1/2}\pi_{j}}\right).$$
(C.28)

Thus, recalling that $P_{\max} = O\left(\min\left\{T^{1/2-c}, N^{1/2-c}\right\}\right)$, it follows that

$$\frac{1}{N-p}\sum_{h=p+1}^{N}\widehat{g}_{j}^{(h)} = \frac{1}{N-p}\sum_{h=p+1}^{N}g_{j}^{(h)} + o_{a.s.}\left(\frac{\left(\ln N\ln T\right)^{\frac{1+\epsilon}{2}}}{T^{1/2}\pi_{j}}\right) = \frac{1}{N-p}\sum_{h=p+1}^{N}g_{j}^{(h)} + o_{a.s.}\left(1\right),$$

under Assumption 6.

Let now $g_j^{u,(h)}$ and $g_j^{e,(h)}$ denote the *h*-th largest eigenvalues of $\frac{1}{T\pi_j} \sum_{t=1}^T E\left(\mathbf{B}_j^0 \mathbf{u}_{j,t}^0 \mathbf{u}_{j,t}^{0\prime} \mathbf{B}_j^{0\prime} d_{j,t} \left(\theta^0\right)\right)$

and $\frac{1}{T\pi_j} \sum_{t=1}^{T} E\left(\mathbf{e}_t \mathbf{e}'_t d_{j,t}\left(\theta^0\right)\right)$ respectively. By Weyl's inequality

$$g_j^{e,(N)} + g_j^{u,(h)} \le g_j^{(h)} \le g_j^{u,(h)} + g_j^{e,(1)},$$

so that

$$g_j^{e,(N)} + \frac{1}{N-p} \sum_{h=p+1}^N g_j^{u,(h)} \le \frac{1}{N-p} \sum_{h=p+1}^N g_j^{(h)} \le \frac{1}{N-p} \sum_{h=p+1}^N g_j^{u,(h)} + g_j^{e,(1)}.$$

By Assumption 2(i)(b), it holds that

$$g_j^{e,(N)} + \frac{1}{N-p} \sum_{h=p+1}^N g_j^{u,(h)} > 0;$$

also, Assumption 2(i)(a) and (D.1) entail

$$\frac{1}{N-p}\sum_{h=p+1}^{N}g_{j}^{u,(h)}+g_{j}^{e,(1)}<\infty.$$

Hence, as $N \to \infty$

$$0 < \lim \inf_{N \to \infty} < \frac{1}{N-p} \sum_{h=p+1}^{N} g_j^{(h)} < \lim \sup_{N \to \infty} < \infty.$$

Putting all together, the lemma follows.

Lemma C.11. We assume that Assumptions 1-5 are satisfied. Then it holds that, for j = D, S

$$E \left\| T^{-1} \sum_{t=1}^{T} \mathbf{u}_{j,t}^{0} d_{j,t} \left(\theta^{0} \right) \right\|^{2} \le c_{0} T^{-1}.$$
 (C.29)

Further, under Assumption 9(ii), as $T \to \infty$

$$T^{-1/2} \sum_{t=1}^{T} \mathbf{u}_{j,t}^{0} d_{j,t} \left(\theta^{0} \right) \xrightarrow{D} N \left(0, V_{\mathbf{u},j} \right), \qquad (C.30)$$

where $V_{\mathbf{u},j} = \lim_{T \to \infty} E\left(T^{-1} \sum_{t,s=1}^{T} \mathbf{u}_{j,t}^{0} \mathbf{u}_{j,s}^{\prime 0} d_{j,t}\left(\theta^{0}\right) d_{j,s}\left(\theta^{0}\right)\right).$

Proof. We begin by showing that $\left\{\mathbf{f}_{j,t}^{0}d_{j,t}\left(\theta^{0}\right)\right\}$ is a ρ -mixing sequence with mixing number $\tilde{\rho}_{j,m}$

such that $\sum_{m=1}^{\infty} \tilde{\rho}_{j,m}^{1/2} < \infty$. Note that $\mathbf{f}_{j,t}^{0} d_{j,t} (\theta^{0})$ is a measurable transformation of $\{\mathbf{f}_{j,t}^{0}, z_{t}, \epsilon_{t}\}$. Let now $\mathfrak{I}_{-\infty}^{t}$ and $\mathfrak{I}_{t+m}^{\infty}$ be the σ -fields generated by $\{\mathbf{f}_{j,t'}^{0}, z_{t'}, \epsilon_{t'}\}_{t'=-\infty}^{t}$ and $\{\mathbf{f}_{j,t'}^{0}, z_{t'}, \epsilon_{t'}\}_{t'=t+m}^{\infty}$ respectively, and similarly $\mathfrak{T}_{-\infty}^{t}$ and $\mathfrak{T}_{t+m}^{\infty}$ be the σ -fields generated by $\{\mathbf{f}_{j,t'}^{0}, z_{t'}, \epsilon_{t'}\}_{t'=-\infty}^{t}$ and $\{\mathbf{f}_{j,t'}^{0} d_{j,t'} (\theta^{0})\}_{t'=-\infty}^{t}$ and $\{\mathbf{f}_{j,t'}^{0} d_{j,t'} (\theta^{0})\}_{t'=t+m}^{t}$ respectively. By measurability, $\mathfrak{T}_{-\infty}^{t} \subseteq \mathfrak{T}_{-\infty}^{t}$ and $\mathfrak{T}_{t+m}^{\infty} \subseteq \mathfrak{T}_{t+m}^{\infty}$; hence, $\tilde{\rho}_{j,m} = \tilde{\rho}_{j,m} (\mathfrak{T}_{-\infty}^{t}, \mathfrak{T}_{t+m}^{\infty}) \leq \rho_{j,m} = \rho_{j,m} (\mathfrak{T}_{-\infty}^{t}, \mathfrak{T}_{t+m}^{\infty})$. This therefore entails that $\sum_{m=1}^{\infty} \tilde{\rho}_{j,m}^{1/2} < \sum_{m=1}^{\infty} \rho_{j,m}^{1/2} < \infty$ by Assumption $\mathfrak{I}(i)$, which in turn entails that $\sum_{k=1}^{\infty} \tilde{\rho}_{j} (2^{k}) < \infty$.

Note now that $E \left\| \mathbf{u}_{j,t}^{0} d_{j,t} \left(\theta^{0} \right) \right\|^{2} \leq E \left\| \mathbf{u}_{j,t}^{0} \right\|^{2}$, which is finite by Assumption 3(i). Thus

$$E\left\|\sum_{t=1}^{T}\mathbf{u}_{j,t}^{0}d_{j,t}\left(\theta^{0}\right)\right\|^{2} \leq E\max_{1\leq k\leq T}\left\|\sum_{t=1}^{k}\mathbf{u}_{j,t}^{0}d_{j,t}\left(\theta^{0}\right)\right\|^{2} \leq c_{0}T,$$

where the last inequality follows from Lemma 1 in Peligrad, Utev, and Wu (2007). This proves (C.29). As far as (C.30) is concerned, it follows readily upon checking the assumptions of Theorem 0 in Peligrad (1987). \Box

D Proofs

Henceforth, we use E^* and V^* to denote, respectively, the expected value and the variance with respect to P^* .

Proof of Lemma A.1. The lemma is an adaptation of Lemma 1 in Trapani (2018). In order to prove it, note that Assumption 4(v) entails that, for $j = \mathcal{D}, \mathcal{S}$

$$\frac{1}{T\pi_{j}}\sum_{t=1}^{T}E\left[\left(\mathbf{B}_{j}^{0}\mathbf{u}_{j,t}^{0}+\mathbf{e}_{t}\right)\left(\mathbf{B}_{j}^{0}\mathbf{u}_{j,t}^{0}+\mathbf{e}_{t}\right)'d_{j,t}\left(\theta^{0}\right)\right]$$

= $\mathbf{B}_{j}^{0}\frac{1}{T\pi_{j}}\sum_{t=1}^{T}E\left(\mathbf{u}_{j,t}^{0}\mathbf{u}_{j,t}^{0\prime}d_{j,t}\left(\theta^{0}\right)\right)\mathbf{B}_{j}^{0\prime}+\frac{1}{T\pi_{j}}\sum_{t=1}^{T}E\left(\mathbf{e}_{t}\mathbf{e}_{t}'d_{j,t}\left(\theta^{0}\right)\right)$

We begin by showing that

$$g_{j}^{(i)} \left(\mathbf{B}_{j}^{0} \frac{1}{T \pi_{j}} \sum_{t=1}^{T} E\left(\mathbf{u}_{j,t}^{0} \mathbf{u}_{j,t}^{0\prime} d_{j,t} \left(\theta^{0} \right) \right) \mathbf{B}_{j}^{0\prime} \right) \begin{cases} \in \left(\underline{c}_{j}^{(i)} N, \overline{c}_{j}^{(i)} N \right) & \text{for } 0 \le i \le P_{j}^{0} \\ = 0 & \text{for } i \ge P_{j}^{0} + 1 \end{cases},$$
(D.1)

for $j = \mathcal{D}, \mathcal{S}$, where $0 < \underline{c}_{j}^{(i)} \leq \overline{c}_{j}^{(i)} < \infty$. We show this result for the case $j = \mathcal{D}$ only - the case $j = \mathcal{S}$ is repetitive. By the multiplicative Weyl's inequality (Merikoski and Kumar (2004)) we have

$$g_{\mathcal{D}}^{(i)}\left(\mathbf{B}_{\mathcal{D}}^{0}\frac{1}{T\pi_{\mathcal{D}}}\sum_{t=1}^{T}E\left(\mathbf{u}_{\mathcal{D},t}^{0}\mathbf{u}_{\mathcal{D},t}^{0\prime}d_{\mathcal{D},t}\left(\theta^{0}\right)\right)\mathbf{B}_{\mathcal{D}}^{0\prime}\right) \geq g_{\mathcal{D}}^{(\min)}\left(\frac{1}{T\pi_{\mathcal{D}}}\sum_{t=1}^{T}E\left(\mathbf{u}_{\mathcal{D},t}^{0}\mathbf{u}_{\mathcal{D},t}^{0\prime}d_{\mathcal{D},t}\left(\theta^{0}\right)\right)\right)g_{\mathcal{D}}^{(i)}\left(\mathbf{B}_{\mathcal{D}}^{0\prime}\mathbf{B}_{\mathcal{D}}^{0}\right),$$
$$g_{\mathcal{D}}^{(i)}\left(\mathbf{B}_{\mathcal{D}}^{0}\frac{1}{T\pi_{\mathcal{D}}}\sum_{t=1}^{T}E\left(\mathbf{u}_{\mathcal{D},t}^{0}\mathbf{u}_{\mathcal{D},t}^{0\prime}d_{\mathcal{D},t}\left(\theta^{0}\right)\right)\mathbf{B}_{\mathcal{D}}^{0\prime}\right) \leq g_{\mathcal{D}}^{(\max)}\left(\frac{1}{T\pi_{\mathcal{D}}}\sum_{t=1}^{T}E\left(\mathbf{u}_{\mathcal{D},t}^{0}\mathbf{u}_{\mathcal{D},t}^{0\prime}d_{\mathcal{D},t}\left(\theta^{0}\right)\right)\right)g_{\mathcal{D}}^{(i)}\left(\mathbf{B}_{\mathcal{D}}^{0\prime}\mathbf{B}_{\mathcal{D}}^{0}\right),$$

by Assumption 3(ii) it holds that

$$g_{\mathcal{D}}^{(\min)}\left(\frac{1}{T\pi_{\mathcal{D}}}\sum_{t=1}^{T}E\left(\mathbf{u}_{\mathcal{D},t}^{0}\mathbf{u}_{\mathcal{D},t}^{0\prime}d_{\mathcal{D},t}\left(\theta^{0}\right)\right)\right) > 0.$$

Moreover, by Assumption 3(iv) we have

$$g_{\mathcal{D}}^{(i)} \left(\mathbf{B}_{\mathcal{D}}^{0} \mathbf{B}_{\mathcal{D}}^{0'} \right) \begin{cases} \geq c_{1}^{(i)} N & \text{for } 1 \leq i \leq P_{\mathcal{D}}^{0} \\ = 0 & \text{for } i \geq P_{\mathcal{D}}^{0} + 1 \end{cases}$$

•

Equation (D.1) now follows immediately. Consider now

$$\frac{1}{T\pi_{\mathcal{D}}}\sum_{t=1}^{T}\mathbf{e}_{t}\mathbf{e}_{t}'d_{\mathcal{D},t}\left(\theta^{0}\right) = \frac{1}{T\pi_{j}}\sum_{t=1}^{T}\epsilon_{t}\epsilon_{t}'d_{\mathcal{D},t}\left(\theta^{0}\right) - \bar{\epsilon}_{\mathcal{D}}\bar{\epsilon}_{\mathcal{D}}',$$

so that

$$g_{\mathcal{D}}^{(1)}\left(\frac{1}{T\pi_{\mathcal{D}}}\sum_{t=1}^{T}E\left(\mathbf{e}_{t}\mathbf{e}_{t}^{\prime}d_{\mathcal{D},t}\left(\theta^{0}\right)\right)\right) \leq g_{\mathcal{D}}^{(1)}\left(\frac{1}{T\pi_{\mathcal{D}}}\sum_{t=1}^{T}E\left(\epsilon_{t}\epsilon_{t}^{\prime}d_{\mathcal{D},t}\left(\theta^{0}\right)\right)\right) + g_{\mathcal{D}}^{(1)}\left(\overline{\epsilon}_{\mathcal{D}}\overline{\epsilon}_{\mathcal{D}}^{\prime}\right).$$

The first term is bounded by Assumption 2(i). Also, after some algebra we have

$$g_{\mathcal{D}}^{(1)}\left(\overline{\epsilon}_{\mathcal{D}}\overline{\epsilon}_{\mathcal{D}}'\right) \leq \max_{1\leq i\leq N} \sum_{k=1}^{N} \frac{1}{\left(T\pi_{\mathcal{D}}\right)^{2}} \sum_{t=1}^{T} \sum_{s=1}^{T} \left| E\left(\epsilon_{i,t}\epsilon_{k,s}d_{\mathcal{D},t}\left(\theta^{0}\right)d_{\mathcal{D},s}\left(\theta^{0}\right)\right) \right|,$$

which is bounded by Assumption 2(ii). The proof of the lemma now follows immediately along

the same lines as the proof of Lemma 1 in Trapani (2018).

Proof of Theorem A.2. We prove the theorem for j = 1 - again, the proof for the case j = 2 is just a repetition of the same arguments. Note that

$$\left|\widehat{g}_{1}^{(i)} - g_{1}^{(i)}\right| \leq \left\|\widehat{\Sigma}_{\mathcal{S}} - \Sigma_{\mathcal{D}}\right\|_{op}$$

so that, by symmetry

$$\left| \hat{g}_{1}^{(i)} - g_{1}^{(i)} \right| \leq \left| \sum_{h=1}^{N} \sum_{l=1}^{N} \left(\frac{1}{T \pi_{\mathcal{D}}} \sum_{t=1}^{T} \left(\tilde{R}_{h,t} \tilde{R}_{l,t} \hat{d}_{\mathcal{D},t} - E\left(\tilde{R}_{h,t} \tilde{R}_{l,t} \hat{d}_{1,t} \right) \right) \right)^{2} \right|^{1/2}.$$
(D.2)

We define the short-hand notation $\delta_{h,l,t} = X_{h,t}X_{l,t}d_{\mathcal{D},t}(\theta^0) - E(X_{h,t}X_{l,t}d_{\mathcal{D},t}(\theta^0))$; based on this, (D.2) can be rewritten as

$$\left| \widehat{g}_{1}^{(i)} - g_{1}^{(i)} \right| \leq \left| \sum_{h=1}^{N} \sum_{l=1}^{N} \left(\frac{1}{T \pi_{\mathcal{D}}} \sum_{t=1}^{T} \left(\delta_{h,l,t} + \widetilde{R}_{h,t} \widetilde{R}_{l,t} \left(\widehat{d}_{\mathcal{D},t} - d_{\mathcal{D},t} \left(\theta^{0} \right) \right) \right) \right)^{2} \right|^{1/2}.$$
 (D.3)

By repeated use of the C_r -inequality we have

$$\begin{aligned} \left| \hat{g}_{1}^{(i)} - g_{1}^{(i)} \right| &\leq c_{0} \left| \sum_{h=1}^{N} \sum_{l=1}^{N} \left(\frac{1}{T\pi_{\mathcal{D}}} \sum_{t=1}^{T} \delta_{h,l,t} \right)^{2} + \sum_{h=1}^{N} \sum_{l=1}^{N} \left(\frac{1}{T\pi_{\mathcal{D}}} \sum_{t=1}^{T} \tilde{R}_{h,t} \tilde{R}_{l,t} \left(\hat{d}_{\mathcal{D},t} - d_{\mathcal{D},t} \left(\theta^{0} \right) \right) \right)^{2} \right|^{1/2} \\ &\leq c_{1} \left| \sum_{h=1}^{N} \sum_{l=1}^{N} \left(\frac{1}{T\pi_{\mathcal{D}}} \sum_{t=1}^{T} \delta_{h,l,t} \right)^{2} \right|^{1/2} + c_{2} \left| \sum_{h=1}^{N} \sum_{l=1}^{N} \left(\frac{1}{T\pi_{\mathcal{D}}} \sum_{t=1}^{T} \tilde{R}_{h,t} \tilde{R}_{l,t} \left(\hat{d}_{\mathcal{D},t} - d_{\mathcal{D},t} \left(\theta^{0} \right) \right) \right)^{2} \right|^{1/2}. \end{aligned}$$

$$(D.4)$$

Using Lemma C.9, we have

$$E \max_{1 \le h \le N, 1 \le l \le N, 1 \le t \le T} \sum_{h'=1}^{h} \sum_{l'=1}^{l} \left(\sum_{t'=1}^{t} \delta_{h',l',t'} \right)^2 \le c_0 N^2 T;$$

XXVII

thus, by Lemma C.1 and Markov inequality, we have

$$\sum_{h=1}^{N} \sum_{l=1}^{N} \left(\frac{1}{T\pi_{\mathcal{D}}} \sum_{t=1}^{T} \delta_{h,l,t} \right)^{2} = o_{a.s.} \left(\frac{N^{2}T}{T^{2}\pi_{\mathcal{D}}^{2}} \left(\ln N \right)^{2+\epsilon} \left(\ln T \right)^{1+\epsilon} \right),$$

for every $\epsilon > 0$. Considering the second term in (D.4), convexity implies that

$$\sum_{h=1}^{N} \sum_{l=1}^{N} \left(\frac{1}{T \pi_{\mathcal{D}}} \sum_{t=1}^{T} \tilde{R}_{h,t} \tilde{R}_{l,t} \left(\hat{d}_{\mathcal{D},t} - d_{\mathcal{D},t} \left(\theta^{0} \right) \right) \right)^{2} \leq \frac{1}{T \pi_{\mathcal{D}}^{2}} \sum_{h=1}^{N} \sum_{l=1}^{N} \sum_{t=1}^{N} \tilde{R}_{h,t}^{2} \tilde{R}_{l,t}^{2} \left(\hat{d}_{\mathcal{D},t} - d_{\mathcal{D},t} \left(\theta^{0} \right) \right)^{2}.$$
(D.5)

Hence, applying (C.23) to (D.5) it follows that

$$\sum_{h=1}^{N} \sum_{l=1}^{N} \left(\frac{1}{T\pi_{\mathcal{D}}} \sum_{t=1}^{T} \widetilde{R}_{h,t} \widetilde{R}_{l,t} \left(\widehat{d}_{\mathcal{S},t} - d_{\mathcal{D},t} \left(\theta^{0} \right) \right) \right)^{2} \le c_{0} \left| \widehat{\theta} - \theta^{0} \right|^{2} \frac{1}{T\pi_{\mathcal{D}}^{2}} \sum_{h=1}^{N} \sum_{l=1}^{N} \sum_{t=1}^{N} \widetilde{R}_{h,t}^{2} \widetilde{R}_{l,t}^{2}.$$

Equation (C.24) entails

$$E\left(\frac{1}{N^2T}\sum_{h=1}^N\sum_{l=1}^N\sum_{t=1}^T\widetilde{R}_{h,t}^2\widetilde{R}_{l,t}^2\right) \le c_0.$$

The maximal inequality for rectangular sums (Moricz (1983)) now yields

$$E \max_{1 \le h \le N, 1 \le l \le N, 1 \le t \le T} \sum_{h'=1}^{h} \sum_{l'=1}^{l} \sum_{t'=1}^{t} \widetilde{R}_{h',t'}^2 \widetilde{R}_{l',t'}^2 \le c_0 N^2 T (\ln N)^2 \ln T,$$

which, by Lemma C.1 and Markov inequality, yields

$$\sum_{h=1}^{N} \sum_{l=1}^{N} \sum_{t=1}^{T} \widetilde{R}_{h,t}^{2} \widetilde{R}_{l,t}^{2} = o_{a.s.} \left(N^{2} T \left(\ln N \right)^{3+\epsilon} \left(\ln T \right)^{2+\epsilon} \right),$$

for every $\epsilon > 0$. Then, by Lemma C.7 we obtain the general result

$$\left|\sum_{h=1}^{N}\sum_{l=1}^{N}\left(\frac{1}{T\pi_{\mathcal{D}}}\sum_{t=1}^{T}\widetilde{R}_{h,t}\widetilde{R}_{l,t}\left(\widehat{d}_{1,t}-d_{\mathcal{D},t}\left(\theta^{0}\right)\right)\right)^{2}\right|^{1/2}=o_{a.s.}\left(\frac{N}{T\pi_{\mathcal{D}}}\left(\ln N\right)^{\frac{3+\epsilon}{2}}\left(\ln T\right)^{\frac{2+\epsilon}{2}}\overline{v}_{N,T}\left(\epsilon\right)\right);$$

recalling that we have assumed $\eta = 1$, the desired result follows.

Proof of Theorem A.3. The proofs are similar to those of Theorems 3 and 4 in Horváth and

XXVIII

Trapani (2019), and therefore we only report the main arguments to save space. We begin by showing (A.12). It follows from Lemma A.1 and Theorem A.2 that, for all i and $j = \mathcal{D}, \mathcal{S}$

$$P\left\{\omega: \lim_{\min(N,T)\to\infty} \left(\frac{N^{-\varrho_j}\widehat{g}_j^{(i)}}{\overline{g}_j(p)} - \varkappa N^{1-\varrho_j}\right) = \infty\right\} = 1.$$

Thus, by continuity, we can assume henceforth that

$$\lim_{\min(N,T)\to\infty} \exp\left(-\varkappa N^{1-\varrho_j}\right)\psi\left(\widehat{g}_j^{(i)}\right) = \infty.$$
(D.6)

Let $\Phi(\cdot)$ denote the standard normal distribution; it holds that

$$\begin{split} M^{-1/2} \sum_{m=1}^{M} \left(\zeta_{j,m}^{(i)}\left(s\right) - \frac{1}{2} \right) &= M^{-1/2} \sum_{m=1}^{M} \left(I\left\{ \xi_{i,m}^{(j)} \le 0 \right\} - \frac{1}{2} \right) + M^{-1/2} \sum_{m=1}^{M} \left(\Phi\left(\frac{s}{\psi\left(\hat{g}_{j}^{(i)}\right)}\right) - \frac{1}{2} \right) \\ &+ M^{-1/2} \sum_{m=1}^{M} \left[I\left\{ \xi_{i,m}^{(j)} \le \frac{s}{\psi\left(\hat{g}_{j}^{(i)}\right)} \right\} - I\left\{ \xi_{i,m}^{(j)} \le 0 \right\} - \left(\Phi\left(\frac{s}{\psi\left(\hat{g}_{j}^{(i)}\right)}\right) - \frac{1}{2} \right) \right]. \end{split}$$

By definition

$$E^* \zeta_{j,m}^{(i)}(s) = \Phi\left(\frac{s}{\psi\left(\widehat{g}_j^{(i)}\right)}\right),$$

$$V^* \zeta_{j,m}^{(i)}(s) = \Phi\left(\frac{s}{\psi\left(\widehat{g}_j^{(i)}\right)}\right) \left[1 - \Phi\left(\frac{s}{\psi\left(\widehat{g}_j^{(i)}\right)}\right)\right].$$

Thus

$$E^* \int_{-\infty}^{\infty} \left| M^{-1/2} \sum_{m=1}^{M} \left[I\left\{ \xi_{i,m}^{(j)} \le \frac{s}{\psi\left(\hat{g}_j^{(i)}\right)} \right\} - I\left\{ \xi_{i,m}^{(j)} \le 0 \right\} - \left(\Phi\left(\frac{s}{\psi\left(\hat{g}_j^{(i)}\right)}\right) - \frac{1}{2} \right) \right] \right|^2 d\Phi(s)$$
$$= \int_{-\infty}^{\infty} E^* \left| \left[I\left\{ \xi_{i,1}^{(j)} \le \frac{s}{\psi\left(\hat{g}_j^{(i)}\right)} \right\} - I\left\{ \xi_{i,1}^{(j)} \le 0 \right\} - \left(\Phi\left(\frac{s}{\psi\left(\hat{g}_j^{(i)}\right)}\right) - \frac{1}{2} \right) \right] \right|^2 d\Phi(s),$$

on account of the independence of the $\xi_{i,m}^{(j)}$ s across m. Also

$$\begin{split} E^*\left(I\left\{\xi_{i,1}^{(j)} \le \frac{s}{\psi\left(\hat{g}_{j}^{(i)}\right)}\right\} - I\left\{\xi_{i,1}^{(j)} \le 0\right\}\right) &= \Phi\left(\frac{s}{\psi\left(\hat{g}_{j}^{(i)}\right)}\right) - \frac{1}{2}\\ V^*\left(I\left\{\xi_{i,1}^{(j)} \le \frac{s}{\psi\left(\hat{g}_{j}^{(i)}\right)}\right\} - I\left\{\xi_{i,1}^{(j)} \le 0\right\}\right) &= \left(\Phi\left(\frac{s}{\psi\left(\hat{g}_{j}^{(i)}\right)}\right) - \frac{1}{2}\right)\left(\frac{3}{2} - \Phi\left(\frac{s}{\psi\left(\hat{g}_{j}^{(i)}\right)}\right)\right)\\ &\le \Phi\left(\frac{s}{\psi\left(\hat{g}_{j}^{(i)}\right)}\right) - \frac{1}{2}.\end{split}$$

Hence, we have

$$\int_{-\infty}^{\infty} E^* \left| \left[I \left\{ \xi_{i,1}^{(j)} \le \frac{s}{\psi\left(\hat{g}_j^{(i)}\right)} \right\} - I \left\{ \xi_{i,1}^{(j)} \le 0 \right\} - \left(\Phi \left(\frac{s}{\psi\left(\hat{g}_j^{(i)}\right)} \right) - \frac{1}{2} \right) \right] \right|^2 d\Phi\left(s\right) \\ \le \int_{-\infty}^{\infty} \left| \Phi \left(\frac{s}{\psi\left(\hat{g}_j^{(i)}\right)} \right) - \frac{1}{2} \right| d\Phi\left(s\right) \le \frac{1}{\sqrt{2\pi}\psi\left(\hat{g}_j^{(i)}\right)} \int_{-\infty}^{\infty} |s| \, d\Phi\left(s\right) = \frac{1}{\pi\psi\left(\hat{g}_j^{(i)}\right)},$$

which drifts to zero by (D.6). Also

$$\int_{-\infty}^{\infty} M^{-1/2} \sum_{m=1}^{M} \left(\Phi\left(\frac{s}{\psi\left(\widehat{g}_{j}^{(i)}\right)}\right) - \frac{1}{2} \right) d\Phi\left(s\right) \le \frac{M}{2\pi \left|\psi\left(\widehat{g}_{j}^{(i)}\right)\right|^{2}} \int_{-\infty}^{\infty} s^{2} d\Phi\left(s\right) = \frac{M}{2\pi \left|\psi\left(\widehat{g}_{j}^{(i)}\right)\right|^{2}} d\Phi\left(s\right) = \frac{M}{2\pi \left|\psi$$

Hence, using (A.11), we conclude via Markov's inequality that

$$\begin{split} \Upsilon_{j}^{(i)} &= \int_{-\infty}^{\infty} \left\{ \frac{2}{\sqrt{M}} \sum_{m=1}^{M} \left(I\left\{\xi_{i,m}^{(j)} \leq 0\right\} - \frac{1}{2} \right) \right\}^{2} d\Phi\left(s\right) + o_{P^{*}}(1) \\ &= \left\{ \frac{2}{\sqrt{M}} \sum_{m=1}^{M} \left(I\left\{\xi_{i,m}^{(j)} \leq 0\right\} - \frac{1}{2} \right) \right\}^{2} + o_{P^{*}}(1), \end{split}$$

and therefore the desired result follows from the Central Limit Theorem for Bernoulli random variables.

We now turn to showing (A.13). Again Lemma A.1 and Theorem A.2 entail that

$$P\left\{\omega: \lim_{\min(N,T)\to\infty}\frac{N^{-\varrho_j}\widehat{g}_j^{(i)}}{\overline{g}_j\left(p\right)} = 0\right\} = 1.$$

By continuity, this means that we can assume from now on that

$$\lim_{\min(N,T)\to\infty}\psi\left(\widehat{g}_{j}^{(i)}\right) = 1.$$
(D.7)

Consider

$$\zeta_{j,m}^{(i)}(s) - \frac{1}{2} = I\left\{\xi_{i,1}^{(j)} \le \frac{s}{\psi\left(\widehat{g}_{j}^{(i)}\right)}\right\} \pm \Phi\left(\frac{s}{\psi\left(\widehat{g}_{j}^{(i)}\right)}\right) - \frac{1}{2}.$$

Then we have

$$E^* \int_{-\infty}^{\infty} \left| M^{-1/2} \sum_{m=1}^{M} \left(I \left\{ \xi_{i,1}^{(j)} \le \frac{s}{\psi\left(\widehat{g}_j^{(i)}\right)} \right\} - \frac{1}{2} \right) \right|^2 d\Phi\left(s\right) \\ = E^* \left(I \left\{ \xi_{i,1}^{(j)} \le \frac{s}{\psi\left(\widehat{g}_j^{(i)}\right)} \right\} - \Phi\left(\frac{s}{\psi\left(\widehat{g}_j^{(i)}\right)}\right) \right)^2 + M \int_{-\infty}^{\infty} \left(\Phi\left(\frac{s}{\psi\left(\widehat{g}_j^{(i)}\right)}\right) - \frac{1}{2} \right)^2 d\Phi\left(s\right).$$

Given that

$$E^*\left(I\left\{\xi_{i,1}^{(j)} \le \frac{s}{\psi\left(\widehat{g}_j^{(i)}\right)}\right\} - \Phi\left(\frac{s}{\psi\left(\widehat{g}_j^{(i)}\right)}\right)\right)^2 < \infty,$$

by Markov's inequality it follows that

$$\int_{-\infty}^{\infty} \left[M^{-1/2} \sum_{m=1}^{M} \left(I\left\{ \xi_{i,1}^{(j)} \le \frac{s}{\psi\left(\widehat{g}_{j}^{(i)}\right)} \right\} - \Phi\left(\frac{s}{\psi\left(\widehat{g}_{j}^{(i)}\right)}\right) \right) \right]^{2} d\Phi\left(s\right) = O_{P^{*}}(1),$$

for almost all realizations of $\{e_j, b_j, -\infty < j < \infty\}$. Thus, as $\min(M, N, T_j) \to \infty$ for $j = \mathcal{D}, \mathcal{S}$

$$\frac{1}{4M}\Upsilon_{j}^{(i)} = \int_{-\infty}^{\infty} \left(\Phi(s) - \frac{1}{2}\right)^{2} d\Phi(s) + O_{P^{*}}(1).$$
(D.8)

Hence the proof of (A.13) is complete.

XXXI

Proof of Theorem 1. See the proof of (the identical) Theorem 3 in Trapani (2018). $\hfill\square$

Proof of Theorem 2. The result follows from combining (A.24) and (A.25), with

$$\hat{\gamma}_{j,\mathbf{g}} - \gamma_{j,\mathbf{g}}^{0} = O_P\left(\frac{1}{N^{1/2}}\right) + O_P\left(\frac{T^{1/2}}{T_j}\right) + O_P\left(C_{N,T}^{-2}\right) + O_P\left(T^{-1}\right).$$
(D.9)

Proof of Theorem 3. The result readily follows by combining the results in Theorem A.5 (see also Theorem 3 in Giglio and Xiu (2021)). \Box

Proof of Lemma A.2. We report the proof using the assumption that the true $d_{j,t}$ has been used (also in the computation of \hat{T}_j); extending to the case where $\hat{d}_{j,t}$ is employed is straightforward in light of the results derived above. It holds that

$$\widehat{e}_{j,i} = \alpha_{j,i} + \overline{\epsilon}_{j,i} + \left(\frac{1}{T_j}\sum_{t=1}^T d_{j,t}\right)\left(\gamma_{j,0} - \widehat{\gamma}_{j,0}\right) + \frac{1}{T_j}\sum_{t=1}^T \left(\boldsymbol{\beta}_{j,i}'\left(\boldsymbol{\gamma}_{j,1} + \mathbf{u}_{j,t}\right) - \widehat{\boldsymbol{\beta}}_{j,i}'\left(\widehat{\boldsymbol{\gamma}}_{j,1} + \widehat{\mathbf{u}}_{j,t}\right)\right) d_{j,t},$$

where

$$\overline{\epsilon}_{j,i} = \frac{1}{T_j} \sum_{t=1}^T \epsilon_{i,t} d_{j,t}.$$

Also, applying Theorem 3.4 and Corollary 3.2 in Massacci (2017) (recalling also (24)) yields

$$\frac{1}{N}\sum_{i=1}^{N} \left\| \beta_{j,i} - \widehat{\beta}_{j,i} \right\|^{2} = o_{P}(1), \qquad (D.10)$$

$$\left\|\frac{1}{T_{j}}\sum_{t=1}^{T}\left(\mathbf{u}_{j,t}-\widehat{\mathbf{u}}_{j,t}\right)\right\|^{2} \leq \frac{T}{T_{j}}\frac{1}{T}\sum_{t=1}^{T}\left\|\mathbf{u}_{j,t}-\widehat{\mathbf{u}}_{j,t}\right\|^{2} = o_{P}\left(1\right),$$
(D.11)

having omitted rotation matrices for the sake of the notation. Note now that

$$\frac{1}{N} \sum_{i=1}^{N} \hat{e}_{j,i}^{2} = \frac{1}{N} \sum_{i=1}^{N} \alpha_{j,i}^{2} \\
+ \frac{1}{N} \sum_{i=1}^{N} \left(\bar{\epsilon}_{j,i} + (\gamma_{j,0} - \hat{\gamma}_{j,0}) + \frac{1}{T_{j}} \sum_{t=1}^{T} \left(\beta'_{j,i} \left(\gamma_{j,1} + \mathbf{u}_{j,t} \right) - \hat{\beta}'_{j,i} \left(\hat{\gamma}_{j,1} + \hat{\mathbf{u}}_{j,t} \right) \right) d_{j,t} \right)^{2} \\
+ \frac{2}{N} \sum_{i=1}^{N} \alpha_{j,i} \left(\bar{\epsilon}_{j,i} + (\gamma_{j,0} - \hat{\gamma}_{j,0}) + \frac{1}{T_{j}} \sum_{t=1}^{T} \left(\beta'_{j,i} \left(\gamma_{j,1} + \mathbf{u}_{j,t} \right) - \hat{\beta}'_{j,i} \left(\hat{\gamma}_{j,1} + \hat{\mathbf{u}}_{j,t} \right) \right) d_{j,t} \right).$$

XXXII

The LLN entails

$$\frac{1}{N}\sum_{i=1}^{N}\alpha_{j,i}^{2} = \sigma_{\alpha,j}^{2} + o_{P}(1);$$

upon showing that

$$\frac{1}{N}\sum_{i=1}^{N} \left(\bar{\epsilon}_{j,i} + (\gamma_{j,0} - \hat{\gamma}_{j}) + \frac{1}{T_{j}}\sum_{t=1}^{T} \left(\beta_{j,i}' \left(\gamma_{j,1} + \mathbf{u}_{j,t} \right) - \hat{\beta}_{j,i}' \left(\hat{\gamma}_{j,1} + \hat{\mathbf{u}}_{j,t} \right) \right) d_{j,t} \right)^{2} = o_{P}\left(1 \right),$$

and using (repeatedly) the Cauchy-Schwartz inequality, the lemma follows. Now

$$\frac{1}{N}\sum_{i=1}^{N} \left(\bar{\epsilon}_{j,i} + (\gamma_{j,0} - \hat{\gamma}_{j}) + \frac{1}{T_{j}}\sum_{t=1}^{T} \left(\boldsymbol{\beta}_{j,i}'\left(\boldsymbol{\gamma}_{j,1} + \mathbf{u}_{j,t}\right) - \hat{\boldsymbol{\beta}}_{j,i}'\left(\hat{\gamma}_{j,1} + \hat{\mathbf{u}}_{j,t}\right) \right) d_{j,t} \right)^{2} \\
\leq c_{0} \left(\frac{1}{N}\sum_{i=1}^{N} \bar{\epsilon}_{j,i}^{2} + (\gamma_{j,0} - \hat{\gamma}_{j})^{2} + \frac{1}{N}\sum_{i=1}^{N} \left(\frac{1}{T_{j}}\sum_{t=1}^{T} \left(\boldsymbol{\beta}_{j,i}'\left(\boldsymbol{\gamma}_{j,1} + \mathbf{u}_{j,t}\right) - \hat{\boldsymbol{\beta}}_{j,i}'\left(\hat{\boldsymbol{\gamma}}_{j,1} + \hat{\mathbf{u}}_{j,t}\right) \right) \right)^{2} d_{j,t} \right).$$

By (A.24), $\gamma_{j,0} - \hat{\gamma}_j = o_P(1)$. Also, combining Theorem 2 and (D.10)-(D.11), it is easy to see that

$$\frac{1}{N}\sum_{i=1}^{N}\left(\frac{1}{T_{j}}\sum_{t=1}^{T}\left(\boldsymbol{\beta}_{j,i}^{\prime}\left(\boldsymbol{\gamma}_{j,1}+\mathbf{u}_{j,t}\right)-\widehat{\boldsymbol{\beta}}_{j,i}^{\prime}\left(\widehat{\boldsymbol{\gamma}}_{j,1}+\widehat{\mathbf{u}}_{j,t}\right)\right)\right)^{2}d_{j,t}=o_{P}\left(1\right)$$

Finally we have

$$\max_{1 \le i \le N} E\left(\overline{\epsilon}_{j,i}^{2}\right) = \max_{1 \le i \le N} \frac{1}{T_{j}^{2}} \sum_{t=1}^{T} \sum_{s=1}^{T} E\left(\epsilon_{i,t} \epsilon_{i,s} d_{j,t} d_{j,s}\right) \le c_{0} T_{j}^{-1},$$

using Assumption 4(ii). Hence

$$\frac{1}{N}\sum_{i=1}^{N}\overline{\epsilon}_{j,i}^{2}=o_{P}\left(1\right).$$

Putting all together, the desired result follows.

Proof of Theorem A.4. We present the full version of the proof of (A.24) only; the other two results follow from similar arguments and, where possible, we omit the details to avoid repetition. We begin by showing that the estimation error $\hat{P}_j - P_j^0$ is negligible. To this end, we assume without loss of generality that $\hat{P}_j \geq P_j^0$, and we omit dependence on θ or θ^0 whenever possible.

XXXIII

Let

$$\widehat{\mathbf{H}}_{jj}^{\mathbf{\Gamma}\widehat{P}_{j}}\widehat{\mathbf{\Gamma}}_{j}=\widehat{\mathbf{H}}_{jj}^{*\mathbf{\Gamma}}\widehat{\mathbf{\Gamma}}_{j}^{*}+\widetilde{\mathbf{H}}_{jj}^{\mathbf{\Gamma}}\widetilde{\mathbf{\Gamma}}_{j},$$

where $\widehat{\mathbf{H}}_{jj}^{*\mathbf{\Gamma}}$ is $\left(P_{j}^{0}+1\right) \times \left(P_{j}^{0}+1\right)$, $\widehat{\mathbf{\Gamma}}_{j}^{*}$ is $\left(P_{j}^{0}+1\right) \times 1$, $\widetilde{\mathbf{H}}_{jj}^{\mathbf{\Gamma}}$ is $\left(P_{j}^{0}+1\right) \times \left(\widehat{P}_{j}-P_{j}^{0}\right)$ and $\widetilde{\mathbf{\Gamma}}_{j}$ is $\left(\widehat{P}_{j}-P_{j}^{0}\right) \times 1$, defined such that $\widehat{\mathbf{H}}_{jj}^{\mathbf{\Gamma}\widehat{P}_{j}} = \left[\widehat{\mathbf{H}}_{jj}^{*\mathbf{\Gamma}}|\widetilde{\mathbf{H}}_{jj}^{\mathbf{\Gamma}}\right]$ and $\widehat{\mathbf{\Gamma}}_{j} = \left(\widehat{\mathbf{\Gamma}}_{j}^{*\prime}, \widetilde{\mathbf{\Gamma}}_{j}^{\prime}\right)^{\prime}$. We begin by showing that

$$\left\|\widehat{\mathbf{H}}_{jj}^{\widehat{\Gamma}\widehat{P}_{j}}\right\|_{F} = O_{P^{*}}\left(1\right). \tag{D.12}$$

Note that

$$\left\|\widehat{\mathbf{H}}_{jj}^{\widehat{\mathbf{P}}_{j}}\right\|_{F} \leq \left\|\frac{\mathbf{U}_{j}\left(\theta^{0}\right)\mathbf{U}_{j}\left(\theta\right)'}{T}\right\|_{F} \left\|\frac{\mathbf{B}_{j}'\left(\theta^{0}\right)\widehat{\mathbf{B}}_{j}\left(\theta\right)}{N}\right\|_{F} \left\|\widehat{\mathbf{V}}_{j}\left(\theta\right)^{-1}\right\|_{F};$$

the first term is bounded by Assumption 3(i); also note that, using the triangular inequality

$$\begin{split} \left\| \frac{\mathbf{B}_{j}'\left(\boldsymbol{\theta}^{0}\right)\widehat{\mathbf{B}}_{j}\left(\boldsymbol{\theta}\right)}{N} \right\|_{F} &\leq \left\| \frac{\mathbf{B}_{j}'\left(\boldsymbol{\theta}^{0}\right)\mathbf{B}_{j}\left(\boldsymbol{\theta}^{0}\right)}{N} \right\|_{F} + \left\| \frac{\mathbf{B}_{j}'\left(\widehat{\mathbf{B}}_{j}\left(\boldsymbol{\theta}\right) - \mathbf{B}_{j}\left(\boldsymbol{\theta}^{0}\right)\widehat{\mathbf{H}}_{jj}^{\Gamma\widehat{P}_{j}}\right)}{N} \right\|_{F} \\ &\leq \left\| \frac{\mathbf{B}_{j}'\left(\boldsymbol{\theta}^{0}\right)\mathbf{B}_{j}\left(\boldsymbol{\theta}^{0}\right)}{N} \right\|_{F} + \left\| \frac{\mathbf{B}_{j}}{N^{1/2}} \right\|_{F} \left\| \frac{\widehat{\mathbf{B}}_{j}\left(\boldsymbol{\theta}\right) - \mathbf{B}_{j}\left(\boldsymbol{\theta}^{0}\right)\widehat{\mathbf{H}}_{jj}^{\Gamma\widehat{P}_{j}}}{N^{1/2}} \right\|_{F} \end{split}$$

The first term is bounded by virtue of Assumption 3(iv), whereas the second one can be shown to be dominated by using the same arguments as in footnote 5 in Bai (2003). Indeed, Finally, note that

$$P^*\left(\left\|\widehat{\mathbf{V}}_j\left(\theta\right)^{-1}\right\|_F \ge M\right)$$

$$= P^*\left(\left\|\widehat{\mathbf{V}}_j\left(\theta\right)^{-1}\right\|_F \ge M|\widehat{P}_j = P_j^0\right) + P^*\left(\left\|\widehat{\mathbf{V}}_j\left(\theta\right)^{-1}\right\|_F \ge M|\widehat{P}_j \neq P_j^0\right)$$

$$\le P^*\left(\left\|\widehat{\mathbf{V}}_j\left(\theta\right)^{-1}\right\|_F \ge M|\widehat{P}_j = P_j^0\right) + P^*\left(\widehat{P}_j \neq P_j^0\right)$$

$$= o_{P^*}\left(1\right),$$

by Theorem 1, with $g^{\left(P_{j}^{0}\right)}\left(\widehat{\Sigma}_{j,\widetilde{\mathbf{R}}}\left(\theta\right)\right) > 0$ implied by Lemma C.10. By similar passages, it is easy to see that

$$\left\|\widehat{\mathbf{\Gamma}}_{j}\right\|_{F} = O_{P^{*}}\left(1\right). \tag{D.13}$$

XXXIV

We now show that

$$\left\|\widetilde{\mathbf{H}}_{jj}^{\Gamma}\widetilde{\mathbf{\Gamma}}_{j}\right\|_{F} = o_{P^{*}}\left(1\right),\tag{D.14}$$

for almost all realizations of the sample. Indeed, $\|\widetilde{\mathbf{H}}_{jj}^{\Gamma}\widetilde{\Gamma}_{j}\|_{F} \leq \|\widetilde{\mathbf{H}}_{jj}^{\Gamma}\|_{F} \|\widetilde{\Gamma}_{j}\|_{F}$, and $\|\widetilde{\Gamma}_{j}\|_{F}$ is bounded by (D.13). Also

$$\left\|\widetilde{\mathbf{H}}_{jj}^{\mathbf{\Gamma}}\right\|_{F} = tr\left(\widetilde{\mathbf{H}}_{jj}^{\mathbf{\Gamma}\prime}\widetilde{\mathbf{H}}_{jj}^{\mathbf{\Gamma}}\right) \leq \left(\widehat{P}_{j} - P_{j}^{0}\right)g^{(1)}\left(\widetilde{\mathbf{H}}_{jj}^{\mathbf{\Gamma}\prime}\widetilde{\mathbf{H}}_{jj}^{\mathbf{\Gamma}}\right),$$

with $\hat{P}_j - P_j^0 = o_{P^*}(1)$ by Theorem 1 and $g^{(1)}\left(\widetilde{\mathbf{H}}_{jj}^{\mathbf{\Gamma}'}\widetilde{\mathbf{H}}_{jj}^{\mathbf{\Gamma}}\right)$ bounded by (D.12). Therefore, $\widehat{\mathbf{H}}_{jj}^{\mathbf{\Gamma}\widehat{P}_j}\widehat{\mathbf{\Gamma}}_j = \widehat{\mathbf{H}}_{jj}^{*\mathbf{\Gamma}}\widehat{\mathbf{\Gamma}}_j^* + o_{P^*}(1)$. Turning to $\widehat{\mathbf{H}}_{jj}^{*\mathbf{\Gamma}}\widehat{\mathbf{\Gamma}}_j^*$, let $\widehat{\mathbf{\Gamma}}_j^{*P_j^0}$ be the estimate obtained when using P_j^0 , and consider

$$\widehat{\mathbf{H}}_{jj}^{*\boldsymbol{\Gamma}}\widehat{\boldsymbol{\Gamma}}_{j}^{*} = \left(\widehat{\mathbf{H}}_{jj}^{*\boldsymbol{\Gamma}} \pm \widehat{\mathbf{H}}_{jj}^{\boldsymbol{\Gamma}}\right) \left(\widehat{\boldsymbol{\Gamma}}_{j}^{*} \pm \widehat{\boldsymbol{\Gamma}}_{j}^{*P_{j}^{0}}\right).$$

By the same arguments as in footnote 5 in Bai (2003), it is easy to see that $\left\|\widehat{\Gamma}_{j}^{*} - \widehat{\Gamma}_{j}^{*P_{j}^{0}}\right\|_{F} = o_{P^{*}}(1);$ then it follows immediately that

$$\widehat{\mathbf{H}}_{jj}^{*\Gamma}\widehat{\mathbf{\Gamma}}_{j}^{*} = \widehat{\mathbf{H}}_{jj}^{\Gamma}\widehat{\mathbf{\Gamma}}_{j}^{*P_{j}^{0}} + o_{P^{*}}\left(1\right)$$

which, combined with (D.14), yields

$$\widehat{\mathbf{H}}_{jj}^{\Gamma}\widehat{\mathbf{\Gamma}}_{j} = \widehat{\mathbf{H}}_{jj}^{\Gamma}\widehat{\mathbf{\Gamma}}_{j}^{*P_{j}^{0}} + o_{P^{*}}\left(1\right).$$
(D.15)

We now conclude the proof of (A.24) by showing that

$$\widehat{\mathbf{H}}_{jj}^{\Gamma}\widehat{\Gamma}_{j}^{*P_{j}^{0}}=\Gamma_{j}+o_{P}\left(1\right).$$

Let $\widehat{\mathbf{X}}_{j}^{P_{j}^{0}}(\theta_{0}) = \left[\iota_{N}, \widehat{\mathbf{B}}_{j}^{P_{j}^{0}}(\theta_{0})\right]$. By the same logic as in the previous passages, and using Theorem 3.4 in Massacci (2017), we have

$$\frac{1}{T}\sum_{t=1}^{T}\widehat{\mathbf{\Gamma}}_{j,t}^{*P_{j}^{0}}\widehat{d}_{j,t} = \left(\widehat{\mathbf{X}}_{j}^{\widehat{P}_{j}'}(\widehat{\theta})\widehat{\mathbf{X}}_{j}^{\widehat{P}_{j}}(\widehat{\theta})\right)^{-1}\widehat{\mathbf{X}}_{j}^{\widehat{P}_{j}'}(\widehat{\theta})\frac{1}{T}\sum_{t=1}^{T}\mathbf{R}_{t}\widehat{d}_{j,t}$$

$$= \left(\widehat{\mathbf{X}}_{j}^{P_{j}^{0}'}(\theta_{0})\widehat{\mathbf{X}}_{j}^{P_{j}^{0}}(\theta_{0})\right)^{-1}\widehat{\mathbf{X}}_{j}^{P_{j}^{0}'}(\theta_{0})\frac{1}{T}\sum_{t=1}^{T}\mathbf{R}_{t}\widehat{d}_{j,t} + o_{P^{*}}\left(1\right) + o_{P}\left(1\right).$$

XXXV

Also

$$\left(\widehat{\mathbf{X}}_{j}^{P_{j}^{0}\prime}(\theta_{0}) \widehat{\mathbf{X}}_{j}^{P_{j}^{0}}(\theta_{0}) \right)^{-1} \widehat{\mathbf{X}}_{j}^{P_{j}^{0}\prime}(\theta_{0}) \frac{1}{T} \sum_{t=1}^{T} \mathbf{R}_{t} \widehat{d}_{j,t}$$

$$= \left(\widehat{\mathbf{X}}_{j}^{P_{j}^{0}\prime}(\theta_{0}) \widehat{\mathbf{X}}_{j}^{P_{j}^{0}}(\theta_{0}) \right)^{-1} \widehat{\mathbf{X}}_{j}^{P_{j}^{0}\prime}(\theta_{0}) \frac{1}{T} \sum_{t=1}^{T} \mathbf{R}_{t} d_{j,t} \left(\theta^{0} \right)$$

$$+ \left(\widehat{\mathbf{X}}_{j}^{P_{j}^{0}\prime}(\theta_{0}) \widehat{\mathbf{X}}_{j}^{P_{j}^{0}}(\theta_{0}) \right)^{-1} \widehat{\mathbf{X}}_{j}^{P_{j}^{0}\prime}(\theta_{0}) \frac{1}{T} \sum_{t=1}^{T} \mathbf{R}_{t} \left(\widehat{d}_{j,t} - d_{j,t} \left(\theta^{0} \right) \right)$$

$$= I + II.$$

$$(D.16)$$

Noting that

$$\frac{1}{T}\sum_{t=1}^{T}\mathbf{R}_{t}\left(\hat{d}_{j,t}-d_{j,t}\left(\theta^{0}\right)\right) \leq \left(\frac{1}{T}\sum_{t=1}^{T}\|\mathbf{R}_{t}\|^{2}\right)^{1/2} \left(\frac{1}{T}\sum_{t=1}^{T}\left(\hat{d}_{j,t}-d_{j,t}\left(\theta^{0}\right)\right)^{2}\right)^{1/2},$$

and using Theorem 3.4 in Massacci (2017), it follows that

$$\frac{1}{T}\sum_{t=1}^{T}\mathbf{R}_{t}\left(\widehat{d}_{j,t}-d_{j,t}\left(\theta^{0}\right)\right)=O_{P}\left(T^{-1}\right),$$

so that, in equation (D.16), we have $II = O_P(T^{-1})$. Also, by Corollary 3.1(a) in Massacci (2017), $\left\|\widehat{\mathbf{B}}_{j}^{P_{j}^{0}} - \mathbf{B}_{j}^{0}\widehat{\mathbf{H}}_{jj}\right\|^{2} = O_P(NC_{N,T}^{-2})$, which in turn implies

$$\widehat{\mathbf{X}}_{j}^{P_{j}^{0}}(\theta_{0}) - (\boldsymbol{\iota}_{N}, \mathbf{B}_{j}^{0}\widehat{\mathbf{H}}_{jj}) = \widehat{\mathbf{X}}_{j}^{P_{j}^{0}}(\theta_{0}) - \mathbf{X}_{j}^{0}\widehat{\mathbf{H}}_{jj}^{\boldsymbol{\Gamma}} = O_{P}(NC_{N,T}^{-2});$$

thus

$$\frac{1}{T}\sum_{t=1}^{T}\widehat{\mathbf{\Gamma}}_{j,t}^{*P_{j}^{0}}\widehat{d}_{j,t} = \left(\frac{\widehat{\mathbf{H}}_{jj}^{\Gamma'}\mathbf{X}_{j}^{0'}\mathbf{X}_{j}^{0}\widehat{\mathbf{H}}_{jj}^{\Gamma}}{N}\right)^{-1} \left(\frac{\widehat{\mathbf{H}}_{jj}^{\Gamma'}\mathbf{X}_{j}^{0'}}{N}\right) \left(\frac{1}{T}\sum_{t=1}^{T}\mathbf{R}_{t}d_{j,t}\left(\theta^{0}\right)\right) + O_{P}(C_{N,T}^{-2}) + O_{P}\left(T^{-1}\right) + o_{P^{*}}\left(1\right).$$

XXXVI

Let $\alpha = (\alpha_1, ..., \alpha_N)'$. Recalling (4), it follows that

$$\frac{1}{T} \sum_{t=1}^{T} \widehat{\Gamma}_{j,t}^{*P_{j}^{0}} \widehat{d}_{j,t} - \frac{T_{j}}{T} \left(\widehat{\mathbf{H}}_{jj}^{\Gamma} \right)^{-1} \Gamma_{j}^{0}$$

$$= \frac{T_{j}}{T} \left(\frac{\widehat{\mathbf{H}}_{jj}^{\Gamma'} \mathbf{X}_{j}^{0'} \mathbf{X}_{j}^{0} \widehat{\mathbf{H}}_{jj}^{\Gamma}}{N} \right)^{-1} \left(\frac{\widehat{\mathbf{H}}_{jj}^{\Gamma'} \mathbf{X}_{j}^{0'} \alpha_{j}}{N} \right)$$

$$+ \left(\frac{\widehat{\mathbf{H}}_{jj}^{\Gamma'} \mathbf{X}_{j}^{0'} \mathbf{X}_{j}^{0} \widehat{\mathbf{H}}_{jj}^{\Gamma}}{N} \right)^{-1} \left(\frac{\widehat{\mathbf{H}}_{jj}^{\Gamma'} \mathbf{X}_{j}^{0'}}{N} \right) \left(\frac{1}{T} \sum_{t=1}^{T} \mathbf{B}_{j}^{0} \mathbf{u}_{j,t}^{0} d_{j,t} \left(\theta^{0} \right) \right)$$

$$+ \left(\frac{\widehat{\mathbf{H}}_{jj}^{\Gamma'} \mathbf{X}_{j}^{0'} \mathbf{X}_{j}^{0} \widehat{\mathbf{H}}_{jj}^{\Gamma}}{N} \right)^{-1} \left(\frac{\widehat{\mathbf{H}}_{jj}^{\Gamma'} \mathbf{X}_{j}^{0'}}{N} \right) \left(\frac{1}{T} \sum_{t=1}^{T} \mathbf{B}_{j}^{0} \mathbf{u}_{j,t}^{0} d_{j,t} \left(\theta^{0} \right) \right)$$

$$+ O_{P} \left(C_{N,T}^{-2} \right) + O_{P} \left(T^{-1} \right) + o_{P*} \left(1 \right)$$

$$= I_{a} + I_{b} + I_{c} + O_{P} \left(C_{N,T}^{-2} \right) + O_{P} \left(T^{-1} \right) + o_{P*} \left(1 \right)$$

Assumptions 3(iv) and 7(i) immediately yields $I_a = O_P\left(\frac{T_j}{N^{1/2}T}\right)$. Consider I_c ; it holds that

$$E\left(\frac{1}{NT}\sum_{t=1}^{T}\mathbf{X}_{j}^{0'}\epsilon_{t}d_{j,t}\left(\theta^{0}\right)\right)^{2}$$

$$=\frac{1}{(NT)^{2}}\sum_{i,k=1}^{N}\sum_{t,s=1}^{T}X_{i,j}^{0}X_{k,j}^{0'}E\left(\epsilon_{i,t}\epsilon_{j,s}d_{j,t}\left(\theta^{0}\right)d_{j,s}\left(\theta^{0}\right)\right)$$

$$\leq\frac{1}{(NT)^{2}}\max_{1\leq i\leq N}\left\|X_{i,j}^{0}\right\|^{2}\sum_{i,k=1}^{N}\sum_{t,s=1}^{T}\left|E\left(\epsilon_{i,t}\epsilon_{j,s}d_{j,t}\left(\theta^{0}\right)d_{j,s}\left(\theta^{0}\right)\right)\right|$$

$$\leq c_{0}\frac{N}{(NT)^{2}}\max_{1\leq i\leq N}\sum_{k=1}^{N}\sum_{t,s=1}^{T}\left|E\left(\epsilon_{i,t}\epsilon_{j,s}d_{j,t}\left(\theta^{0}\right)d_{j,s}\left(\theta^{0}\right)\right)\right| \leq \frac{c_{0}}{NT},$$

by Assumptions 3(iii) and 2(iv); so, putting all together and using Assumption 3(iv), it follows that $I_c = O_P\left(\frac{1}{\sqrt{NT}}\right)$. Turning to I_b , (C.29) yields

$$\left(\frac{\widehat{\mathbf{H}}_{jj}^{\Gamma\prime}\mathbf{X}_{j}^{0\prime}\mathbf{X}_{j}^{0}\widehat{\mathbf{H}}_{jj}^{\Gamma}}{N}\right)^{-1} \left(\frac{\widehat{\mathbf{H}}_{jj}^{\Gamma\prime}\mathbf{X}_{j}^{0\prime}\mathbf{B}_{j}^{0}}{N}\right) \left(\frac{1}{T}\sum_{t=1}^{T}\mathbf{u}_{j,t}^{0}d_{j,t}\left(\theta^{0}\right)\right) = O_{P}\left(\frac{1}{\sqrt{T}}\right)$$
(D.17)

XXXVII

Putting all together, it follows that

$$\frac{1}{T_j} \sum_{t=1}^T \widehat{\Gamma}_{j,t}^{*P_j^0} \widehat{d}_{j,t} - \left(\widehat{\mathbf{H}}_{jj}^{\mathbf{\Gamma}}\right)^{-1} \Gamma_j^0 = O_P\left(\frac{1}{N^{1/2}}\right) + O_P\left(\frac{T^{1/2}}{T_j}\right) + O_P\left(\frac{T}{T_jN}\right) + O_P\left(T^{-1}\right) + o_{P^*}\left(1\right);$$
(D.18)

the desired result now follows immediately. We now turn to showing (A.25). We assume that P_j^0 is known; the estimation error $\hat{P}_j - P_j^0$ can be shown to be negligible on account of the same arguments as above. We have

$$\widehat{\Lambda}_j - \Lambda_j^0 \widehat{\mathbf{H}}_{jj} = \left(\sum_{t=1}^T \widetilde{\mathbf{g}}_t \widehat{\mathbf{u}}_{j,t}^{P_j^0 \prime} \widehat{d}_{j,t}\right) \left(\sum_{t=1}^T \widehat{\mathbf{u}}_{j,t}^{P_j^0} \widehat{\mathbf{u}}_{j,t}^{P_j^0 \prime} \widehat{d}_{j,t}\right)^{-1}.$$

Note that, by Corollary 3.2 in Massacci (2017), it is easy to see that

$$\frac{1}{T}\sum_{t=1}^{T}\widehat{\mathbf{u}}_{j,t}^{P_{j}^{0}}\widehat{\mathbf{u}}_{j,t}^{P_{j}^{0}'}\widehat{d}_{j,t} = \frac{1}{T}\sum_{t=1}^{T}\widehat{\mathbf{H}}_{jj}^{-1}\mathbf{u}_{j,t}^{P_{j}^{0}} \mathbf{u}_{j,t}^{P_{j}^{0}'}\left(\widehat{\mathbf{H}}_{jj}'\right)^{-1}d_{j,t}\left(\theta^{0}\right) + o_{P}\left(1\right).$$
(D.19)

Also

$$\frac{1}{T}\sum_{t=1}^{T}\tilde{\mathbf{g}}_{t}\hat{\mathbf{u}}_{j,t}^{P_{j}^{0}\prime}\hat{d}_{j,t} = \frac{1}{T}\sum_{t=1}^{T}\tilde{\mathbf{g}}_{t}\hat{\mathbf{u}}_{j,t}^{P_{j}^{0}\prime}d_{j,t}\left(\theta^{0}\right) + \frac{1}{T}\sum_{t=1}^{T}\tilde{\mathbf{g}}_{t}\hat{\mathbf{u}}_{j,t}^{P_{j}^{0}\prime}\left(\hat{d}_{j,t} - d_{j,t}\left(\theta^{0}\right)\right) = I + II.$$
(D.20)

Consider II; we have

$$II \leq \left(\frac{1}{T}\sum_{t=1}^{T} \|\widetilde{\mathbf{g}}_{t}\|^{4}\right)^{1/4} \left(\frac{1}{T}\sum_{t=1}^{T} \left\|\widehat{\mathbf{u}}_{j,t}^{P_{j}^{0}}\right\|^{4}\right)^{1/4} \left(\frac{1}{T}\sum_{t=1}^{T} \left(\widehat{d}_{j,t} - d_{j,t}\left(\theta^{0}\right)\right)^{2}\right)^{1/2};$$
(D.21)

using Assumptions 3(i) and 8(i), and using Theorem 3.4 in Massacci (2017), it follows that $II = O_P(N^{-\eta}T^{-1})$. Turning to I, by the definition of $\tilde{\mathbf{g}}_t(\hat{\theta})$ it holds that

$$I = \frac{1}{T} \sum_{t=1}^{T} \left(\mathbf{g}_{t} - \mathbf{a} \right) \widehat{\mathbf{u}}_{j,t}^{P_{j}^{0}} d_{j,t} \left(\theta^{0} \right) + \frac{1}{T} \sum_{t=1}^{T} \left(\mathbf{g}_{t} - \mathbf{a} \right) \frac{1}{T} \sum_{t=1}^{T} \widehat{\mathbf{u}}_{j,t}^{P_{j}^{0}} d_{j,t} \left(\theta^{0} \right) = I_{a} + I_{b}.$$

XXXVIII

Using (C.29) and Assumption 8(*ii*), it follows that $I_b = O_P(T^{-1})$.

$$I_{a} = \frac{1}{T} \sum_{t=1}^{T} \Lambda_{j}^{0} \mathbf{u}_{j,t}^{0} \widehat{\mathbf{u}}_{j,t}^{P_{j}^{0}} d_{j,t} \left(\theta^{0}\right) + \frac{1}{T} \sum_{t=1}^{T} \mathbf{e}_{t} \widehat{\mathbf{u}}_{j,t}^{P_{j}^{0}} d_{j,t} \left(\theta^{0}\right) = I_{a,1} + I_{a,2}.$$

Now

$$\begin{split} I_{a,1} &= \Lambda_{j}^{0} \widehat{\mathbf{H}}_{jj} \widehat{\mathbf{H}}_{jj}^{-1} \left(\frac{1}{T} \sum_{t=1}^{T} \mathbf{u}_{j,t}^{0} \mathbf{u}_{j,t}^{0\prime} \right) \left(\widehat{\mathbf{H}}_{jj}^{\prime} \right)^{-1} d_{j,t} \left(\theta^{0} \right) \\ &+ \Lambda_{j}^{0} \frac{1}{T} \sum_{t=1}^{T} \mathbf{u}_{j,t}^{0} \left(\widehat{\mathbf{u}}_{j,t}^{P_{j}^{0}} - \widehat{\mathbf{H}}_{jj}^{-1} \mathbf{u}_{j,t}^{0} \right)^{\prime} d_{j,t} \left(\theta^{0} \right) \\ &= I_{a,1,1} + I_{a,1,2}. \end{split}$$

Following the same passages as in the proof of Lemma B.2 in Bai (2003), it can be shown that

$$\frac{1}{T}\sum_{t=1}^{T}\mathbf{u}_{j,t}^{0}\left(\widehat{\mathbf{u}}_{j,t}^{P_{j}^{0}}-\widehat{\mathbf{H}}_{jj}^{-1}\mathbf{u}_{j,t}^{0}\right)'d_{j,t}\left(\theta^{0}\right)=O_{P}\left(C_{N,T}^{-2}\right),\tag{D.22}$$

so that $I_{a,1,2} = O_P\left(C_{N,T}^{-2}\right)$. Also

$$I_{a,2} = \left(\frac{1}{T}\sum_{t=1}^{T}\mathbf{e}_{t}\mathbf{u}_{j,t}^{0'}\right)\left(\widehat{\mathbf{H}}_{jj}'\right)^{-1}d_{j,t}\left(\theta^{0}\right) + \frac{1}{T}\sum_{t=1}^{T}\mathbf{e}_{t}\left(\widehat{\mathbf{u}}_{j,t}^{P_{j}^{0}} - \widehat{\mathbf{H}}_{jj}^{-1}\mathbf{u}_{j,t}^{0}\right)'d_{j,t}\left(\theta^{0}\right) = I_{a,2,1} + I_{a,2,2}.$$

Assumption 8(*ii*) entails that $I_{a,2,1} = O_P(T^{-1/2})$. Finally, by the same token as for (D.22), $I_{a,2,2} = O_P(C_{N,T}^{-2})$, whence $I_{a,2} = O_P(T^{-1/2}) + O_P(C_{N,T}^{-2})$. Putting all together, we obtain

$$\widehat{\Lambda}_j - \Lambda_j^0 \widehat{\mathbf{H}}_{jj} = O_P\left(\frac{1}{T^{1/2}}\right) + O_P\left(C_{N,T}^{-2}\right) + O_P\left(T^{-1}\right).$$
(D.23)

Proof of Theorem A.5. Consider (A.26). Upon inspecting (D.18), under (24) and (25), and using Lemma C.8, it is immediate to see that the limiting behaviour of $T^{1/2} \left(\widehat{\Gamma}_{j}^{*P_{j}^{0}} - \left(\widehat{\mathbf{H}}_{jj}^{\Gamma} \right)^{-1} \Gamma_{j}^{0} \right)$ is

XXXIX

driven by

$$\left(\frac{\widehat{\mathbf{H}}_{jj}^{\boldsymbol{\Gamma}\prime}\mathbf{X}_{j}^{0\prime}\mathbf{X}_{j}^{0}\widehat{\mathbf{H}}_{jj}^{\boldsymbol{\Gamma}}}{N}\right)^{-1}\left[\left(\frac{\widehat{\mathbf{H}}_{jj}^{\boldsymbol{\Gamma}\prime}\mathbf{X}_{j}^{0\prime}\mathbf{B}_{j}^{0}}{N}\right)\left(\frac{T}{T_{j}}\frac{1}{T^{1/2}}\sum_{t=1}^{T}\mathbf{u}_{j,t}^{0}d_{j,t}\left(\theta^{0}\right)\right)+\frac{T_{j}}{T}\left(\frac{\widehat{\mathbf{H}}_{jj}^{\boldsymbol{\Gamma}\prime}\mathbf{X}_{j}^{0\prime}\alpha_{j}}{N}\right)\right].$$

The desired result now follows immediately from (C.30) and Assumption 7. Turning to (A.27), combining (D.23) and (25), it follows that the limiting law of $T^{1/2}\left(\widehat{\Lambda}_j - \Lambda_j^0 \widehat{\mathbf{H}}_{jj}\right)$ is driven by

$$\left(\frac{1}{T^{1/2}}\sum_{t=1}^{T}\mathbf{e}_{t}\mathbf{u}_{j,t}^{0\prime}\left(\widehat{\mathbf{H}}_{jj}\right)^{-1}d_{j,t}\left(\theta^{0}\right)\right)\left(\frac{1}{T}\sum_{t=1}^{T}\left(\widehat{\mathbf{H}}_{jj}\right)^{-1}\mathbf{u}_{j,t}^{0}\mathbf{u}_{j,t}^{0\prime}\left(\widehat{\mathbf{H}}_{jj}^{\prime}\right)^{-1}d_{j,t}\left(\theta^{0}\right)\right)^{-1} = \left(\frac{1}{T^{1/2}}\sum_{t=1}^{T}\mathbf{e}_{t}\mathbf{u}_{j,t}^{0\prime}d_{j,t}\left(\theta^{0}\right)\right)\left(\frac{1}{T}\sum_{t=1}^{T}\mathbf{u}_{j,t}^{0}\mathbf{u}_{j,t}^{0\prime}d_{j,t}\left(\theta^{0}\right)\right)^{-1}\widehat{\mathbf{H}}_{jj},$$

and using Assumption 9(ii) the desired result follows immediately.

Proof of Theorem A.6. The proof makes appeal to several arguments which have already been used in the paper and therefore, for the sake of a concise discussion, we omit passages when this does not give rise to ambiguity.

We begin by noting that, by the Frisch-Waugh-Lovell theorem, it holds that

$$\widehat{\gamma}_{j,0} = \left(i'_N \widehat{\mathbf{M}}_{B,j} i_N\right)^{-1} \left(i'_N \widehat{\mathbf{M}}_{B,j} \overline{\mathbf{R}}_j \left(\widehat{\theta}\right)\right),\,$$

having defined

$$\widehat{\mathbf{M}}_{B,j} = \widehat{\mathbf{B}}_j \left(\widehat{\mathbf{B}}'_j \widehat{\mathbf{B}}_j \right)^{-1} \widehat{\mathbf{B}}'_j,$$

for $j = \mathcal{D}, \mathcal{S}$ - note that we use $\widehat{\mathbf{B}}_j$ in lieu of $\widehat{\mathbf{B}}_j^{P_j}$ to make the notation less burdensome. Thus, on account of equation (5), it holds that

$$\widehat{\gamma}_{j,0} = \gamma_{j,0} + \left(i'_N \widehat{\mathbf{M}}_{B,j} i_N\right)^{-1} \left(i'_N \widehat{\mathbf{M}}_{B,j} \alpha_j\right) + \left(i'_N \widehat{\mathbf{M}}_{B,j} i_N\right)^{-1} \left(i'_N \widehat{\mathbf{M}}_{B,j} \frac{1}{T_j} \sum_{t=1}^T \epsilon_t \widehat{d}_{j,t}\right).$$
(D.24)

We begin by noting that, using Corollary 3.1(a) in Massacci (2017), it is easy to see that

$$\left\|\widehat{\mathbf{M}}_{B,j}-\mathbf{M}_{B,j}^{0}\right\|^{2}=O_{P}\left(NC_{N,T}^{-2}\right),$$

so that

$$\frac{i'_{N}\widehat{\mathbf{M}}_{B,j}i_{N}}{N} = \frac{i'_{N}\mathbf{M}_{B,j}^{0}i_{N}}{N} + O_{P}\left(C_{N,T}^{-2}\right) = O_{P}\left(1\right).$$
(D.25)

Also, it holds that

$$E\left(\widehat{\mathbf{B}}_{j}-\mathbf{B}_{j}^{0}\widehat{\mathbf{H}}_{jj}\right)'\alpha_{j}\alpha_{j}'\left(\widehat{\mathbf{B}}_{j}-\mathbf{B}_{j}^{0}\widehat{\mathbf{H}}_{jj}\right)=\sigma_{\alpha,j}^{2}E\left(\widehat{\mathbf{B}}_{j}-\mathbf{B}_{j}^{0}\widehat{\mathbf{H}}_{jj}\right)'\left(\widehat{\mathbf{B}}_{j}-\mathbf{B}_{j}^{0}\widehat{\mathbf{H}}_{jj}\right)=O\left(NC_{N,T}^{-2}\right),$$

again by Corollary 3.1(a) in Massacci (2017); this yields the (non sharp) bound

$$i'_{N}\left(\widehat{\mathbf{M}}_{B,j}-\mathbf{M}_{B,j}^{0}\right)\alpha_{j}=O_{P}\left(N^{1/2}C_{N,T}^{-1}\right).$$
(D.26)

Finally, note that Assumption 7(i) entails that

$$N^{1/2} \left(i'_N \mathbf{M}^0_{B,j} i_N \right)^{-1} \left(i'_N \mathbf{M}^0_{B,j} \alpha_j \right) = O_P \left(1 \right).$$
 (D.27)

Putting (D.25)-(D.27) together, we obtain

$$N^{1/2} \left(i'_{N} \widehat{\mathbf{M}}_{B,j} i_{N} \right)^{-1} \left(i'_{N} \widehat{\mathbf{M}}_{B,j} \alpha_{j} \right) = N^{1/2} \left(i'_{N} \mathbf{M}_{B,j}^{0} i_{N} \right)^{-1} \left(i'_{N} \mathbf{M}_{B,j}^{0} \alpha_{j} \right) + o_{P} \left(1 \right).$$
(D.28)

Consider now

$$i'_{N}\widehat{\mathbf{M}}_{B,j}\frac{1}{T_{j}}\sum_{t=1}^{T}\epsilon_{t}\widehat{d}_{j,t} = i'_{N}\frac{1}{T_{j}}\sum_{t=1}^{T}\epsilon_{t}\widehat{d}_{j,t} + i'_{N}\widehat{\mathbf{B}}_{j}\left(\widehat{\mathbf{B}}'_{j}\widehat{\mathbf{B}}_{j}\right)^{-1}\widehat{\mathbf{B}}'_{j}\frac{1}{T_{j}}\sum_{t=1}^{T}\epsilon_{t}\widehat{d}_{j,t} = I + II.$$
(D.29)

Using Assumption 2(ii) and equation (C.23), it is easy to see that

$$E\left(i'_{N}\frac{1}{T_{j}}\sum_{t=1}^{T}\epsilon_{t}\widehat{d}_{j,t}\right)^{2} = T_{j}^{-2}E\left(\sum_{i,k=1}^{N}\sum_{t,s=1}^{T}\epsilon_{i,t}\epsilon_{k,s}\widehat{d}_{j,t}\widehat{d}_{j,s}\right)$$
$$\leq NT_{j}^{-2}\max_{1\leq k\leq N}\sum_{i=1}^{N}\sum_{t,s=1}^{T}\left|E\left(\epsilon_{i,t}\epsilon_{k,s}\widehat{d}_{j,t}\widehat{d}_{j,s}\right)\right|\leq c_{0}\frac{N}{T_{j}},$$

so that

$$i'_{N} \frac{1}{T_{j}} \sum_{t=1}^{T} \epsilon_{t} \hat{d}_{j,t} = O_{P} \left(N^{1/2} T_{j}^{-1/2} \right).$$
(D.30)

In order to study II in (D.29), note that

$$i'_{N}\widehat{\mathbf{B}}_{j}\left(\widehat{\mathbf{B}}'_{j}\widehat{\mathbf{B}}_{j}\right)^{-1}\widehat{\mathbf{B}}'_{j}\frac{1}{T_{j}}\sum_{t=1}^{T}\epsilon_{t}\widehat{d}_{j,t}$$
(D.31)
$$=i'_{N}\mathbf{B}_{j}^{0}\widehat{\mathbf{H}}_{jj}\left(\widehat{\mathbf{B}}'_{j}\widehat{\mathbf{B}}_{j}\right)^{-1}\widehat{\mathbf{H}}'_{jj}\mathbf{B}_{j}^{0\prime}\frac{1}{T_{j}}\sum_{t=1}^{T}\epsilon_{t}\widehat{d}_{j,t}+i'_{N}\left(\widehat{\mathbf{B}}_{j}-\mathbf{B}_{j}^{0}\widehat{\mathbf{H}}_{jj}\right)\left(\widehat{\mathbf{B}}'_{j}\widehat{\mathbf{B}}_{j}\right)^{-1}\widehat{\mathbf{H}}'_{jj}\mathbf{B}_{j}^{0\prime}\frac{1}{T_{j}}\sum_{t=1}^{T}\epsilon_{t}\widehat{d}_{j,t}$$
$$+i'_{N}\mathbf{B}_{j}^{0}\widehat{\mathbf{H}}_{jj}\left(\widehat{\mathbf{B}}'_{j}\widehat{\mathbf{B}}_{j}\right)^{-1}\left(\widehat{\mathbf{B}}_{j}-\mathbf{B}_{j}^{0}\widehat{\mathbf{H}}_{jj}\right)'\frac{1}{T_{j}}\sum_{t=1}^{T}\epsilon_{t}\widehat{d}_{j,t}$$
$$+i'_{N}\left(\widehat{\mathbf{B}}_{j}-\mathbf{B}_{j}^{0}\widehat{\mathbf{H}}_{jj}\right)\left(\widehat{\mathbf{B}}'_{j}\widehat{\mathbf{B}}_{j}\right)^{-1}\left(\widehat{\mathbf{B}}_{j}-\mathbf{B}_{j}^{0}\widehat{\mathbf{H}}_{jj}\right)'\frac{1}{T_{j}}\sum_{t=1}^{T}\epsilon_{t}\widehat{d}_{j,t}$$
$$=I+II+III+IV.$$

By construction, $(\hat{\mathbf{B}}'_{j}\hat{\mathbf{B}}_{j})^{-1} = O_{p}(N^{-1})$, and $i'_{N}\mathbf{B}^{0}_{j} = O_{P}(N)$. Using Assumption 2(*ii*) and equation (C.23), similarly to (D.30), it also follows that

$$\mathbf{B}_{j}^{0'}T_{j}^{-1}\sum_{t=1}^{T}\epsilon_{t}\widehat{d}_{j,t} = O_{P}\left(N^{1/2}T_{j}^{-1/2}\right),\tag{D.32}$$

so that $I = O_P\left(N^{-1/2}T_j^{-1/2}\right)$. Consider now

$$\mathbf{B}_{j}^{0} - \widehat{\mathbf{B}}_{j}\widehat{\mathbf{H}}_{jj}^{-1}
= T^{-1}\mathbf{B}_{j}^{0}\widehat{\mathbf{H}}_{jj}\left(\widehat{\mathbf{H}}_{jj}^{-1}\mathbf{U}_{j,t}^{0} - \widehat{\mathbf{U}}_{j,t}\right)\widehat{\mathbf{U}}_{j,t}^{\prime}\widehat{\mathbf{H}}_{jj}^{-1} + T^{-1}\epsilon_{j}\left(\widehat{\mathbf{U}}_{j,t}^{\prime} - \mathbf{U}_{j,t}^{0\prime}\left(\widehat{\mathbf{H}}_{jj}^{-1}\right)^{\prime}\right)\widehat{\mathbf{H}}_{jj}^{-1}
+ T^{-1}\epsilon_{j}\mathbf{U}_{j,t}^{0\prime}\left(\widehat{\mathbf{H}}_{jj}^{-1}\right)^{\prime}\widehat{\mathbf{H}}_{jj}^{-1},$$
(D.33)

where ϵ_j is an $N \times T$ matrix with columns $\epsilon_t \widehat{d}_{j,t}$. Using exactly the same arguments as in the proofs of Lemmas B.1 and B.3 in Bai (2003), it holds that $\left\|T^{-1}\left(\widehat{\mathbf{H}}_{jj}^{-1}\mathbf{U}_{j,t}^0 - \widehat{\mathbf{U}}_{j,t}\right)\widehat{\mathbf{U}}_{j,t}'\widehat{\mathbf{H}}\right\| = O_P\left(C_{N,T}^{-2}\right)$ and $\left\|T^{-1}\epsilon_j\left(\widehat{\mathbf{U}}_{j,t}' - \mathbf{U}_{j,t}^{0\prime}\left(\widehat{\mathbf{H}}_{jj}^{-1}\right)'\right)\widehat{\mathbf{H}}\right\| = O_P\left(C_{N,T}^{-2}\right)$. Hence, it follows that

$$i_N' \mathbf{B}_j^0 \left(T^{-1} \widehat{\mathbf{H}}_{jj} \left(\widehat{\mathbf{H}}_{jj}^{-1} \mathbf{U}_{j,t}^0 - \widehat{\mathbf{U}}_{j,t} \right) \widehat{\mathbf{U}}_{j,t}' \right) \widehat{\mathbf{H}}_{jj}^{-1} = O_P \left(N C_{N,T}^{-2} \right).$$
(D.34)

Also

$$i'_{N}T^{-1}\epsilon_{j}\left(\widehat{\mathbf{U}}'_{j,t}-\mathbf{U}^{0'}_{j,t}\left(\widehat{\mathbf{H}}^{-1}_{jj}\right)'\right) = \sum_{i=1}^{N}T^{-1}\sum_{t=1}^{T}\epsilon_{j,i,t}\widehat{d}_{j,t}\left(\widehat{\mathbf{u}}'_{j,t}-\mathbf{u}^{0'}_{j,t}\left(\widehat{\mathbf{H}}^{-1}_{jj}\right)'\right) \\ \leq \sum_{i=1}^{N}\left\|T^{-1}\sum_{t=1}^{T}\epsilon_{j,i,t}\widehat{d}_{j,t}\left(\widehat{\mathbf{u}}'_{j,t}-\mathbf{u}^{0'}_{j,t}\left(\widehat{\mathbf{H}}^{-1}_{jj}\right)'\right)\right\|;$$

again by using the same arguments as in the proof of Lemma B.1 in Bai (2003), it can be shown that

$$E\left\|T^{-1}\sum_{t=1}^{T}\epsilon_{j,i,t}\widehat{d}_{j,t}\left(\widehat{\mathbf{u}}_{j,t}'-\mathbf{u}_{j,t}^{0\prime}\left(\widehat{\mathbf{H}}_{jj}^{-1}\right)'\right)\right\| \leq c_0 C_{N,T}^{-2},$$

which yields that

$$i'_{N}T^{-1}\epsilon_{j}\left(\widehat{\mathbf{U}}'_{j,t}-\mathbf{U}^{0\prime}_{j,t}\left(\widehat{\mathbf{H}}^{-1}_{jj}\right)'\right)=O_{P}\left(NC^{-2}_{N,T}\right).$$
(D.35)

Finally, note that

$$E \left\| i'_{N} T^{-1} \epsilon_{j} \mathbf{U}_{j,t}^{0'} \right\|^{2} = \sum_{i=1}^{N} \sum_{j=1}^{N} T^{-2} \sum_{t=1}^{T} \sum_{s=1}^{T} E\left(\epsilon_{i,t} \epsilon_{j,s} \hat{d}_{j,t} \hat{d}_{j,s}\right) E \left\| \mathbf{u}_{j,t}^{0'} \mathbf{u}_{j,s}^{0} \right\|$$
$$\leq \left(\max_{1 \leq t \leq T} E \left\| \mathbf{u}_{j,t}^{0} \right\|^{2} \right) \frac{N}{T} \max_{1 \leq i \leq N} \sum_{k=1}^{N} T^{-1} \sum_{t=1}^{T} \sum_{s=1}^{T} \left| E\left(\epsilon_{i,t} \epsilon_{k,s} \hat{d}_{j,t} \hat{d}_{j,s}\right) \right| \leq c_{0} \frac{N}{T},$$

after Assumptions 2(ii) and 3(i), and using (C.23), which entails that

$$i'_N T^{-1} \epsilon_j \mathbf{U}_{j,t}^{0\prime} = O_P \left(N^{1/2} T^{-1/2} \right).$$
 (D.36)

Putting (D.34)-(D.36) together, it follows that

$$i'_{N}\left(\widehat{\mathbf{B}}_{j}-\mathbf{B}_{j}^{0}\widehat{\mathbf{H}}_{jj}\right)=O_{P}\left(NC_{N,T}^{-2}\right).$$
(D.37)

Thus, considering (D.31), (D.32) and (D.37) immediately entail that $II = O_P \left(N^{1/2} T_j^{-1/2} C_{N,T}^{-2} \right)$. Turning to *III*, note that, using (D.33)

$$\begin{aligned} \left(\widehat{\mathbf{B}}_{j} - \mathbf{B}_{j}^{0}\widehat{\mathbf{H}}_{jj}\right)' \left(\frac{1}{T_{j}}\sum_{t=1}^{T}\epsilon_{t}\widehat{d}_{j,t}\right) \\ &= \left(\mathbf{B}_{j}^{0}\widehat{\mathbf{H}}_{jj}T^{-1}\left(\widehat{\mathbf{H}}_{jj}^{-1}\mathbf{U}_{j,t}^{0} - \widehat{\mathbf{U}}_{j,t}\right)\widehat{\mathbf{U}}_{j,t}'\widehat{\mathbf{H}}_{jj}^{-1}\right)' \left(\frac{1}{T_{j}}\sum_{t=1}^{T}\epsilon_{t}\widehat{d}_{j,t}\right) \\ &+ \left(T^{-1}\epsilon_{j}\left(\widehat{\mathbf{U}}_{j,t}' - \mathbf{U}_{j,t}^{0\prime}\left(\widehat{\mathbf{H}}_{jj}^{-1}\right)'\right)\widehat{\mathbf{H}}_{jj}^{-1}\right)' \left(\frac{1}{T_{j}}\sum_{t=1}^{T}\epsilon_{t}\widehat{d}_{j,t}\right) \\ &+ \left(T^{-1}\epsilon_{j}\mathbf{U}_{j,t}^{0\prime}\left(\widehat{\mathbf{H}}_{jj}^{-1}\right)'\widehat{\mathbf{H}}_{jj}^{-1}\right)' \left(\frac{1}{T_{j}}\sum_{t=1}^{T}\epsilon_{t}\widehat{d}_{j,t}\right) \\ &= I + II + III. \end{aligned}$$
(D.38)

Using (D.32), it follows that

$$I = \left(\widehat{\mathbf{H}}_{jj}T^{-1}\left(\widehat{\mathbf{H}}_{jj}^{-1}\mathbf{U}_{j,t}^{0} - \widehat{\mathbf{U}}_{j,t}\right)\widehat{\mathbf{U}}_{j,t}'\widehat{\mathbf{H}}_{jj}^{-1}\right)\left('\mathbf{B}_{j}^{0'}\frac{1}{T_{j}}\sum_{t=1}^{T}\epsilon_{t}\widehat{d}_{j,t}\right) = O_{P}\left(N^{1/2}T_{j}^{-1/2}C_{N,T}^{-2}\right);$$

further, we have

$$\sum_{i=1}^{N} \left(\frac{1}{T_j} \sum_{t=1}^{T} \epsilon_{i,t} \widehat{d}_{j,t} \right) T^{-1} \sum_{t=1}^{T} \epsilon_{i,t} \widehat{d}_{j,t} \left(\widehat{\mathbf{u}}_{j,t}' - \mathbf{u}_{j,t}^{0\prime} \left(\widehat{\mathbf{H}}_{jj}^{-1} \right)' \right) = O_P \left(N T_j^{-1/2} C_{N,T}^{-2} \right)$$
$$\sum_{i=1}^{N} \left(\frac{1}{T_j} \sum_{t=1}^{T} \epsilon_{i,t} \widehat{d}_{j,t} \right) \left(\frac{1}{T} \sum_{t=1}^{T} \epsilon_{i,t} \mathbf{u}_{j,t}' \widehat{d}_{j,t} \right) = O_P \left(N T^{-1} \right),$$

whence $II = O_P(NT_j^{-1/2}C_{N,T}^{-2})$ and $III = O_P(NT^{-1})$. Finally, using the same arguments as above, it holds that IV is dominated. Putting all together, we conclude that, in (D.29), $II = O_P(NT^{-1})$, whence

$$i'_{N}\widehat{\mathbf{M}}_{B,j}\frac{1}{T_{j}}\sum_{t=1}^{T}\epsilon_{t}\widehat{d}_{j,t} = O_{P}\left(N^{1/2}T^{-1/2}\right) + O_{P}\left(NT^{-1}\right).$$
(D.39)

Putting (D.25), (D.27) and (D.39) together, it follows that, in (D.24)

$$N^{1/2}\left(\widehat{\gamma}_{j,0} - \gamma_{j,0}\right) = \left(\frac{i'_{N}\mathbf{M}^{0}_{B,j}i_{N}}{N}\right)^{-1} \left(\frac{i'_{N}\mathbf{M}^{0}_{B,j}\alpha_{j}}{N^{1/2}}\right) + O_{P}\left(N^{1/2}T^{-1}\right) + o_{P}\left(1\right).$$
(D.40)

Recalling that, by Lemma C.8, $\hat{\pi}_j - \pi_j = O_P(T^{-1})$ with $\eta > \frac{1}{2}$, the desired result follows from Assumption 7(iii) and the Cramer-Wold device.

Proof of Theorem A.1. The proof is essentially the same as the proof of Theorems 3 and 4 in Horváth and Trapani (2019) and most of it is therefore omitted. In order to understand the role of (A.3), note that

$$\begin{split} M_{\theta}^{-1/2} & \sum_{m=1}^{M_{\theta}} \left(\zeta_{m}^{\theta}\left(s\right) - \frac{1}{2} \right) = M_{\theta}^{-1/2} \sum_{m=1}^{M_{\theta}} \left(\zeta_{m}^{\theta}\left(0\right) - \frac{1}{2} \right) \\ &+ M_{\theta}^{-1/2} \sum_{m=1}^{M_{\theta}} \left(\zeta_{m}^{\theta}\left(s\right) - \zeta_{m}^{\theta}\left(0\right) - \int_{0}^{s/f_{N,T}} \frac{1}{\sqrt{2\pi}} \exp\left(-\frac{1}{2}x^{2}\right) dx \right) \\ &+ M_{\theta}^{-1/2} \sum_{m=1}^{M_{\theta}} \int_{0}^{s/f_{N,T}} \frac{1}{\sqrt{2\pi}} \exp\left(-\frac{1}{2}x^{2}\right) dx. \end{split}$$

By the same passages as in the proof of Theorem 3 in Horváth and Trapani (2019), it is easy to see that

$$M_{\theta}^{-1/2} \sum_{m=1}^{M_{\theta}} \left(\zeta_{m}^{\theta} \left(0 \right) - \frac{1}{2} \right) = O_{P^{*}} \left(1 \right);$$

also

$$E^* \int_{-\infty}^{\infty} \left| M_{\theta}^{-1/2} \sum_{m=1}^{M_{\theta}} \left(\zeta_m^{\theta} \left(s \right) - \zeta_m^{\theta} \left(0 \right) - \int_0^{s/f_{N,T}} \frac{1}{\sqrt{2\pi}} \exp\left(-\frac{1}{2} x^2 \right) dx \right) \right|^2 ds = o\left(1 \right),$$

and

$$\int_{-\infty}^{\infty} \left| M_{\theta}^{-1/2} \sum_{m=1}^{M_{\theta}} \int_{0}^{s/f_{N,T}} \frac{1}{\sqrt{2\pi}} \exp\left(-\frac{1}{2}x^{2}\right) dx \right|^{2} ds = c_{0} \frac{M_{\theta}}{f_{N,T}^{1/2}} = o\left(1\right),$$

after (A.3). The rest of the proof follows immediately.

The Data Analytics for Finance and Macro (DAFM) Research Centre

Forecasting trends is more important now than it has ever been. Our quantitative financial and macroeconomic research helps central banks and statistical agencies better understand their markets. Having our base in central London means we can focus on issues that really matter to the City. Our emphasis on digitalisation and the use of big data analytics can help people understand economic behaviour and adapt to changes in the economy.

Find out more at kcl.ac.uk/dafm

This work is licensed under a Creative Commons Attribution Non-Commercial Non-Derivative 4.0 International Public License.

You are free to: Share – copy and redistribute the material in any medium or format

Under the following terms:

Attribution – You must give appropriate credit, provide a link to the license, and indicate if changes were made. You may do so in any reasonable manner, but not in any way that suggests the licensor endorses you or your use.

Non Commercial – You may not use the material for commercial purposes.

No Derivatives – If you remix, transform, or build upon the material, you may not distribute the modified material.



Data Analytics for Finance & Macro Research Centre

